University of Nebraska - Lincoln DigitalCommons@University of Nebraska - Lincoln

Sociology Theses, Dissertations, & Student Research

Sociology, Department of

Winter 12-2014

CONTEXTUALIZING COUPLES: THREE ESSAYS ON INEQUALITY, STRESS, AND DYADIC FUNCTIONING AS A LONGITUDINAL AND RECIPROCAL PROCESS

Deadric T. Williams University of Nebraska-Lincoln, dwilliams9@unl.edu

Follow this and additional works at: http://digitalcommons.unl.edu/sociologydiss Part of the <u>Counseling Commons</u>, and the <u>Family, Life Course, and Society Commons</u>

Williams, Deadric T., "CONTEXTUALIZING COUPLES: THREE ESSAYS ON INEQUALITY, STRESS, AND DYADIC FUNCTIONING AS A LONGITUDINAL AND RECIPROCAL PROCESS" (2014). Sociology Theses, Dissertations, & Student Research. 34. http://digitalcommons.unl.edu/sociologydiss/34

This Article is brought to you for free and open access by the Sociology, Department of at DigitalCommons@University of Nebraska - Lincoln. It has been accepted for inclusion in Sociology Theses, Dissertations, & Student Research by an authorized administrator of DigitalCommons@University of Nebraska - Lincoln.

CONTEXTUALIZING COUPLES: THREE ESSAYS ON INEQUALITY, STRESS, AND DYADIC FUNCTIONING AS A LONGITUDINAL AND RECIPROCAL

PROCESS

by

Deadric T. Williams

A DISSERTATION

Presented to the Faculty of

The Graduate College at the University of Nebraska

In Partial Fulfillment of Requirements

For the Degree of Doctor of Philosophy

Major: Sociology

Under the Supervision of Professor Jacob E. Cheadle

Lincoln, Nebraska

December, 2014

CONTEXTUALIZING COUPLES: THREE ESSAYS ON INEQUALITY, STRESS, AND DYADIC FUNCTIONING AS A LONGITUDINAL AND RECIPROCAL PROCESS

Deadric T. Williams, Ph.D. University of Nebraska, 2014

Adviser: Jacob E. Cheadle

In this dissertation, I use an integrated theoretical and conceptual model that consists of several theoretical frameworks to examine the following questions: (1) is there a longitudinal and reciprocal association between parental stress/distress and dyadic functioning? (2) does the association change over time? (3) does the association vary across social contexts (e.g., marital status, race/ethnicity, and poverty)? In order to explore these questions, I use longitudinal and dyadic data from the Fragile Families and Child Well-being Study, which follows a cohort of children and their parents from birth to five years of age. Through three separate analytic studies, the results indicate that (a) economic hardship affects both mothers' and fathers' depressive symptoms over time, mothers' depressive symptoms affect family hardships during the earlier years, and hardship and depressive symptoms are associated with distress in the relationship for both parents by their child's fifth birthday. No differences emerge between families (married and cohabiting); however, differences between mothers and fathers were revealed in the analysis for relationship distress; (b) parents' depressive symptoms and cooperative coparenting are longitudinally and reciprocally related. Differences between race and ethnic groups tend be largely contingent upon the developmental age of the child; and (c) the longitudinal and reciprocal association between parental stress and couple's relationship quality was largely unidirectional and only for mothers-that is, couple's relationship quality reduced maternal parenting stress. The findings were similar across families who did not live in poverty over time and for families who lived in persistent poverty. For families who experience transient poverty, only paternal parenting stress was associated with lower levels of couple's relationship quality. All in all, the results demonstrate that individuals within families are interdependent and parents are involved in interlocking trajectories as their child ages and develops over time. The variations

across chapters points to the overall complexity of family life. Thus, rather than driving home a consistent message, the results illustrate that different domains, whether dyadic or individual, personal or interpersonal, move according to their own rules. To positively influence family life, multiple pathways must be targeted if we, as a society, are willing to help families achieve adequate financial support and family stability. These findings enhance our understating of interpersonal and contextual stressors, dyadic functioning, reciprocity within couples, and the importance of cross-partner associations. Copyright

Copyright 2014, Deadric T. Williams

Dedication

This work is dedicated to three sets of people: First, to my great-grandmother, my grandmother, and my mother – three generations of strong, African American women. Second, to my social fathers – men who helped me to overcome my fears, and pushed me to go back to school. Last, but not least, to my children – who provided the motivation to endure my academic journey.

Acknowledgements

Writing and finishing a dissertation involves several people, doing different things, all at the same time – similar to a well-oiled engine. I am forever indebted to the synergy that unfolded throughout the writing process. I would like to thank my adviser, Jacob Cheadle, who helped me to funnel my ideas into workable projects. I have been fortunate to have an advisor who sincerely cared about my overall growth and trajectory by not only meeting regularly with me and providing feedback but also being extremely patient. Our relationship have evolved over time from a teacher-student to mentor-mentee. As a consequence, I've grown comfortable with my own theoretical, analytical, and statistical skills as a professional social scientist.

I would also like to thank additional committee members who have played an important part in my growth and development, namely Bridget Goosby and Julia McQuillan. There were many times that I came to their offices unannounced and began babbling about sociological ideas. Of course, they listened, and in many cases offered constructive feedback. They both helped me in more specific ways too. Bridget often discussed with me about the brutal reality of being an African-American scholar, and how to position myself to be the best scholar I can be. It was Julia's course on Hierarchal Linear Modeling that sparked my interest in longitudinal modeling. Julia was patient with me and other students in the class, and as a consequence, I was able to eventually "get it." I was able to think about data in a new and exciting way. Also, I am thankful to my outside committee member who helped me at different stages of graduate school. Catherine Huddleston-Casas was present during the dissertation proposal defense, and offered much needed constructive criticism that helped me to think more broadly about the dissertation.

I would like to acknowledge additional faculty members who has been helpful my entire time at the University of Nebraska-Lincoln. Dan Hoyt, who was Department Chair when I first enrolled, helped to recruit me to Nebraska. His ability and willingness to help me in the beginning set the stage to be able to write and finish the dissertation. Although Christina Falci did not play a direct role in my dissertation, her role was more indirect. Whether it was during class or at academic conferences, Christina treated me like a true professional. In some ways, having a faculty member outside of your dissertation committee offer professional courtesy, helped me to understand another dimension of the professorship.

Much of the time and energy allocated to graduate school in general, and a dissertation in particular, is often at the expense of time and energy not spent with family. I am grateful to my wonderful wife, partner, and friend, Amber Williams, who helped me in many ways. There were several times in which she encouraged me to write (even when I did not want to). She constantly said: "You need to get this thing done." Indeed, having a supportive partner makes the stress of writing a dissertation more manageable. Also, I must acknowledge my children, mother, siblings, and friends who were often neglected during this process. I am thankful for the limited time I got to spend with each of them.

In addition, I was fortunate to receive funding to support my research from both the Department of Sociology at the University of Nebraska-Lincoln and the American Sociological Association Minority Fellowship Program.

To all, thank you!

Table of Contents

Copyright	iii
Dedication	iv
Acknowledgements	v
CHAPTER I	1
INTRODUCTION	1
CONCEPTUAL AND THEORETICAL MODEL	
BREIF DESCRIPTION OF DATA	7
THE CONTRIBUTIONS	
CHAPTER PREVIEWS	
REFERENCES	13
CHAPTER II	
INTRODUCTION	17
LITERATURE REVIEW	19
Direct Association Between Economic Hardship and Depression	
Reciprocal Association Between Depression and Economic Hardship	23
Economic Hardship and Depression on Relationship Distress	25
Differences Between Families: Married and Cohabitors	
Differences Within Families: Mothers and Fathers	
Additional Factors	
METHODS	
Data	
Measures	
Depressive Symptoms	
Economic Hardship	
Relationship Distress	
Control Variables	
Analytic Strategy	
RESULTS	
Descriptive Statistics	41

Structural Equation Models	
All Couples	
Differences Between Families: Married and Cohabiting	44
Differences Within Families: Mothers and Fathers	46
DISCUSSION	
Implications	
Limitations	53
Strengths	
CONCLUSION	
REFERENCES	
APPENDIX A	
CHAPTER III	
INTRODUCTION	
LITERATURE REVIEW	
Theoretical Perspective	
Depressive Symptoms and Coparenting	
Coparenting and Depressive Symptoms	91
Race/Ethnicity, Parental Depression, and Coparenting	
Coparenting Relationships Over Time	95
Additional Factors	96
METHODS	
Data	
Measures	
Depressive Symptoms	
Coparenting	
Race/Ethnicity	
Additional Factors	
Analytic Strategy	
RESULTS	
Descriptive Statistics	
Actor-Partner Interdependent Models	
Depressive Symptoms and Coparenting	

Coparenting and Depressive Symptoms	107
Race/Ethnicity, Depressive Symptoms, and Coparenting	107
DISCUSSION	109
Implications	112
Limitations	113
Strengths	114
CONCLUSION	115
REFERENCES	116
APPENDIX B	124

CHAPTER IV	
INTRODUCTION	
LITERATURE REVIEW	
Theoretical Perspective	
Parenting Stress and the Quality of Mother-Father Relationshi	ps145
The Quality of Mother-Father Relationships and Parenting	
Stress	
Changes in the Associations as Children Develop Over Time.	
Living in (and out) of Poverty	
Additional Factors	
METHODS	
Data	
Measures	
Parenting Stress	
Relationship Quality	
Poverty Histories	
Control Variables	
Analytic Strategy	
RESULTS	
Descriptive Statistics	
Multivariate Models	
DISCUSSION	
Implications	

Limitations	
Strengths	
CONCLUSION	
REFERENCES	
APPENDIX C	
CHAPTER V	
RE-INTRODUCTION	
DISCUSSION	
Conceptual & Theoretical Model	
Implications	
Limitations	
Strengths	
CONCLUSION	
REFERENCES	

List of Tables

Table 2.1: Sample Statistics (Means, Percentages, and Standard Deviations) among Couples, by Marital Status
Table 2.2: Weighted Means for Economic Hardship, Depressive Symptoms, andRelationship Distress, by Marital Status76
Table 2.3: Exponentiated Beta Coefficients for the Longitudinal and Reciprocal Effectson Economic Hardship and Depressive Symptoms, by Martial Status77
Table 2.4: Standardized Regression Coefficients for the Effects of Economic Hardshipand Depressive Symptoms on Relationship Distress
Table 2.5: Attrition for Mothers and Fathers Between Years 1 - 3 79
Table 2.6: Attrition for Mothers and Fathers Between Years 3 - 5 5
Table 2.7: Relationship Dissolution Between Years 1 - 3 80
Table 2.8: Relationship Dissolution Between Years 3 - 5 80
Table 2.9: Control Variables on the Longitudinal and Reciprocal Effects 81
Table 2.10: Control Variables for the Longitudinal and Reciprocal Effects (Marrieds Only) 82
Table 2.11: Control Variables for the Longitudinal and Reciprocal Effects (Cohabitors Only)
Table 2.12: Control Variables on Relationship Distress
Table 3.1: Weighted Descriptive Statistics by Race/Ethnicity (N = 1052) 127
Table 3.2: Weighted Mean Differences 128
Table 3.3: Parameter Estimates for the Association between Depressive Symptoms and Coparenting 129
Table 3.4: Standardized Coefficients Linking Depressive Symptoms to Coparenting, by Race/Ethnicity
Table 3.5: Exponentiated Estimates Linking Coparenting to Depressive Symptoms, by Race/Ethnicity
Table 3.6: Attrition for Mothers and Fathers Between Years 1 - 3

Table 3.7: Attrition for Mothers and Fathers Between Years 3 - 5
Table 3.8: Relationship Ended Between Years 1 - 3 134
Table 3.9: Relationship Ended Between Years 3 - 5 135
Table 3.10: Standardized Estimates for Control Variables on Coparenting at Year-3 and Year 5 (All Couples) 136
Table 3.11: Negative Binomial Estimates for Control Variables on Depressive Symptoms at Year-3 and Year 5 (All Couples) 137
Table 3.12: Standardize Estimates for Control Variables on Coparenting by Race/Ethnicity (Year-3) 138
Table 3.13: Standardize Estimates for Control Variables on Coparenting by Race/Ethnicity (Year-5) 139
Table 3.14: Negative Binomial Coefficients for Control Variables on Depressive Symptoms by Race/Ethnicity (Year-3)
Table 3.15: Negative Binomial Coefficients for Control Variables on Depressive Symptoms by Race/Ethnicity (Year- 5)
Table 4.1: Descriptive Statistics, by Poverty Histories
Table 4.2: Means on Parenting Stress and Couple's Relationship Quality 182
Table 4.3: Standardized Parameter Estimates 183
Table 4.4: Standardized Parameter Estimates, by Poverty Histories 184
Table 4.5: Attrition for Mothers and Fathers between Years 1 and Year 3 185
Table 4.6: Attrition for Mothers and Fathers between Year 1 and Year 3 185
Table 4.7: Relationship Dissolution for Mothers and Fathers between Year 1 and Year 3
Table 4.8: Relationship Dissolution for Mothers and Fathers between Year 3 and Year 5
Table 4.9: Parameter Estimates for Control Variables on Relationship Quality (Y- Standardized) 187

Table 4.10: Parameter Estimates for Control Variables on Parenting Stress (Y-Standardized)	. 188
Table 4.11: Parameter Estimates for Control Variables on Relationship Quality (Y-Standardized), by No Poverty	. 189
Table 4.12: Parameter Estimates for Control Variables on Parenting Stress (Y-Standardized), by No Poverty	. 190
Table 4.13: Parameter Estimates for Control Variables on Relationship Quality (Y- Standardized), by Transient Poverty	. 191
Table 4.14: Parameter Estimates for Control Variables on Parenting Stress (Y-Standardized), by Transient Poverty	. 192
Table 4.15: Parameter Estimates for Control Variables on Relationship Quality (Y- Standardized), by Persistent Poverty	. 193
Table 4.16: Parameter Estimates for Control Variables on Parenting Stress (Y-Standardized), by Persistent Poverty	. 194

List of Figures

Figure 1.1: Conceptual and Theoretical Framework: A Model Depicting the Association between Parental Stress/Distress, Dyadic Functioning, and Contextual Factors Over Time among Couples with a Young Child
Figure 2.1: Conceptual Model for the Longitudinal and Reciprocal Effects of Economic Hardship and Depressive Symptoms among Couples
Figure 2.2: Conceptual Model Linking Economic Hardship, Parents' Depressive Symptoms, and Parents' Relationship Distress
Figure 3.1: Conceptual Model Linking Depressive Symptoms, Coparenting, and Race/Ethnicity over Time
Figure 3.2: Latino Ancestry among Mothers 125
Figure 3.3: Latino Ancestry among Fathers 125
Figure 3.4: Geographic Location for Mothers, by Race/Ethnicity 126
Figure 3.5: Geographic Location for Fathers, by Race/Ethnicity 126
Figure 4.1: Conceptual Model for Parenting Stress, Couple's Relationship Quality, Poverty Histories Over Time

CHAPTER I

Motivation & Introduction

INTRODUCTION

Over the past decade, increased attention has been given to the importance of strengthening couple's relationships among low-income parents and families. Much of this attention reflects changes in broader family processes such as the growing rate of children born to unmarried parents (Copen, Daniels, and Mosher 2013), the risk of relationship dissolution between parents (Cherlin 2009, 2010), economic inequality associated with various family forms (McLanahan 2004, 2009), and how these processes adversely affect children's well-being (Brown 2010; Thomson, Hanson, and McLanahan 1994). As such, policy makers and researchers aim to identify and address the factors that contribute and impact the stability and quality of relationships among parents with young children. To this end, scholars are making concerted efforts to understand parenthood, the fluidity of couple's relationships, and the social context in which parents and families are embedded (Johnson 2012) with intentions to ascertain ways to reduce family inequality and ensure children's well-being.

A point of emphasis for many empirical and intervention studies examining couple's relationships focuses on parenthood dynamics and the interactional processes between parents after the birth of a child. Indeed, parenthood is major life event that consists of both rewards and strain (Nomaguchi and Milkie 2003). Although having a child is associated with life satisfaction and happiness (Nelson et al. 2013), a plethora of research reveals that parents with young children are at an elevated risk for stress and distress (e.g., Umberson, Pudrovska, and Reczek 2010), and the demands of parenting a young child often create strain on the relationship between intimate partners (Umberson and Reczek 2007). These latter findings are critically important given the adverse effects of parental stress and the quality of parental relationships have on families and children's wellbeing (Benzies, Harrison, and Magill-Evans 2004; Crnic, Gaze, and Hoffman 2005; Cui, Donnellan, and Conger 2007). In fact, scholars have paid close attention to ways in which stress affects family and dyadic functioning (Lavee 2013; Randall and Bodenmann 2009), and the importance of positive and supportive family relationships in reducing stress and distress (Cutrona 1996; Thoits 2011; Umberson and Montez 2010). These studies, however, have been part of two separate research agendas. On the one hand, there is a line of research examining the extent to which stress/distress affects dyadic functioning. On the other hand, there are studies exploring how dyadic functioning (e.g., spousal/partner support) affects stress/distress. Consequently, a critical question remains largely unanswered: how do parental stress/distress and positive dyadic functioning *influence one another?* Specifically, is the association from parental stress/distress to dyadic functioning, vice versa, or both?

This dissertation attempts to bridge these separate research agendas in order to understand the stress-dyadic functioning conundrum while also providing empirical insight on couple's relationships, which may be valuable for intervention and programmatic efforts designed to build strong families. Thus, *the central purpose of this dissertation is to examine indices of parental stress/distress and dyadic functioning as a longitudinal and reciprocal process*. Moreover, the association between parental stress/distress and dyadic behaviors may be a function of their child's developmental stage. For instance, developmental perspectives rest upon the notion that individuals change over time (e.g., Elder 1998), and with such changes come different needs for both children and intimate partners. Specifically, as parents' respond to their child's everchanging needs, mothers and fathers often encounter different stressors, and the nature and quality of the intimate partnership also changes—for better or for worse. Even more, both parental stress/distress and dyadic functioning may also vary across social context. Social context refers to the content and conditions that affect and shape the experiences that many individuals and families encounter (Karney and Bradbury 2005). This research uses marital status, race and ethnicity, and poverty as individual and familial characteristics that can shape the nature of familial context (Huston 2000; Johnson 2012).

Thus, using panel data from the Fragile Families and Child Well-being Study, this dissertation attempts to contribute to and build on prior research on parental stress/distress and dyadic functioning among couples, how these processes vary over time, and across social contexts. Specifically, the research is organized around three fundamental questions: (1) *Is there a longitudinal and reciprocal association between parental stress/distress and dyadic functioning?* (2) *Does the association change over time?* (3) *Does the association vary across social contexts (e.g., marital status, race/ethnicity, and poverty)?*

CONCEPTUAL AND THEORETICAL MODEL

Figure 1.1 displays the conceptual model that informs the theoretical relationships between constructs and guides the empirical analyses across the analytic studies in Chapters II, III, and IV. The figure encompasses an integrated framework that joins together several theories, and presents the direction of hypothesized associations between the key factors: (a) stress/distress, (b) dyadic functioning, (c) contextual factors, and (d) the role of time (i.e., children's developmental stages). Notably, this framework builds on prior models that emphasize the importance of stress and contextual factors for studying intimate partners (Huston 2000; Johnson 2012; Karney and Bradbury 1995) by incorporating both direct and reciprocal pathways between parental stress/distress and dyadic functioning, and examining the continuity and change in these processes as a function of children's early developmental stages. The brief review of research presented below is organized in accordance with the conceptual and theoretical model implemented in this dissertation.

<INSERT FIGURE 1.1 HERE>

First, Path A represents the association from parental stress/distress to dyadic functioning. Drawing from family stress (Conger, Conger, and Martin 2010) and stress spillover (Bolger et al. 1989; Neff and Karney 2007) perspectives, the expectation is that stress/distress, whether economic (e.g., economic hardship), interpersonal (e.g., parenting stress), or psychological (e.g., depressive symptoms), directly impacts dyadic functioning between partners. Stress often leads to conflict and/or withdrawal between intimate partners, creating a contagion of stressful experiences that affects the well-being of couple dyads. Reciprocally, Path B demonstrates that indices of dyadic functioning is associated with parental stress/distress. This path relies on research that suggests that being in a positive, supportive, and well-functioning relationship gives individuals a sense of emotional sustenance (love, care, attention, etc.) that tends to work as a coping resource in lowering stress (Cutrona 1996; Don and Mickelson 2012; Lavee 2013). Taken together, Paths A and B indicates the hypothesis that there is reciprocity between parental stress/distress and dyadic functioning.

Paths C and D indicate that individual and family characteristics (e.g., family status, race/ethnicity, poverty) may moderate the association between parental stress and dyadic functioning. These characteristics typically provide the content that shape familial context (Johnson 2012; Karney and Bradbury 2005). Given that *cohabiting couples* (compared to married couples), race and ethnic minorities (compared to Whites), and *families living in poverty* (compared to families living out of poverty) tend to experience higher levels of psychological distress (Umberson et al. 2010), increased levels of stress in parenting (Cooper et al. 2009; Nomaguchi and House 2013; Raikes and Thompson 2005), and display lower levels of relationship quality (Brown and Booth 1996a; Bulanda and Brown 2007; McLanahan and Beck 2010), it is expected that these factors may moderate the association between parental stress/distress and dyadic functioning in a number of ways. For example, because stress/distress is more prevalent in the aforementioned social groups, the negative impact of stress/distress on dyadic functioning may be exacerbated for these groups (Path C). In addition, being in positive and supportive relationship may be more meaningful for each social group (described above), and thus attenuate the levels of parental stress/distress (Path D). Equally, no differences may emerge in the associations as stress may affect all couples, and positive, well-functioning relationships may reduce stress/distress across social group.

Finally, drawing from life course and developmental perspectives (Elder 1998), the conceptual model includes the notion of time (i.e., children's developmental stages) as an important factor that may affect the extent to which parental stress and dyadic functioning are associated. These perspectives contend that as children grow and change, so do their needs and demands. As such, parents are faced with adjusting to those needs which may affect both parental stress/distress and dyadic functioning between parents. The association between stress/distress and dyadic functioning may change as a function of their child's developmental stage in two ways: On the one hand, there is reason to expect that children's early development (i.e., infant to toddler) may be most critical for parents because children need more attention and care compared to older developmental stages (i.e., toddler to preschool) leading to an increase in parental stress, which puts strain on couple's relationships (Milkie et al. 2004; Nomaguchi and Milkie 2003). Likewise, having a supportive partner during this early stage may prove to be particularly helpful relative to later stages. On the other hand, the toddler to preschool developmental stages may prove to be more challenging compared to earlier phases because children tend to become more independent and begin to test parental limits, thus creating stress and strain (Putnick et al. 2010; Schoppe-Sullivan et al. 2004). Moreover, dyadic functioning may be most needed during the toddler to preschool years than earlier developmental years. Thus, in either case, there is some expectation that stress/distress, dyadic functioning, and children's stage of development operate in tandem.

Taken together, the conceptual model presented in Figure 1.1 provides an integrated framework for addressing the longitudinal and reciprocal relationship between stress/distress and dyadic functioning, the contextual factors that may moderate the association, and the importance of the continuity and change in these associations as children develop over time. The framework not only highlights the complexity of intimate relationships but also gives equal importance to an array of factors that

contributes to the functioning of parents with a young child. This integrated framework is especially important given the policy and programmatic efforts to strengthen couple's relationships.

BREIF DESCRIPTION OF DATA

To test the conceptual and theoretical model presented above, this dissertation uses data from the Fragile Families and Child Well-being Study (FFCW). The FFCW study is a longitudinal and representative sample (when weighted) of births in large U.S. cities with populations of 200,000 or more in the late 1990s and early 2000s, and includes an oversample of unmarried parents and a comparable married sample (Reichman et al. 2001). Data for the baseline survey was collected between 1998 and 2000. Mothers were interviewed in the hospital within 48 hours after given birth, and fathers were interviewed in the hospital or as soon as possible following the birth. Both parents were reinterviewed when the child was one- (1999-2002), three- (2001-2003), and five- (2003-2005) years of age. The FFCW study is a rich dataset that includes several factors that tap into several indices of parental stress/distress, the nature and quality of intimate relationships, and includes a host of demographic and contextual variables. Moreover, these data provide identical measures of parental stress/distress and dyadic functioning for both parents over time which allows researchers to examine the continuity and change in these factors. Thus, these data are ideal for examining the conceptual and theoretical model above.

The samples for each empirical chapter consist of couples living together (married and cohabiting) at the baseline survey and who remained together at the 1-, 3-, and 5-year follow-up surveys. The sample size varies across studies due to attrition, relationship

dissolution, and missing data on key variables of interest. Indeed, multiple imputation was used in each empirical chapter to partly address these concerns. Overall, the results in each study can be generalized to U.S. couples living together in urban cities over the first five years after a child's birth.

THE CONTRIBUTIONS

This dissertation makes a number of important contributions to the extant research on parental stress, dyadic functioning, and couples with young children. First, by examining an integrated model of parental stress and dyadic functioning with longitudinal data, my research tests and expands on important findings that have been largely studied unidirectionally. My contribution is to examine parental stress and dyadic functioning as potentially dynamic and reciprocal processes using longitudinal data. Therefore the analyses allow both parental stress and dyadic functioning to be evaluated simultaneously and on equal footing. Findings from this dissertation offer important insights into the parental stressors that infringe upon intimate relationships, and highlight the relational processes between parents that helps to reduce parental stress/distress.

Second, by elaborating these processes across time as children develop and across social contexts (i.e. marital status, race/ethnicity, poverty), this research seeks to further illuminate the ways in which families experience their lives. Indeed, both a child's developmental stage and social context have serious implications over the life course of many families. Because children pose critical challenges to their parents at different developmental stages, the findings can help reveal when parents' relationships are most vulnerable and assist programmatic efforts by helping couples at the most critical times. Indeed, low stress levels and positive relationship quality benefits children's long-term

overall growth trajectory (e.g., Brown 2010). Moreover, given that individual and family characteristics shapes family context in many ways, the findings can assist in creating more culturally sensitive and context specific ways to help couples and families.

Third, information regarding parental stress/distress and dyadic functioning is leveraged from both mothers and fathers. Prior research has used mothers' reports to gauge the dyadic relationship. With an increase in the availability in dyadic data, researchers are beginning to examine familial processes in more holistic ways. These data allow for more critical theorizing and rigorous analytic techniques that takes into account the complex nature of families. Moreover, among heterosexual couples, experiencing stress/distress and levels of dyadic functioning are unequally distributed between parents, especially after the birth of a child. For instance, mothers are more likely to engage in early parenting and report lower levels of relationship quality than fathers (Bianchi and Milkie 2010; Umberson et al. 1996). As such, highlighting differences between mother and fathers may help to develop sex-specific programmatic efforts in reducing parental stress/distress and enhancing intimate partnerships. Indeed, some studies have shown that a couple-focused approach may prove to be more promising for policies aimed to sustain relationship health compared to parent-focused approaches (Cowan and Cowan 2008).

Finally, because prior research guiding intervention efforts to strengthen couple's relationship were disproportionately guided by samples of White, middle-class married two-parent families, this research uses a representative sample of births to a diverse set of parents living in urban areas. All in all, this research highlights how parental stress/distress and indices of dyadic functioning are linked as their child develop from an

infant to toddlerhood to preschool, and informs policies and programs designed to strengthen couples and families.

CHAPTER PREVIEWS

The aforementioned conceptual and theoretical model discussed above guides the subsequent analytic chapters (Chapters II, III, and IV). First, Chapter II examines the association between economic hardship, parents' depressive symptoms, and relationship distress. The chapter addresses (a) the longitudinal, reciprocal, and dyadic association between economic hardship and parents' depressive symptoms as their child ages over time, (b) the effects of economic hardship and parents' depressive symptoms is associated with relationship distress for both parents, (c) whether the effects differ across family status (married vs. cohabitors), and (d) examines differences within families (mothers vs. fathers). There is evidence that economic hardship affects both mothers' and fathers' depressive symptoms over time, mothers' depressive symptoms affect family hardships during the earlier years, and hardship and depressive symptoms are associated with distress in the relationship for both parents by their child's fifth birthday. No differences emerge between families (married and cohabiting); however, differences between mothers and fathers were revealed in the analysis for relationship distress: hardship on relationship distress mattered more for fathers than mothers, and depressive symptoms mattered more for mothers than fathers.

Chapter III examines the longitudinal and reciprocal association between parents' depressive symptoms and cooperative coparenting, and whether the associations differ across race and ethnic groups. To evaluate the associations, data analyses were executed using a series of actor-partner interdependent models that were age-specific: from infant

to toddler, and from toddler to preschool. The findings reveal that parents' depressive symptoms and cooperative coparenting are longitudinally and reciprocally related. Differences between race and ethnic groups tend be largely contingent upon the developmental age of the child.

Chapter IV explores the longitudinal and reciprocal association between parenting stress and couple's relationship quality, and whether these processes differ across family poverty history. In order to gauge the dyadic nature of the couple, I operationalized couple's relationship quality by combing both parents view of the relationship. The results indicate that the longitudinal and reciprocal association between parental stress and couple's relationship quality was largely unidirectional and only for mothers—that is, couple's relationship quality reduced maternal parenting stress. The findings were similar across families who did not live in poverty over time and for families who lived in persistent poverty. For families who experience transient poverty, only paternal parenting stress was associated with lower levels of couple's relationship quality.

Chapter V summarizes the research questions and evaluates the overall conceptual and theoretical model by addressing whether the hypothesized relationships between constructs were supported or not supported, and whether the results corroborates the findings from prior studies. The second part of Chapter V integrates the earlier chapters. In addition, I discuss future research, address strengths and weaknesses, and provide implications for research and policy.



Figure 1.1: Conceptual and Theoretical Framework: A Model Depicting the Association between Parental Stress/Distress, Dyadic Functioning, and Contextual Factors Over Time among Couples with a Young Child

REFERENCES

- Benzies, Karen M., Margaret J. Harrison, and Joyce Magill-Evans. 2004. "Parenting Stress, Marital Quality, and Child Behavior Problems at Age 7 Years." *Public Health Nursing* 21(2):111–21.
- Bianchi, Suzanne M., and Melissa A. Milkie. 2010. "Work and Family Research in the First Decade of the 21st Century." *Journal of Marriage and Family* 72(3):705–25.
- Bolger, Niall, Anita DeLongis, Ronald C. Kessler, and Elaine Wethington. 1989. "The Contagion of Stress across Multiple Roles." *Journal of Marriage and the Family* 175–83.
- Brown, Susan L. 2010. "Marriage and Child Well-Being: Research and Policy Perspectives." *Journal of Marriage and Family* 72(5):1059–77.
- Brown, Susan L., and Alan Booth. 1996. "Cohabitation versus Marriage: A Comparison of Relationship Quality." *Journal of Marriage and the Family* 668–78.
- Bulanda, Jennifer Roebuck, and Susan L. Brown. 2007. "Race-Ethnic Differences in Marital Quality and Divorce." *Social Science Research* 36(3):945–67.
- Cherlin, Andrew. 2009. The Marriage-Go-Round. NY: Vintage Books.
- Cherlin, Andrew. 2010. "Demographic Trends in the United States: A Review of Research in the 2000s." *Journal of Marriage and Family* 72(3):403–19.
- Conger, Rand D., Katherine J. Conger, and Monica J. Martin. 2010. "Socioeconomic Status, Family Processes, and Individual Development." *Journal of Marriage and Family* 72(3):685–704.
- Cooper, Carey E., Sara S. McLanahan, Sarah O. Meadows, and Jeanne Brooks-Gunn. 2009. "Family Structure Transitions and Maternal Parenting Stress." *Journal of Marriage and Family* 71(3):558–74.
- Copen, C., Kimberly Daniels, and William D. Mosher. 2013. "First Premarital Cohabitation in the United States: 2006–2010 National Survey of Family Growth." *National Health Statistics Report* 64.
- Cowan, P., and C. Cowan. 2008. "Diverging Family Policies to Promote Children's Well-Being in the UK and US: Some Relevant Data from Family Research and Intervention Studies." *Journal of Children's Services* 3(4):4–16.
- Crnic, Keith A., Catherine Gaze, and Casey Hoffman. 2005. "Cumulative Parenting Stress across the Preschool Period: Relations to Maternal Parenting and Child Behavior at Age 5." *Infant and Child Development* 14:117–32.

- Cui, Ming, M. Brent Donnellan, and Rand D. Conger. 2007. "Reciprocal Influences between Parents' Marital Problems and Adolescent Internalizing and Externalizing Behavior." *Developmental Psychology* 43(6):1544.
- Cutrona, Carolyn E. 1996. Social Support in Couples: Marriage as a Resource in Times of Stress. Sage Publications, Inc.
- Don, Brian Paul, and Kristin D. Mickelson. 2012. "Paternal Postpartum Depression: The Role of Maternal Postpartum Depression, Spousal Support, and Relationship Satisfaction." *Couple and Family Psychology: Research and Practice* 1(4):323.
- Elder, Glen H. Jr. 1998. "The Life Course as Developmental Theory." *Child Development* 69(1):1–12.
- Huston, Ted L. 2000. "The Social Ecology of Marriage and Other Intimate Unions." *Journal of Marriage and Family* 62(2):298–320.
- Johnson, Matthew D. 2012. "Healthy Marriage Initiatives: On the Need for Empiricism in Policy Implementation." *American Psychologist* 67(4):296–308.
- Karney, Benjamin R., and Thomas N. Bradbury. 1995. "The Longitudinal Course of Marital Quality and Stability: A Review of Theory, Method, and Research." *Psychological Bulletin* 118(1):3–34.
- Karney, Benjamin R., and Thomas N. Bradbury. 2005. "Contextual Influences on Marriage Implications for Policy and Intervention." *Current Directions in Psychological Science* 14(4):171–74.
- Lavee, Yoav. 2013. "Stress Processes in Families and Couples." Pp. 159–76 in Handbook of Marriage and the Family. Springer.
- McLanahan, Sara. 2004. "Diverging Destinies: How Children Are Faring under the Second Demographic Transition." *Demography* 41(4):607–27.
- McLanahan, Sara. 2009. "Fragile Families and the Reproduction of Poverty." *The Annals* of the American Academy of Political and Social Science 621(1):111–31.
- McLanahan, Sara, and Audrey N. Beck. 2010. "Parental Relationships in Fragile Families." *The Future of children/Center for the Future of Children, the David and Lucile Packard Foundation* 20(2):17.
- Milkie, Melissa A., Marybeth J. Mattingly, Kei M. Nomaguchi, Suzanne M. Bianchi, and John P. Robinson. 2004. "The Time Squeeze: Parental Statuses and Feelings about Time with Children." *Journal of Marriage and Family* 66(3):739–61.

- Neff, Lisa A., and Benjamin R. Karney. 2007. "Stress Crossover in Newlywed Marriage: A Longitudinal and Dyadic Perspective." *Journal of Marriage and Family* 69(3):594–607.
- Nelson, S. Katherine, Kostadin Kushlev, Tammy English, Elizabeth W. Dunn, and Sonja Lyubomirsky. 2013. "In Defense of Parenthood Children Are Associated With More Joy than Misery." *Psychological Science* 24(1):3–10.
- Nomaguchi, Kei, and Amanda N. House. 2013. "Racial-Ethnic Disparities in Maternal Parenting Stress The Role of Structural Disadvantages and Parenting Values." *Journal of Health and Social Behavior* 54(3):386–404.
- Nomaguchi, Kei M., and Melissa A. Milkie. 2003. "Costs and Rewards of Children: The Effects of Becoming a Parent on Adults' Lives." *Journal of Marriage & Family* 65(2):356–74.
- Putnick, Diane L. et al. 2010. "Stability, Continuity, and Similarity of Parenting Stress in European American Mothers and Fathers across Their Child's Transition to Adolescence." *Parenting: Science and Practice* 10(1):60–77.
- Raikes, H. Abigail, and Ross A. Thompson. 2005. "Efficacy and Social Support as Predictors of Parenting Stress among Families in Poverty." *Infant Mental Health Journal* 26(3):177–90.
- Randall, Ashley K., and Guy Bodenmann. 2009. "The Role of Stress on Close Relationships and Marital Satisfaction." *Clinical Psychology Review* 29(2):105– 15.
- Reichman, Nancy E., Julien O. Teitler, Irwin Garfinkel, and Sara S. McLanahan. 2001. "Fragile Families: Sample and Design." *Children and Youth Services Review* 23(4/5):303–26.
- Schoppe-Sullivan, Sarah J., Sarah C. Mangelsdorf, Cynthia A. Frosch, and Jean L. McHale. 2004. "Associations between Coparenting and Marital Behavior from Infancy to the Preschool Years." *Journal of Family Psychology* 18(1):194.
- Thoits, Peggy A. 2011. "Mechanisms Linking Social Ties and Support to Physical and Mental Health." *Journal of Health and Social Behavior* 52(2):145–61.
- Thomson, Elizabeth, Thomas L. Hanson, and Sara S. McLanahan. 1994. "Family Structure and Child Well-Being: Economic Resources vs. Parental Behaviors." *Social Forces* 73(1):221–42.
- Umberson, Debra, Meichu D. Chen, James S. House, Kristine Hopkins, and Ellen Slaten. 1996. "The Effect of Social Relationships on Psychological Well-Being: Are Men and Women Really so Different?" *American Sociological Review* 837–57.

- Umberson, Debra, and Jennifer Karas Montez. 2010. "Social Relationships and Health: A Flashpoint for Health Policy." *Journal of Health and Social Behavior* 51(1 supplement):S54–S66.
- Umberson, Debra, Tetyana Pudrovska, and Corinne Reczek. 2010. "Parenthood, Childlessness, and Well-Being: A Life Course Perspective." *Journal of Marriage and Family* 72(3):612–29.
- Umberson, Debra, and Corinne Reczek. 2007. "Interactive Stress and Coping around Parenting: Explaining Trajectories of Change in Intimate Relationships over the Life Course." *Advances in Life Course Research* 12:87–121.

CHAPTER II

Economic Hardship, Parents' Depressive Symptoms, and Relationship Distress among Couples

INTRODUCTION

Young children place many demands on parents, and these demands can contribute to lower levels of psychological well-being. The general explanation is that children increase parents' daily stressors, which, in turn, leads to elevated levels of depressive symptoms (Evenson and Simon 2005; Umberson et al. 2010). In fact, studies show that depressive symptoms affect roughly 10-15% of new mothers (Breese McCoy 2011), and about 2-10% of new fathers (Paulson and Bazemore 2010; Wee et al. 2011). Additionally, because parenthood requires significant financial obligations, parents experiencing economic hardship may find the parenting role especially stressful, putting mothers and fathers at a greater risk for depression (Bronte-Tinkew et al. 2007; Manuel et al. 2012). Reciprocally, parents' post-birth depressive symptomology may lead to a decline in employment opportunities, work hours, and household income creating financial hardships within families (Ennis, Hobfoll, and Schroder 2000; Gupta and Huston 2009). Understanding both economic hardship and depressive symptoms, and how they are linked together, after the birth of a child is important because these stressors create elevated levels of distress and conflict between partners (Conger, Conger, and Martin 2010) affecting relationship quality and stability (Hardie and Lucas 2010).

The stress process and family stress theories have been important frameworks for furthering our understanding of the link between economic hardship and depressive symptoms (Pearlin et al. 2005), and how these factors contribute to relationship outcomes in families (Conger, Conger, and Martin 2010; Westman and Vinokur 1998). Although these theories are associated with separate research agendas, there has been a recent push toward integration (Milkie 2010) to highlight the extent to which stress affects both individual and familial processes, and to underscore the interdependent nature of individuals within families (e.g., Cox and Paley 1997). The combination of stress process and family stress theories may be particularly important because hardship contributes to individuals' mental health, while mental health can also contributes to family hardships. These associations may linger over time creating a reciprocal influence.

In addition, both hardship and mental health can create strain and antagonism between partners leading to relationship dissolution. Addressing both individual- and couple-level responses to adverse conditions paints a more holistic picture on the adverse ways in which stress affect families. Thus, the purpose of this study is to (a) examine the links between family economic hardship and parents' depressive symptoms as a longitudinal dyadic process among couples 1-, 3-, and 5-years after the birth of a child, and to (b) understand the extent to which both hardship and depressive symptoms are associated with mother's and fathers' report of relationship distress five years after the birth of a child. Further, given the differences in financial and psychological resources between families (married and cohabiting) and differences in stress exposure within families (mothers and fathers), this chapter also (c) examines whether the processes vary by marital status and gender.

The current study extends previous research in several ways. First, I address the association between economic hardship and depressive symptoms using a longitudinal

dataset that contains a diverse sample of biological parents (mothers and fathers) 1-, 3-, and 5- years after the birth of a child. Understanding how hardship and depressive symptoms unfold over time among couples with young children is important because of the adverse consequences for children's well-being (McLoyd 1998; Ramchandani et al. 2008; Sobolewski and Amato 2005). Second, I examine whether economic hardship and depressive symptoms are associated with relationship distress 5-years after the birth of a child—a time that is critical for the stability for some parents (Cherlin 2010b). Last, I examine whether these processes vary between married and cohabiting couples, and between mothers and fathers. This research presents information on the ways in which couples living together in urban cities during five years after the birth of a child respond to individual and family level stressors.

LITERATURE REVIEW

Figures 2.1 and 2.2 outline the conceptual models guiding the analyses and are used to orient the review of prior research. In Figure 2.1, the conceptual model presents relationships among economic hardship and depressive symptoms as a longitudinal and reciprocal process. In the current study, the stress process framework is utilized to understand the direct effects of early hardship on later depressive symptoms. Some scholars refer to this as the *causation hypothesis* (Muntaner et al. 2013)—which suggests that having fewer economic resources places individuals at risk for mental health problems. The reciprocal effect from early depressive symptoms to later economic hardship has been referred to as the *selection hypothesis*¹—that is, individuals with higher

¹ The current study does not empirically examine the causation versus selection debate. The debate is mentioned here to recognize the plethora of research that has addressed the relationship between socioeconomic status and mental health. Thus, the inferences and conclusions drawn in this chapter do not add to this specific debate. Rather, the current study seeks to understand the extent to which economic hardship and depressive symptoms may covary over time for mothers and fathers after the birth of a child.

levels of depressive symptoms are "selected" into a lower socioeconomic position because their mental health status hinders and prevents them from obtaining upward social mobility through the inability to work (Eaton 2001; Muntaner et al. 2013). Moreover, there is reason to believe that these processes may vary across families, and within families. For example, married couples have more financial resources and better health, when compared to cohabiting couples (Sassler 2010), and within many heterosexual relationships, family processes are organized around gender (Ridgeway 2011). The current study examines how economic hardship and depressive symptoms may reciprocally influence each other over time among couples after the birth of a child. In Figure 2.1, Path A represents the direct effect of economic hardship on depressive symptoms, Path B represents the reciprocal effects leading from depressive symptoms to economic hardship, and Paths C and D highlight the moderating effects of marital status and parents' gender, respectively. The model also takes into account the stability and change in these processes as children develop over time.

<INSERT FIGURE 2.1 HERE>

Figure 2.2 shows the conceptual model displaying the association of economic hardship and depressive symptoms on relationship distress. Similar to marital distress, which reflects couples who have considered divorce or separation (Booth, Johnson, and Edwards 1983; Conger, Rueter, and Elder 1999), the term relationship distress is used because the sample includes both married and cohabiting couples. For the conceptual model (Figure 2.1), I draw on features of family stress theories to examine the direct effects of hardship and depressive symptoms on relationship distress (Conger, Conger, and Martin 2010), and the extent to which one partner's depressive symptoms affect
relationship distress in the other partner (Larson and Almeida 1999a). Economic hardship interferes with the quality of relationships because financial difficulty produces strain and conflict between partners that often leads to poor relationship quality and dissolution (Conger, Conger, and Martin).

Similarly, depressive symptoms affect couple dyads because depressed individuals tend to show less affection and support, and become hostile toward their partner, which often leads to unfavorable relationship outcomes for families (Conger, Conger, Martin 2010; Kouros and Cummings 2011). Taken together, these theories take into account relationship distress as both an individualistic and dyadic process. Similar to the model presented in Figure 2.1, this model takes into account whether the associations vary between and within families. Specifically, in Figure 2.2, Path A represents the direct effects of economic hardship on relationship distress; Path B indicates the effect of depressive symptoms on relationship distress. Paths C and D demonstrates the moderating effects of marital status and parents' gender, respectively.

Direct Association Between Economic Hardship and Depression

Proponents of the stress process perspective argue that economic hardship is a source of stress that is manifested in psychological well-being (Pearlin et al. 1981, 2005). More specifically, economic hardship represents adverse material and economic conditions (e.g., poor housing, inability to pay bills, etc.) that ultimately influence families by affecting individuals' physical health, anxiety, anger, and depression (Edin and Kissane 2010; Goosby 2007; Kahn and Pearlin 2006; McLoyd 1990). For parents with a young child, hardships may interfere with social roles such as being a good mother and father (Avison and Turner 1988; Pearlin et al. 1981, 2005; Ross and Huber 1985). As such, the

inability to carry out socially prescribed roles produces depressed feelings such as despair and hopelessness which takes a toll on parents' overall psychological well-being with persistent economic hardship mattering more than episodic experiences (Kahn and Pearlin 2006; Lynch, Kaplan, and Shema 1997).

Prior studies have consistently found that economic hardship leads to higher levels of depressive symptoms (Heflin and Iceland 2009; Kinnunen and Feldt 2004; Manuel et al. 2012; Mirowsky and Ross 1999, 2001; Solantaus, Leinonen, and Punamaki 2004; Zimmerman and Katon 2005). These studies, however, have focused largely on individual-level data. Examining the association between hardship and depression for couples suggests that although financial hard times are a shared experience within families, the ways in which individuals respond to hardships may depend on the family member. There is some evidence that economic hardship leads to higher levels of depressive symptoms for both partners in couple dyads. For example, Ross and Huber (1985), using a national sample of married couples (N=340 couples), found that economic hardship increases both spouses levels of depressive symptoms. Similarly, Wickrama and colleagues (Wickrama et al. 2010, 2012) reported that economic hardship was significantly related to higher levels of depressive symptoms, and other health outcomes, for married couples during the middle years of life. In another study, using a sample of urban and rural African American couples, (Conger et al. 2002) found that economic pressure is associated with higher levels of depressed mood for both spouses. Other studies using dyadic data reported similar findings (Conger et al. 1999; Kinnunen and Feldt 2004; Parke et al. 2004).

Although these prior studies have increased our understanding of the association between economic hardship and depression in couples, the findings are based on either cross-sectional data (Conger et al. 2002; Ross and Huber 1985) or longitudinal samples that are disproportionately White, middle-aged married couples (Wickrama et al. 2012). Understanding hardship and depressive symptoms is especially important for parents with a young child because children of depressed parents tend to have adverse developmental outcomes such as cognitive and language deficiencies and poor mental health (NICH and Human Development Early Child Care Research Network 2005; Petterson and Albers 2001; Sohr-Preston and Scaramella 2006). The current study moves this line of research forward by examining the effect of economic hardship on depressive symptoms for a diverse sample of couples during their child's early developmental periods: infant, toddler, and preschool. Thus, as shown in Figure 2.1, Path A, I hypothesize that *higher economic hardship levels are associated with elevated depressive symptoms for both mothers and fathers over time.*

Reciprocal Association Between Depression and Economic Hardship

Although there are reasons to believe that socioeconomic status and mental health are mutually reinforcing, many empirical studies have highlighted that economic hardship leads to depressive symptoms rather than vice versa. Yet, there is a strong theoretical argument that contends that higher levels of depressive symptoms are associated with lower socioeconomic status. This line of reasoning suggests that individuals with mental health problems are less likely to obtain and maintain employment opportunities (Dohrenwend et al. 1992; Wadsworth and Achenbach 2005) which, in turn, leads to the inability to purchase essential household goods and services (i.e., food, shelter, health care, etc.). Given the prevalence of psychological distress for parents with young children (Breese McCoy 2011; Evenson and Simon 2005; Umberson et al. 2010), seeking and maintaining employment may be especially challenging, and thus putting families at risk for economic hardship.

Prior research has examined the effects of mental health on economic outcomes. Many of these studies, however, focus on the causation versus selection debate—a debate that spans across nearly five decades (Aneshensel, Phelan, and Bierman 2013; Lorant et al. 2003; Muntaner et al. 2004). Contemporary studies show that depressive symptoms hinder economic opportunities such as employment and income. For example, in a study using panel data from the National Longitudinal Survey of Youth (NLSY), Dooley, Prause, and Ham-Rowbottom (2000) found that respondents who were depressed at the prior wave had an elevated risk for being unemployed subsequently. Similarly, two studies using longitudinal data from Australia, Butterworth et al. (2012) and Olesen et al. (2013) found that respondents' mental health status was significantly associated with the overall time being unemployed. These results are consistent with Whooley et al.'s (2002) study that revealed that depression increased the odds of unemployment; and, respondents who were at risk for depression at the baseline survey were twice as likely to have lower levels of income compared to those without a risk of depression. All in all, these studies show that psychological well-being can affect economic outcomes.

Still, prior studies examining the reciprocal association between depressive symptoms and economic outcomes have not examined the effects as a longitudinal dyadic process among couples with a young child. Again, this may be an important yet critical time as parents begin to regain their post-birth physical, emotional, and psychological equilibrium (Tucker et al. 2010). As such, parents with depression symptoms may need more time at home before returning to work, or may find pursuing employment opportunities challenging (Gupta and Huston 2009). Subsequently, families may begin to suffer economically; and, the adverse economic conditions can lead to both short- and long-term negative consequences for families and children (Edin and Kissane 2010; Sobolewski and Amato 2005). Although previous studies do not examine the effects of depressive symptoms on economic hardship explicitly, prior research shows that depressive symptoms affect other economic outcomes such as income levels and employment status, which are highly related to families' economic and material wellbeing. Thus, I expect that parents' depressive symptoms will affect mothers and fathers' economic well-being in similar ways. As shown in Figure 2.1, Path B, I hypothesize that *mother and father's depressive symptoms will be associated with higher levels of economic hardship over time*.

Economic Hardship and Depression on Relationship Distress

Although the association between economic outcomes (i.e., hardships, employment status, income) and depressive symptoms are involved in a complex mutual reinforcing process, it is rather straightforward that the effects of both economic factors and depression symptoms have detrimental effects on the quality and stability of intimate partnerships (Donnellan et al. 2009; Edin and Kissane 2010; Kalmijn, Loeve, and Manting 2007). The general explanation suggests that financial difficulty and hardships place stress and strain on couples, which leads to conflict, lower levels of relationship quality, and divorce (Conger, Conger, and Martin 2010; White and Rogers 2000).

Several studies have shown that adverse financial circumstances are detrimental to the quality of intimate relationships. For example, Conger et al.'s (1990, 1999) earlier work shows that economic strain lowers marital quality and increases marital distress. More recent studies reveal that economic hardship is associated with higher levels of conflict for young married and cohabiting couples (Hardie and Lucas 2010). Relatedly, studies also demonstrate that economic factors such as unemployment, income, and neighborhood poverty are associated with violent interactions between partners (Benson et al. 2003; Cunradi et al. 2000; Cutrona et al. 2003; DeMaris et al. 2003; Fox and Chancey 1998; Fox et al. 2002). Research also shows that family debt affects marital satisfaction (Dew 2007, 2008), and some married couples tend to argue about money more often compared to other types of disagreements (Papp, Cummings, and Goeke-Morey 2009). These studies show that families' economic circumstances play a vital role in the quality and stability of intimate relationships. Thus, as shown in Figure 2.2, Path A, I hypothesize that *experiencing higher levels of economic hardship increases relationship* distress among mothers and fathers.

<INSERT FIGURE 2.2 HERE>

Depression or depressive symptoms are associated with declines in various relationship quality indices. This is largely due to the fact that depressed individuals are more likely to display behaviors of withdrawal and irritability, provide less support, and show anger toward their intimate partner, which ultimately creates distress between partners (Conger, Conger, and Martin 2010; Coyne 1976; Fiske and Peterson 1991). Depressive symptoms affect relationship outcomes in two ways: First, an individual's own depression leads to lower levels of relationship quality. Many of these studies have been guided by Rand Conger's Family Stress Model. For example, two studies showed that both wife and husband emotional distress is significantly associated with higher levels of marital conflict (Conger et al. 2002, 1999). In a study of European and Mexican American families, Parke et al. (2004) found that mother and father depressive symptoms were linked to higher levels of marital problems. Additional studies show similar results indicating that wife and husband psychological distress is significantly associated with lower levels of martial adjustment (Kinnunen and Feldt 2004) and hostile martial interaction (Solantaus et al. 2004) for both spouses. These studies provide clear evidence that depressive symptoms can affect the quality of intimate relationships. Thus, as shown in Figure 2.2, Path B, I hypothesize that *both parent's depressive symptoms will be associated with higher levels of their own relationship distress (actor effect).*

Second, studies also show that depressive symptoms in one partner can negatively affect how the other partner views the relationship—creating an emotional contagion or crossover effects within families (Hatfield, Cacioppo, and Rapson 1994; Larson and Almeida 1999b; Westman and Vinokur 1998). Emotional crossover is an inter-individual dyadic process where the mood or emotion of an individual leads to the similar mood of another individual (Larson and Almeida 1999b; Westman and Etzion 1995; Westman 2001). Mood transference from one person to another is much more apparent in intimate relationships because these individuals share a common space and life experience (Hatfield et al. 1994; Song, Foo, and Uy 2008). Thus, due to the interdependent nature of families, understanding mood transference of one partner to the other captures the dyadic, interdependent nature of intimate relationships; and, it shines light on the complicated ways in which individuals experience the world and influence those they are closest to (Collins 2004).

Indeed, scholars have paid close attention to emotional contagion in dyadic relationships, and research shows that moods and emotions can be symmetric processes in couple dyads (Westman 2001). For example, in a study of 68 couples, one partner's depressed mood was related to the other partner feeling negative about the relationship (Thompson and Bolger 1999), and when husbands or wives report higher health symptom levels, partners display higher negative mood and lower positive mood (Yorgason, Almeida, and Neupert 2006). Among newlywed couples, (Neff and Karney 2007) found that when wives experienced higher levels of stress, husbands reported lower levels of marital satisfaction, but not vice versa. Other studies have generally supported the negative association between emotional distress and relationship quality (Proulx, Helms, and Buehler 2007; Pruchno, Wilson-Genderson, and Cartwright 2009; Du Rocher Schudlich, Papp, and Cummings 2011; Whisman and Uebelacker 2009). Because couples share and have so much invested together, taking a dyadic approach to understanding how mothers and fathers influence one another is critical for detailing the interdependent nature of family life (Cox and Paley 1997; Neff and Karney 2007; O'Brien 2005). Thus, I offer the following hypothesis: As shown in Figure 2.2, path (c), *mother's depressive* symptoms will be associated with higher levels of relationship distress for fathers; and, father's depressive symptoms will be associated with higher levels of mother's relationship distress (cross-partner effect).

Differences Between Families: Married and Cohabitors

Nonmarital cohabitation has increased over time and, in some cases, provides intimate partners a transitional stage prior to marrying (Bumpass and Lu 2000; Kennedy and Bumpass 2008; Manning and Cohen 2012; Smock 2000). Recent estimates show that 48% of women cohabited as a first union in 2006-2010 compared to only 34% of women in 1995 (Copen et al. 2013). Cohabiting and marital relationships are often selective: that is, individuals with greater resources are more likely to marry and individuals with fewer resources are more likely to cohabit (Bumpass and Lu 2000; Smock 2000). As a result, cohabitors and married couples differ in several ways. For example, compared to married couples, cohabitors tend to have fewer financial resources, higher levels of psychological distress, and display lower levels relationship quality (Brown and Booth 1996b; Brown 2000; McLanahan and Percheski 2008; McLanahan 2009; Sassler 2010). These differences have broader implications for the ways in which economic factors and depressive symptoms affect individual and family outcomes.

Marital status may influence the association between economic hardship, depressive symptoms, and relationship distress; however, the hypothesized direction of the effects is equivocal. Although prior research has examined the association between economic hardship and depressive symptoms among couple dyads, these studies have focused largely on married couples (Conger, Conger, and Martin 2010). This is surprising given that cohabitors and married couples differ in terms of economic resources, mental health, and relationship quality and stability, and thus the effects may be stronger for cohabiting couples. Alternatively, these factors may affect married and cohabiting couples equally—resulting in no differences. A prior study addressing marital status differences on the effects of hardship and relationship quality found no differences between married and cohabiting couples (Halliday Hardie and Lucas 2010). Moreover, in a study of cohabiting couples, Wu and Pollard (2000) found that economic factors affected relationship instability. A finding that is also consistent with married couples. Thus, although there may be mean-level differences between groups, potential differences in the effects are less clear. Given that the direction of the effects is equivocal, no specific hypothesis is offered.

Differences Within Families: Mothers and Fathers

Gendered behaviors and expectations are often enacted in intimated partnerships, and the home environment acts as fertile ground by which gender is produced and reproduced (Ridgeway 2011). As a consequence, mothers and fathers often develop gender-specific roles, and not being able to carry out those roles may lead to feelings of frustration. Thus, the ways in which heterosexual parents respond to specific stressors are often gendered. For example, the Western notion of masculinity often associates exconomic providing with men (Bernard 1981; Christiansen and Palkovitz 2001); therefore, experiencing economic hardship is likely to be more detrimental to fathers than mothers, affecting both their mental health and relationship outcomes (Christiansen and Palkovitz 2001).

Alternatively, because mothers are more likely to engage in child care responsibilities compared to fathers (Bianchi and Milkie 2010), economic hardship may hinder mothers to effectively carry out such obligations, thus leading to higher levels of psychological and relationship distress. Indeed, prior studies have shown that economic hardship is associated with elevated levels of depressive symptoms for both partners (Conger et al. 2002; Tucker et al. 2010; Wickrama et al. 2010). Only one study, however, formally tested gendered differences and found no differences (Wickrama et al. 2010). Thus, more work is needed to address potential gender differences in the longitudinal and reciprocal effects of economic hardship and depressive symptoms that includes more diverse longitudinal samples, variations in relationship status of the romantic couple, and during an important transition such as a child's birth.

In addition, epidemiological studies document that depressive symptoms are more prevalent among women than men (Elliott 2001; Kessler 2003); however, such disparity reflects a broad range of issues such as power differences in heterosexual relationships (Nolen-Hoeksema, Larson, and Grayson 1999), differences in stress exposure (Thoits 2010), and other forms of gendered beliefs and practices (Rosenfield and Mouzon 2013). These differences have implications for the well-being of individuals and families. For example, women often provide the emotion work for families (Erickson 2005; Hochschild 1979), and feel responsible for managing the overall quality of familial relationships (Vogel and Karney 2002). Thus, mothers and father's depressive symptoms may make relationships unmanageable for mothers and thus lead to distress between partners. In contrast, men are more likely to withdraw in interpersonal relationships with high depression levels (Davila et al. 2003; Heavey, Layne, and Christensen 1993); therefore, fathers may want to end the relationship as a response to his own and his partner's depressive symptoms. Thus, the actor and partner effects of depressive symptoms may be divergent between mothers and fathers. Alternatively, poor mental health among parents with a young child may become equally dire for the quality of the relationship—resulting in no differences between parents. Understanding how mothers and fathers respond to their own and their partner's depressive symptoms shines light on

the gendered processes in heterosexual couples, and also delineate the effects of each parent's depressive symptoms on relationship distress. Although the effects may differ between mothers and fathers, the direction of the association is less clear. Thus, this portion of the study is exploratory and no specific hypothesis is offered.

Additional Factors

In order to account for potential spuriousness, the analyses take into account additional variables that are linked to economic hardship, depressive symptoms, and relationship distress. Thus, the current study employs the following control variables: parental age and education is associated with economic hardship and depressive symptoms (Miech and Shanahan 2000; Mirowsky and Ross 2001, 2002), and relationship quality (Umberson et al. 2005). Racial/ethnic minorities are more likely to be economically disadvantage (Lin and Harris 2008), African Americans tend to display lower levels of relationship quality (McLoyd et al. 2000) compared to whites, and Latinos and African Americans have lower rates of mental disorders in comparison to whites (Kessler 2005). Social support is related to economic, mental health, and relationship outcomes (Bradbury, Fincham, and Beach 2000; Thoits 2010; Umberson and Montez 2010). Fathers' incarceration history and domestic violence has adverse mental health, economic, and relationship consequences (Johnson and Ferraro 2000; Western, Lopoo, and McLanahan 2004). Parents' physical health is associated with both socioeconomic status and mental health (Gallo and Matthews 2003; Webb et al. 2008), and children's physical health is associated with parents' relationship quality (Reichman, Corman, and Noonan 2004). Studies are increasingly beginning to document the consequences of multipartnered fertility (i.e., having biological children with more than one partner). For example,

studies have documented that multipartnered fertility is associated with depression (Bronte-Tinkew, Horowitz, and Scott 2009; Turney and Carlson 2011) and parental relationships (Carlson and Furstenberg 2007; McLanahan 2009). The number of children parents have together is associated with both financial problems and depression (McLanahan and Adams 1987; Turney and Carlson 2011). Couples who experience a first birth are at an elevated risk for depression compared to couples who experience a higher-order birth (Mirowsky and Ross 2002). Poverty and employment status is associated with economic challenges, poor mental health, and relationship quality (Dooley et al. 2000; Iceland and Bauman 2007; McLanahan 2009; Thoits 2010). Accordingly, I include these control variables to adjust for spurious effects in the statistical analyses.

METHODS

Data

Data for this study are from the Fragile Families and Child Well-being Study (FFCW). The FFCW is a nationally representative, longitudinal study that follows an urban birth cohort of 4,898 children and their parents (3,712 unmarried and 1,186 married births) in 20 U.S. cities with populations of 200,000 or more. The study is based on a stratified, multistage probability sample with an oversample of unmarried births in urban cities. The study began 1998-2000 and contains 4,898 mothers and 3,830 fathers. At baseline, mothers were interviewed in person while in the hospital within 48 hours of the birth, and fathers were interviewed in person or by phone once he was located (for more detailed information, see (Reichman et al. 2001). Parents were re-interviewed when the child was one, three, and five years of age. The response rate for eligible mothers and fathers at baseline was 86% and 78%, respectively. Subsequent 1-, 3-, and 5-year follow-ups yielded 90%, 88%, and 87% response rates for eligible mothers, and 74%, 72%, and 70% for eligible fathers (Wellbeing 2008).

The sample includes couples (biological mothers and fathers of the focal child) who were living together (either married or cohabiting) at the baseline survey and participated in the subsequent surveys and has no missing values on the focal variables. All survey waves are used; however, the main analyses are based on the follow-up waves because the focal variables (i.e., economic hardship and depressive symptoms) were measured at the 1-, 3-, and 5-year waves only. The selection criteria of the sample for the data analyses are parents who were either married or cohabiting from the 1-year to the 3-year surveys, and from the 3-year to the 5-year follow-up waves. This is done in order to maintain couples who were romantically involved consistently over all survey waves.

As a result of the selection criteria, of the couples living together at the 1-year follow-up (N = 2, 341), 347 cases were dropped (15%) from the sample because either the mother or father was not interviewed at the 3-year survey, 317 cases (14%) were dropped because the parents ended their relationship between the 1- and 3-year surveys, and 24 cases (1%) were dropped because information on depressive symptoms or economic hardship were missing. This resulted in a sample of 1,653 (71%). For these couples who were living together at the 3-year follow-up, 239 cases (14%) were dropped because either mother or father were not interviewed at the 5-year follow-up, 199 cases (12%) because the relationship ended between the 3- and 5-year waves, and 21 cases (1%) because of missing data on economic hardship and depressive symptoms. This

resulted in a sample of 1,218 (74%) couples who were romantically involved and consistently living together over the 1-, 3-, and 5-year follow-up waves.

Further, the analyses are separated by marital status at the time of the baby's birth (baseline). There were some couples who were not married at the baseline but became married by the 1-year follow-up (N = 129). For these couples, they were included with the cohabiting couples because couples who marry after the birth of a child are more similar to cohabiting couple than married couples (McLanahan 2006). Taking this approach is in line with family policy agendas that are concerned about the relationship status of parents at the time of a child's birth. Notably, the substantive results do not change if couples who married subsequently after the birth of their child were included in the married sample.

In analyses of attrition, the excluded cases were more likely to be racial/ethnic minorities, somewhat younger, and had lower levels of education compared to parents who remain in the sample (see Tables 2.5 through 2.8). Comments about potential implications of attrition are in the Discussion section. To maximize sample size, multiple imputation technique was employed to impute missing data on the covariates, but not the focal endogenous variables for which Full Information Maximum Likelihood (FIML) was used (Acock 2005; Allison 2002; Enders and Bandalos 2001). Ten multiple imputation data sets were constructed using imputation by chained equations in Stata and then the analyses were conducted and combined using Rubin's rules (Rubin and Little 2002) in Mplus. City sampling weights are used for the descriptive statistics to adjust for oversampling of nonmarital births but not for the analyses because the study controls for

key characteristics associated with the weights (e.g., marital status at the birth of the child, age, race, and education; see (Winship and Radbill 1994).

Measures

Depressive Symptoms

At each wave were assessed using the Composite International Diagnostic Interview-Short Form for Major Depression (CIDI-SF), which is a comprehensive, standardized instrument used to assess the presence of mental disorders as specified by the Diagnostic and Statistical Manual of Mental Disorders (DSM-III-R; American Psychiatric Association, 1994). Respondents were asked the following stem questions: (a) "During the past 12 months, has there ever been a time when you felt sad, blue, or depressed for two or more weeks in a row? (b) "During the past 12 months, has there ever been a time lasting two weeks or more when you lost interest in most things like hobbies, work, or activities that usually give you pleasure?"

Respondents who affirmed these questions were asked about the following symptoms: (1) "losing interest," (2) "feeling tired," (3) "changes in weight," (4) "trouble sleeping," (5) "trouble concentrating," (6) "feeling down," and (7) "thoughts about death." Each symptom is a dichotomous variable with the value of 1 indicating the presence of a symptom. For mothers, α reliability scores at the 1-, 3-, and 5-year follow-up waves were .90, .86, and .93, respectively. For fathers, α reliability scores at the 1-, 3-, and 5-year follow-up waves were .82, .86, and .86, respectively. The items were summed creating a range from 0 to 8 symptoms. Previous studies using the FFCW study have assessed depressive symptoms for both mothers and fathers (Bronte-Tinkew et al. 2009; Heflin and Iceland 2009).

Economic Hardship

At each wave economic hardship was measured by mothers responses to seven dichotomous indicators of whether they (1) "received free meals," (2) "had trouble paying rent or mortgage," (3) "had trouble paying gas/electric bill," (4) "borrowed money from friends or family to pay bills," (5) "been evicted," (6) "moved in with relatives," and (7) "someone needed a doctor but couldn't go." Reliability estimates across waves were .59, .62, and .62, respectively. The items were summed creating a range from 0 to 8 number of economic hardship. Despite the modest reliability, this measure is commonly used in national surveys (Beverly 2001). Since the items for depressive symptoms and economic hardship are dichotomous, Kuder-Richardson coefficient of reliability was calculated.

Relationship Distress

At the 5-year survey both parents relationship distress (mother's $\alpha = .84$, father's $\alpha = .79$) was measured using three trichotomous items (1=never, 3=often) that asked mothers and fathers (1) "how often they thought their relationship might be in trouble," (2) "discussed ending the relationship with their partner," and (3) "talked to a close friend about a break-up." The items were averaged, which indicates that higher scores reflect higher levels of relationship distress. As stated previously, prior studies employing similar items have focused on married couples and used the term marital distress (Booth et al. 1983; Conger et al. 1999). The current study, however, refers to these items as relationship distress because the sample consists of both married and cohabitating couples.

Control Variables

Control variables were included in all statistical models (described above) that are expected to be associated with economic hardship, depressive symptoms, and relationship distress. Identical measures were used for both mothers and fathers. Each control variable is represented for both mothers and fathers at the baseline survey (unless otherwise specified). Mothers and fathers' age was measured (in years) as continuous variables. Mother's race was measured with the following dummy variables: *White* (reference), *Black*, *Hispanic*, and *other*. A separate dummy variable was included to represent parents who differed on race/ethnicity (i.e., mixed race couple). Mothers and fathers' education level was measured using four categories: (1) less than high school (reference), (2) high school or equivalent, (3) some college or tech training, and (4) college graduate or more. Physical health (measured at the 1-year follow-up for parents and child) was measured by asking parents the following question: "In general, how is your health?" Mothers reported child's health with responses ranging from (1) poor to (5) excellent. *Employment* status was measured with a dichotomous item indicating whether each parent "Did any regular work for pay last week?" Response were (0) no and (1) yes.

Social support was measured with a dichotomous question (0=no, 1=yes) asking both parents "since child was born, have you received any financial help or money from anyone other than [partner]?" Mothers reported on fathers' *incarceration history* (at the 1-year follow-up) indicating whether fathers have ever been in jail or prison (0=no, 1=yes). Mothers also reported on *domestic violence* (at the 1-year follow-up) indicating were she ever seriously hurt in fight with father (0=no, 1=yes). Mothers reported the number of children in the household at the 1-year follow-up. Parents' fertility history was gauged with two separated measures: First, a measured was created to indicate whether the focal child is a higher order birth or first birth (0=first birth, 1=higher order birth). Second, a measured was created to indicate a series of dummy variables indicating *multipartnered fertility* (at 1-year follow-up) which reflects whether mothers and fathers reported having a child with another partner: neither parent has a child by another partner (reference), father has child by another partner only, mother has child by another partner only, and both parents has a child by another partner. Poverty status (at the 1-year follow-up) was measured using the household income-to-needs ratio based on the official U.S. poverty thresholds from the Census Bureau (adjusted for household composition and year). The variable was dichotomized to indicate that a ratio of 1 or less reflects a family lived in poverty, and a ratio above 1 reflects a family lived above the poverty line (, 0=no poverty, 1=poverty). *Marital status* was measured by a dichotomous variable with (0) indicating married (reference group) and (1) indicating cohabitation.

Analytic Strategy

Structural equation modeling (SEM; Bollen 1989) was employed using Mplus 6.11 (Muthen and Muthen 2010) to estimate the hypothesized paths in Figures 2.1 and 2.2. SEM is a useful statistical technique for handling dyadic data because it allows longitudinal and reciprocal paths between partners to be estimated over time (Kenny, Kashy, and Cook 2006). For the longitudinal and reciprocal effects of economic hardship and depressive symptoms (Figure 2.1), the model was estimated using Negative Binomial Regression with Maximum Likelihood Estimation with robust standard errors. This was done to take into account the over-dispersion of both economic hardship and depressive symptoms as count variables. The parameter estimates are exponentiated, and thus the interpretation is as follows: a unit change in the independent variable is associated with a multiplicative change in the expected count in the dependent variable.

Second, for the relationship distress analysis (Figure 2.2), the model was estimated using standard regression with Full Information Maximum Likelihood (FIML) estimation technique. The analyses were executed using an Actor-Partner Interdependent Model (APIM; Kenny et al. 2006) with distinguishable dyads (i.e., mothers and fathers) to examine the direct effect of each parent's own depressive symptoms on their own relationship distress, and the effect of each parent's depressive symptoms on their partner's relationship distress. The variables are standardized, and thus reflect the following interpretation: a one standard deviation change in the independent variable is associated with a standard deviation change in the endogenous variable.

To address whether the effects differ between families (married and cohabiting) and within families (mothers and fathers), the longitudinal and cross-sectional models were both re-estimated by marital status using multi-group SEM. For the longitudinal model, the standard difference test (i.e., evaluating model fit by calculating an x^2 difference between an unconstrained model with the paths estimated freely compared to a constrained model with the paths equal across groups) could not be achieved because models using Maximum Likelihood Estimation with robust standard errors with imputed data do not yield a correction factor, which makes χ^2 difference test impossible. Rather, to test differences between groups, the parameter estimates were tested using t-test parameter constraints. This allows the estimates for married mothers to be tested against the estimates for cohabiting mothers, and likewise for married and cohabiting fathers. To explore gender differences within families (mothers and fathers), the parameters were constrained to test whether the effects are significantly different between mothers and fathers using t-tests of parameter constraints. The parameter for mothers is subtracted from the parameter for fathers (e.g., $b_{(fathers)} - b_{(mothers)}$) producing a new parameter (e.g., $b_{(fathers-mothers)}$). A standard error is estimated in M*plus* for hypothesis testing.

RESULTS

Descriptive Statistics

Table 2.1 presents the mean, percentages, and standard deviations for the demographic characteristics for mothers and fathers by marital status (weighted using the city sampling weights). Married couples were slightly older, more likely to be non-Hispanic white, have higher levels of education, more likely to be employed, and less likely to be living in poverty. Cohabitors received more social support, cohabiting fathers were more likely to have a child by another partner. All parents were in good health.

<Table 2.1 about here>

Table 2.2 shows the means for economic hardship, depressive symptoms, and relationship distress (weighted using city sampling weights), and significant differences by marital status evaluated using two-group mean comparison t tests to test differences between groups (e.g., married mothers versus cohabiting mothers, etc.), and two-sample (paired) mean comparison t tests to test differences within groups (e.g., married mothers versus married fathers). In regards to between families, the results show that, on average, cohabiting couples experience higher levels of economic hardship at the 1-, 3-, and 5- year surveys than married couples, and the means were statistically different at the Year-3 and Year-5 survey years.

In addition, cohabiting mothers had higher levels of depressive symptoms compared to married mothers, and the means were significantly different at the Year-1 and Year-3 survey waves. For fathers, cohabitors and marrieds displayed similar levels of depressive symptoms across survey years and there were no significant differences. Cohabiting mothers and fathers displayed higher levels of relationship distress compared to married mothers and fathers, respectively. Significant differences only emerged between married and cohabiting mothers in relationship distress.

In regards to differences within families, the two-sample (paired) mean comparison tests reveal that married mothers have significant higher levels of depressive symptoms at the Year-3 survey wave compared to married fathers. No differences emerged within cohabiting couples for depressive symptoms. For relationship distress, married and cohabiting mothers have significant higher mean levels compared to married and cohabiting fathers, respectively. No differences were present for married or cohabiting couples in regards to relationship distress.

<Tables 2.2 about here>

Structural Equation Models

All Couples

The first research question was whether there was a significant longitudinal and reciprocal association between economic hardship and depressive symptoms for couples after the birth of a child. To address this question, the analyses were executed using Negative Binomial Regression to model the outcomes in a structural equation modeling framework. The results are shown in Table 2.3. For all couples, in respect to Years 1 to 3, the results for the *direct effects* show that as the number of hardships increase, depressive

symptoms for mothers increase by a factor of 1.27 (or increase by 27%; p < .01). For Years 3 to 5, a similar effect was revealed, with depressive symptoms increasing by a factor of 1.21 (p < .01). For fathers, in respect to Years 1 to 3, the results show that when the number of hardship increases, there is a 22% (p < .01) increase in the count of depressive symptoms. For Years 3 to 5, the effects were not significant (exp(b) = 1.04, p > .10). All in all, the findings support the stress-process perspective (*i.e., direct effects*) for mothers, and give partial support for fathers.

The reciprocal pathways between depressive symptoms and economic hardship were also examined, and the results are presented in Table 2.3. For all couples, in respect to Years 1 to 3, the results for the *reciprocal effects* show that when the number of depressive symptoms increase, hardships increase by a factor of 1.09 (p < .01). In respect to Years 3 to 5, the effects were not significant. For fathers, the reciprocal association between depressive symptoms and economic hardship did not yield significant results from Years 1 to 3 or Years 3 to 5. Although the results partially support the *reciprocal effects* for fathers. In other words, there is a small effect of maternal mental health on hardship shortly after the birth of a child. Overall, the longitudinal and reciprocal effects show that economic hardship affects both parents' mental health more so than vice versa.

The third research hypothesis suggests that economic hardship and depressive symptoms are both associated with lower levels of relationship distress for couples by their child fifth birthday. To test this hypothesis, I relied on cross-sectional data (Year-5 only) using an Actor-Partner Interdependent Model (APIM). As shown in Table 2.4, the findings show that for every standard deviation increase in economic hardship, relationship distress increases by .20 standard deviations (p < .001) for mothers, and increases .06 standard deviations (p < .05) for fathers. There were significant *actor effects* for mothers and fathers: that is, mother's depressive symptoms was associated with a .06 standard deviation increase in her own report of relationship distress; for fathers, depressive symptoms were associated with a .10 standard deviations in his own report of relationship distress. There is also some evidence for *cross-partner effects*: for every standard deviation increase in father's depressive symptoms, mother's relationship distress increase by .05 standard deviations; the cross-partner association from mother's depressive symptoms to father's relationship distress was marginal statistical significant (b=.01, p < .10). These findings support the research hypothesis that both economic hardship and depressive symptoms have a direct and significant association with relationship distress for mothers and fathers five years after the birth of a child. Moreover, although the effects are small, the findings give some support for the crosspartner association between one partner's depressive symptoms and the other partner's relationship distress. Simply put, both economic hardship and depressive symptoms are associated with dyadic partners' contemplating ending the relationship in the years following the birth of a child.

Differences Between Families: Married and Cohabiting

Next, Table 2.3 also presents the longitudinal and reciprocal effects for both married and cohabiting couples. For married mothers, when the number of hardships increase, depressive symptoms increase for Years 1 to 3 by a factor of 1.34 (or 34%, p < .01); and, for Years 3 to 5 the count of depressive symptoms increase by a factor of 1.60 (p < .01). For cohabiting mothers, the association between economic hardship and depressive

symptoms from Years 1 to 3 were statistically significant ($\exp(b) = 1.30$, p < .01), and marginally significant for Years 3 to 5 ($\exp(b) = 1.15$, p < .10). Differences emerged between married and cohabiting mothers in the effects from Years 3 to 5 only ($t_{diff} = 2.11$, p < .05), indicating that the effects are stronger for married mothers than cohabiting mothers as noted with subscript a. For fathers, concerning Year-1 to Year-3, the effect of economic hardship on depressive symptoms were only significant for cohabitors (exp(b) = 1.22, p < .05); whereas for Years 3 to 5, the effects were only significant for married fathers ($\exp(b) = 1.39$, p < .05). There were no significant differences between married and cohabiting fathers. In regards to the reciprocal effects for mothers, the number of depressive symptoms was associated with a increase in the count of economic hardship by 16% (p < .01) for married mothers and 7% (p < .05) for cohabiting mothers. The effects were not statistically different for mothers, however. The reciprocal effects were not significant for fathers, and no significant differences emerged between married and cohabiting fathers. All in all, the only differences that emerge were between married and cohabiting mothers in the association between hardship and depressive symptoms during the later years (i.e., Year-3 to Year-5).

Turning to the relationship distress model (Table 2.4), the effects of economic hardship and depressive symptoms on relationship distress were also examined separately by marital status. For married couples, economic hardship was significantly associated with higher levels of relationship distress for mothers (b=.25, p < .001) and for fathers (b=.11, p < .05). For cohabiting couples, economic hardship was associated with higher levels of relationship distress for mothers (b=.18, p < .001) but not for fathers (b=.03, p > .10). In regards to the actor effects, married mother's depressive symptoms were

associated with her own relationship distress (b=.09, p < .05), and married father's depressive symptoms were associated with his own relationship distress (.18, p < .001). Similarly, the effects were significant for cohabiting mothers (b=.19, p < .001) and fathers (b=.31, p < .001). In the end, the results demonstrate that economic hardship and each parent's own depressive symptoms were associated with relationship distress for married couples; for cohabiting couples, hardship and depressive symptoms were related to relationship distress for mothers, whereas only cohabiting fathers' own depressive symptoms mattered for relationship distress.

For the partner effects, married mother's depressive symptoms had a marginal association with father's relationship distress (b = .07, p < .10) but married father's depressive symptoms had no significant effect on married mother's relationship distress. For cohabiting couples, a standard deviation increase in father's depressive symptoms was associated with a .14 standard deviation (p < .001) for mother's relationship distress; the partner effect from mother's depressive symptoms to father's relationship distress was not significant. The differences between married and cohabiting couples were examined by comparing the fit of an unconstrained model with a constrained model by calculating the difference in x^2 which test whether the groups are significantly different. The x^2 difference test was not statistically significant (p = .33), thus indicating that the effects were not significantly different between married and cohabiting couples.

Differences Within Families: Mothers and Fathers

Last, differences between mothers and fathers were also tested. In regards to the longitudinal and reciprocal association between economic hardship and depressive symptoms, no gender differences emerged within married or cohabiting families. The distress model results, however, revealed similar gender differences within married and cohabiting relationships (see Table 2.4). For example, there were differences between mother and fathers in the effect of hardship on relationship distress for married couples ($t_{diff} = 4.16$, p < .001) as noted with superscript a, and cohabiting couples ($t_{diff} = 3.37$, p < .01) with the effect stronger for fathers than mothers (see superscript b). Gender differences also emerged with the effect of mother's and father's depressive symptoms on father's relationship distress for married couples ($t_{diff} = -2.12$, p < .05) as noted with superscript c, and cohabiting couples ($t_{diff} = -5.00$, p < .001) noted with superscript d, with the effects stronger for mother's depressive symptom than father's depressive symptoms. In summary, the direct effect of economic hardship on relationship distress was stronger for fathers than mothers; and, mother's depressive symptoms on father's relationship distress (partner effect) were stronger than father's own depressive symptoms (actor effects).

DISCUSSION

The purpose of the present study was (1) to examine the longitudinal and reciprocal effects of economic hardship and depressive symptoms for couples 1-, 3-, and 5-years after the birth of a child; (2) to examine whether economic hardship and depressive symptoms lead to higher levels of relationship distress for couples by their child's fifth birthday; (3) to test whether the effects vary between families (married and cohabitors), and (4) within families (mothers and fathers). Using data from a diverse nationally representative sample of urban couples after the birth of a child, the current study reveals how inequality, health, and relationship quality unfold and are associated over time for families, and the potential implications for children's well-being.

The longitudinal and reciprocal effects of economic hardship and depressive symptoms show support for direct effects (for both parents) and significant reciprocal effects for mothers only. More specifically, the findings are consistent with the *stress process perspective* for parents with a young child, which corroborate previous research that examined the association between economic hardship and depressive symptoms (Barnett 2008; Conger, Conger, and Martin 2010). The consistent positive and significant effects for mothers reflects the added stress and strain that financial hard times place on families (Conger 2010; Edin and Kissane 2010). These findings are particularly important during a child's early developmental stages (i.e., infant, toddler, and preschool) as children of depressed mothers tend to experience a host of adverse outcomes (e.g., Ramchandani et al. 2008).

Because mothers are much more likely to engage in early childcare activities than fathers (Bianchi and Milkie 2010), not having adequate resources to carry out such obligations put mothers at risk for lower levels of psychological well-being. For fathers, however, the effects are only significant from Years 1 to 3. The short-lived significant effects may be due, in part, to the different ways fathers manifest economic stress, especially as hardships continue over time. For example, men tend to respond to stress with negative interpersonal behavior (i.e., conflict, etc.). In fact, prior studies using the Fragile Families' data show that economic hardship leads to higher levels of discord for fathers (Paat 2011; Williams, Cheadle, and Goosby 2013).

In addition, the findings give some support for the reciprocal effects, but the findings were only significant for mothers from Years 1 to 3. These results suggest that higher levels of depressive symptoms 1-year after a child's birth may decrease mothers'

financial contributions to the household which, in turn, leads to higher levels of economic hardship. Given the prevalence of depressive symptoms for mothers after a child's birth (Umberson et al. 2010), the significant findings from Years 1 to 3 is not surprising, although the non-significant results from Years 3 to 5 are somewhat surprising. This may indicate that the effects of mother's depressive symptoms are less damaging to the family's economic well-being over time. For fathers, no reciprocal effects between depressive symptoms and economic hardship emerge at either survey years. This may be indicative of the lower levels of father's depressive symptoms may be more damaging to parents' mental health than parents' mental health affecting economic hardship.

In the analysis for relationship distress, the results give support for family stress theories. More specifically, economic hardship had a significant direct effect on relationship distress for mothers and fathers five years after the birth of a child. Given that the sample in this study includes both cohabiting and married mothers, these findings are consistent with other studies showing that economic hardship plays a major role in marital distress (Gudmunson et al. 2007) and a family's finances play a more critical role for unmarried, low-income couples than married higher income couples (Gibson-Davis, Edin, and McLanahan 2005). Moreover, the findings also indicate that mothers' and fathers' depressive symptoms are associated with higher levels of relationship distress, which points to the importance of mental health and the quality of intimate relationships for sustaining romantic partnerships. Thus, similar to previous research, this study shows that depressive symptoms affect the quality of intimate partnerships for both parents (Conger, Conger, and Martin 2010; Kouros and Cummings 2011), while also elaborating on these prior findings by showing that mental health is a strong antecedent that affects relationship stability five years after the birth of a child—which is a critical time for both couple's stability (Cherlin 2009b, 2010b) and child development (Cheadle 2008). Thus, these results have broader implications not only for couples but also children's well-being (McLanahan 2004b).

Family scholars also suggest that one partner's mood can lead to how the other partner evaluates the overall quality of the relationship (i.e., *emotional crossover*). In the current study, the results are consistent with previous studies highlighting emotional crossover in couple dyads (Thompson and Bolger 1999; Yorgason et al. 2006). The findings show that each parent's depressive symptoms are associated with their partner's relationship distress—though the effects are marginally significant for mother's depressive symptoms and father's relationship distress. The overall conclusion is that dyadic outcomes such as parents' relationship distress are not only sensitive to family inequality (i.e., economic hardship) but also interpersonal stressors (i.e., partner's depressive symptoms). These findings corroborates Milkie's (2010) argument that incorporating the stress process and family stress theories can be fruitful in understanding how stress affects both individuals and families. Although relationship distress can further exacerbate depressive symptoms within couples, the current study is unable to determine if this is the case due to data limitations. Nevertheless, these findings are especially important in light of family policies that promote marriage through strengthening relationships.

With respect to differences between married and cohabiting couples, the findings show that the longitudinal and reciprocal effects of economic hardship and depressive

symptoms were similar for both family types. The only significant difference emerges over Years 3 to 5 for mothers: that is, the effect of hardship on depressive symptoms was stronger for married mothers than cohabiting mothers. This suggests that hardship is more detrimental for married mothers' mental health, which may be largely due to the expectations that marriage provides financial security (Waller and McLanahan 2005). Thus, experiencing financial difficulty may contradict their expectations, and subsequently lead to psychological distress. Moreover, the relationship distress results yield no significant differences between married and cohabiting couples. These finding are surprising given the differences between married and cohabiting couples in terms of economic, psychological, and relationship well-being (Smock 2000). In the current study, married couples, compared to cohabiting couples, have fewer hardships, depressive symptoms, and lower levels of relationship distress. Yet, the longitudinal and crosssectional results yield no significant differences between groups. Similar finding from a study comparing the effects of economic hardship and relationship quality between married and cohabitors found no differences (Hardie and Lucas 2010). Overall, the findings suggest that although differences emerge between married and cohabiting couples at the mean level, both family forms, after a recent birth, are vulnerable to economic and psychological adversity.

Turning to differences within families (mothers and fathers), the longitudinal results reveal no differences between mothers and fathers among couples in married or cohabiting relationships. In the analysis for relationship distress, however, differences did emerge; and, the differences are similar for mothers and fathers in both married and cohabiting families. For example, the effect of economic hardship on parents' relationship distress was significantly different between mothers and fathers for both married and cohabiting families, with the effects stronger for fathers than mothers. Also, there were significant differences between mother's depressive symptoms and father's depressive symptoms on father's relationship distress, with the effects stronger for mothers than fathers. These differences between mothers and fathers map onto gender perspectives in families—at least for fathers. That is, hardship infringes on the provider role (Christiansen and Palkovitz 2001), and mother's mental health may lead fathers to withdraw (Davila et al. 2003) in interpersonal relationships, thus creating distress between parents.

Implications

In terms of implications for research, the results help to shine light on the need for additional studies to take a more integrated theoretical approach to capture broader individual and family processes. Indeed, scholars have consistently argued for this approach (Milkie, Bierman, and Schieman 2008; Milkie 2010). In addition, the dyadic findings highlight the importance of families as a system (Cox and Paley 1997), and is consistent with the notion of "linked lives" in life course research, which suggest that the lives of family members are interdependent (Elder, Johnson, and Crosnoe 2003). An important topic for future research is to examine the association between hardship, mental health, and relationship distress over longer periods of time, and whether these factors affect child outcomes, especially as children become young adults. In addition, this line of research could move forward by addressing other factors that may moderate the association of the key variables in the study, including race/ethnicity, education levels, and the number of children—as these factors (and others) may exacerbate economic hardship, depressive symptoms, and relationship distress.

These findings are also important in light of family policies that are designed to promote healthy marriage and other partnerships through strengthening relationships. The results provide some evidence that policies should not only focus on building relationship skills but also ameliorating structural conditions (i.e., economic hardship) and psychological distress. This is critical given the rise in poverty and unemployment, and the decrease in household income from 2009 to 2010 (DeNavas-Walt, Proctor, and Smith 2011). Thus, policy efforts that are designed to strengthen couples' relationships should also work in tandem with policies that target the economy, family economic circumstances, and mental health services.

Limitations

Although this study demonstrates that economic hardship and depressive symptoms affect individuals and families, some limitations must be noted that may hinder the generalizability of the results. First, the results for the current study can only be generalized to parents who had a child in the late 1990s while living in urban cities with a population of 200,000 or more. Thus, inferences cannot be made to populations outside the scope of the initial research design (e.g., parents in rural areas, or childless parents). Second, as with many studies, there is potential for missing variable bias. Even in longitudinal studies, it is difficult to take into account changes that may have occurred between surveys. To help reduce missing variable bias, control variables were added to the statistical models that were theoretically meaningful that might affect the endogenous variables in the study (see Tables 2.9 through 2.12). Last, couples who ended their

relationship between waves tend to experience higher levels of depressive symptoms and economic hardship. Thus, the estimates may be downwardly biased. This issue, however, may be most pertinent for cohabiting couples as most married couples stayed together over time.

Strengths

Despite the limitations, the results in the current study extend research on the links between economic hardship, depressive symptoms, and relationship distress in several ways. First, prior research has focused largely on small, cross-sectional samples; however, recent research is beginning to use longitudinal data (Wickrama et al. 2010, 2012)—albeit with homogenous samples (i.e., predominately white, middle-aged, married couples). Here, however, the current study used a diverse sample of couples and examined economic hardship and depressive symptoms as a dynamic and dyadic process. In doing so, the current study leveraged data from both partners rather than being limited to only one (usually the mother). The second extension is a focus on married and cohabiting heterosexual couples. This allows for the examination of between and within family processes. Given the unequal distribution of resources between families and the differential stress exposure between parents, the current approach paints a more holistic picture about inequality, stress, and well-being in couple dyads than can be found in most traditional studies that focus only one partner (usually the mother).

Third, one strength of the FFCW study is that it follows a cohort of recent births and their parents across critical developmental stages for children—infant, toddler, and preschool. Thus, understanding how families fare under dire economic conditions is particularly important because of the detrimental effects these factors have on children's well-being (Amato and Cheadle 2008; Zilanawala and Pilkauskas 2012), and parent depression affects other family processes such as parenting (Paulson, Dauber, and Leiferman 2006). Fourth, the current study addresses whether familial and individual stressors affect a dyadic outcome such as relationship distress five years after the birth of a child. This is particularly important as many couples are at risk for dissolution by the child's fifth birthday (Cherlin 2010b), and parent's dissolution can be transferred intergenerational by creating further inequality for their children (Amato and Cheadle 2005). Thus, the aforementioned stressors are not only detrimental to couples, but also can extend beyond the couple dyad by "spilling over" to children.

CONCLUSION

In conclusion, the current study focused on urban parents who had a child in the late 1990s to underscore how inequality, mental health, and relationship distress unfold within couple dyads. This research focused on parents with young children because the birth of a child is a stressful time economically (Demo and Cox 2000) and it can put strain on parent's health and intimate relationships (Umberson et al. 2005). Thus, the results in the present study is particularly valuable because the early childhood years set the stage for children's successful academic (Cheadle 2008) and behavioral development (Bradley et al. 2001), and long-term educational attainment outcomes (Cheadle and Goosby 2010). Notably, healthy parents (Ramchandani et al. 2008) and healthy relationships between parents (Amato 2010) support these processes. Using both longitudinal and cross-sectional data, the current study presents evidence that economic hardship affects mothers and father's mental health, with more enduring negative effects for mothers than fathers. Moreover, hardship and mental health creates distress between intimate partners, whereby both parents are contemplating to end the relationship. Taken together, the results highlight the significance of examining changes in mental health as a function of economic hard times; and, the extent to which family-level and interpersonallevel stressors affect couple dyads.
REFERENCES

- Acock, Alan C. 2005. "Working with Missing Values." *Journal of Marriage and Family* 67(4):1012–28.
- Allison, P. D. 2002. "Missing Data: Quantitative Applications in the Social Sciences." British Journal of Mathematical and Statistical Psychology 55(1):193–96.
- Amato, Paul R. 2010. "Research on Divorce: Continuing Trends and New Developments." *Journal of Marriage & Family* 72(3):650–66.
- Amato, Paul R., and Jacob Cheadle. 2005. "The Long Reach of Divorce: Divorce and Child Well-Being across Three Generations." *Journal of Marriage and Family* 67(1):191–206.
- Amato, Paul R., and Jacob E. Cheadle. 2008. "Parental Divorce, Marital Conflict and Children's Behavior Problems: A Comparison of Adopted and Biological Children." Social Forces 86(3):1139–61.
- Aneshensel, Carol S., Jo C. Phelan, and Alex Bierman. 2013. "The Sociology of Mental Health: Surveying the Field." Pp. 1–19 in *Handbook of the Sociology of Mental Health*. Springer.
- Avison, W. R., and R. J. Turner. 1988. "Stressful Life Events and Depressive Symptoms: Disaggregating the Effects of Acute Stressors and Chronic Strains." *Journal of Health and Social Behavior* 253–64.
- Barnett, Melissa. 2008. "Economic Disadvantage in Complex Family Systems: Expansion of Family Stress Models." *Clinical Child & Family Psychology Review* 11(3):145–61.
- Bendheim-Thomas Center on Child Wellbeing. 2008. Introduction to the Fragile Families Public Use Data. Princeton University.
- Benson, Michael L., Greer L. Fox, Alfred DeMaris, and Judy Van Wyk. 2003. "Neighborhood Disadvantage, Individual Economic Distress and Violence against Women in Intimate Relationships." *Journal of Quantitative Criminology* 19(3):207–35.
- Benzies, Karen M., Margaret J. Harrison, and Joyce Magill-Evans. 2004. "Parenting Stress, Marital Quality, and Child Behavior Problems at Age 7 Years." *Public Health Nursing* 21(2):111–21.
- Bernard, Jessie. 1981. "The Good-Provider Role: Its Rise and Fall." *American Psychologist* 36(1):1–12.

- Beverly, Sondra G. 2001. "Material Hardship in the United States: Evidence from the Survey of Income and Program Participation." *Social Work Research* 25(3):143.
- Bianchi, Suzanne M., and Melissa A. Milkie. 2010. "Work and Family Research in the First Decade of the 21st Century." *Journal of Marriage and Family* 72(3):705–25.
- Bolger, Niall, Anita DeLongis, Ronald C. Kessler, and Elaine Wethington. 1989. "The Contagion of Stress across Multiple Roles." *Journal of Marriage and the Family* 175–83.
- Bollen, Kenneth A. 1989. Structural Equations with Latent Variable. New York: Wiley.
- Booth, Alan, David Johnson, and J. N. Edwards. 1983. "Measuring Marital Stability." *Journal of Marriage and Family* 45(2):387–94.
- Bradbury, Thomas N., Frank D. Fincham, and Steven R. H. Beach. 2000. "Research on the Nature and Determinants of Marital Satisfaction: A Decade in Review." *Journal of Marriage and Family* 62(4):964–80.
- Bradley, Robert H., Robert F. Corwyn, Margaret Burchinal, Harriette Pipes McAdoo, and Cynthia García Coll. 2001. "The Home Environments of Children in the United States Part II: Relations with Behavioral Development through Age Thirteen." *Child Development* 72(6):1868–86.
- Breese McCoy, Sarah J. 2011. "Postpartum Depression: An Essential Overview for the Practitioner." *Southern Medical Journal* 104(2):128–132
- Bronte-Tinkew, J., A. Horowitz, and M. E. Scott. 2009. "Fathering With Multiple Partners: Links to Children's Well-Being in Early Childhood." *Journal of Marriage and Family* 71(3):608–31.
- Bronte-Tinkew, Jacinta, Kristin A. Moore, Gregory Matthews, and Jennifer Carrano. 2007. "Symptoms of Major Depression in a Sample of Fathers of Infants: Sociodemographic Correlates and Links to Father Involvement." *Journal of Family Issues* 28(1):61–99.
- Brown, Susan L. 2000. "The Effect of Union Type on Psychological Well-Being: Depression among Cohabitors versus Marrieds." *Journal of Health and Social Behavior* 41:241–55.
- Brown, Susan L. 2010. "Marriage and Child Well-Being: Research and Policy Perspectives." *Journal of Marriage and Family* 72(5):1059–77.
- Brown, Susan L., and Alan Booth. 1996. "Cohabitation versus Marriage: A Comparison of Relationship Quality." *Journal of Marriage and the Family* 668–78.

- Bulanda, Jennifer Roebuck, and Susan L. Brown. 2007. "Race-Ethnic Differences in Marital Quality and Divorce." *Social Science Research* 36(3):945–67.
- Bumpass, Larry L., and Hsien-Hen Lu. 2000. "Trends in Cohabitation and Implications for Children's Family Contexts in the United States." *Population Studies* 54(1):29–41.
- Butterworth, Peter, Liana S. Leach, Jane Pirkis, and Margaret Kelaher. 2012. "Poor Mental Health Influences Risk and Duration of Unemployment: A Prospective Study." *Social Psychiatry and Psychiatric Epidemiology* 47(6):1013–21.
- Carlson, Marcia J., and Frank Furstenberg. 2007. "The Consequences of Multi-Partnered Fertility for Parental Involvement and Relationships." *Princeton, NJ: Bendheim-Thomas Center for Research on Child Wellbeing*.
- Cheadle, Jacob E. 2008. "Parent Educational Investment and Children's General Knowledge Development." *Social Science Research* 38(2):477–91.
- Cheadle, Jacob E., and Bridget J. Goosby. 2010. "Birth Weight, Cognitive Development, and Life Chances: A Comparison of Siblings from Childhood into Early Adulthood." *Social Science Research* 39(4):570–84.
- Cherlin, Andrew. 2009. The Marriage-Go-Round. NY: Vintage Books.
- Cherlin, Andrew. 2010. "Demographic Trends in the United States: A Review of Research in the 2000s." *Journal of Marriage and Family* 72(3):403–19.
- Christiansen, Shawn L., and Rob Palkovitz. 2001. "Why the 'Good Provider' Role Still Matters Providing as a Form of Paternal Involvement." *Journal of Family Issues* 22(1):84–106.
- Collins, Randall. 2004. Interaction Ritual Chains. Princeton, NJ: Princeton Univ. Press.
- Conger, Rand D. et al. 1990. "Linking Economic Hardship to Marital Quality and Instability." *Journal of Marriage and the Family* 52(3):643–56.
- Conger, Rand D. et al. 2002. "Economic Pressure in African American Families: A Replication and Extension of the Family Stress Model." *Developmental Psychology* 38(2):179–93.
- Conger, Rand D., Katherine J. Conger, and Monica J. Martin. 2010. "Socioeconomic Status, Family Processes, and Individual Development." *Journal of Marriage and Family* 72(3):685–704.
- Conger, Rand D., Katherine J. Conger, and Monica J. Martin. 2010. "Socioeconomic Status, Family Processes, and Individual Development." *Journal of Marriage and Family* 72(3):685–704.

- Conger, Rand D., Martha A. Rueter, and Glen H. Elder Jr. 1999. "Couple Resilience to Economic Pressure." *Journal of Personality and Social Psychology* 76(1):54–71.
- Cooper, Carey E., Sara S. McLanahan, Sarah O. Meadows, and Jeanne Brooks-Gunn. 2009. "Family Structure Transitions and Maternal Parenting Stress." *Journal of Marriage and Family* 71(3):558–74.
- Copen, C., Kimberly Daniels, and William D. Mosher. 2013. "First Premarital Cohabitation in the United States: 2006–2010 National Survey of Family Growth." *National Health Statistics Report* 64.
- Cowan, P., and C. Cowan. 2008. "Diverging Family Policies to Promote Children's Well-Being in the UK and US: Some Relevant Data from Family Research and Intervention Studies." *Journal of Children's Services* 3(4):4–16.
- Cox, Martha J., and Blair Paley. 1997. "Families as Systems." Annual Review of Psychology 48(1):243–67.
- Coyne, James C. 1976. "Depression and the Response of Others." *Journal of Abnormal Psychology* 85(2):186.
- Crnic, Keith A., Catherine Gaze, and Casey Hoffman. 2005. "Cummulative Parenting Stress Across the Preschool Period: Relations to Maternal Parenting and Child Behavior at Age 5." *Infant and Child Development* 14:117–32.
- Cui, Ming, M. Brent Donnellan, and Rand D. Conger. 2007. "Reciprocal Influences between Parents' Marital Problems and Adolescent Internalizing and Externalizing Behavior." *Developmental Psychology* 43(6):1544.
- Cunradi, Carol B., Raul Caetano, Catherine Clark, and John Schafer. 2000.
 "Neighborhood Poverty as a Predictor of Intimate Partner Violence Among White, Black, and Hispanic Couples in the United States: A Multilevel Analysis." Annals of Epidemiology 10(5):297–308.
- Cutrona, C. E. et al. 2003. "Neighborhood Context and Financial Strain as Predictors of Marital Interaction and Marital Quality in African American Couples." *Personal Relationships* 10(3):389–409.
- Cutrona, Carolyn E. 1996. Social Support in Couples: Marriage as a Resource in Times of Stress. Sage Publications, Inc.
- Davila, Joanne, Benjamin R. Karney, Todd W. Hall, and Thomas N. Bradbury. 2003. "Depressive Symptoms and Marital Satisfaction: Within-Subject Associations and the Moderating Effects of Gender and Neuroticism." *Journal of Family Psychology* 17(4):557.

- DeMaris, Alfred, Michael L. Benson, Greer L. Fox, Terrence Hill, and Judy Van Wyk. 2003. "Distal and Proximal Factors in Domestic Violence: A Test of an Integrated Model." *Journal of Marriage and Family* 65(3):652–67.
- Demo, David H., and Martha J. Cox. 2000. "Families with Young Children: A Review of Research in the 1990s." *Journal of Marriage and Family* 62(4):876–95.
- DeNavas-Walt, Carmen, Bernadette D. Proctor, and Jessica C. Smith. 2011. *Income, Poverty, and Health Coverage in the United States: 2010.* Washington, D.C.: U.S. Census Bureau.
- Dew, Jeffrey. 2007. "Two Sides of the Same Coin? The Differing Roles of Assets and Consumer Debt in Marriage." *Journal of Family and Economic Issues* 28(1):89– 104.
- Dew, Jeffrey. 2008. "Debt Change and Marital Satisfaction Change in Recently Married Couples*." *Family Relations* 57(1):60–71.
- Dohrenwend, Bruce P. et al. 1992. "Socioeconomic Status and Psychiatric Disorders: The Causation-Selection Issue." *Science*.
- Don, Brian Paul, and Kristin D. Mickelson. 2012. "Paternal Postpartum Depression: The Role of Maternal Postpartum Depression, Spousal Support, and Relationship Satisfaction." *Couple and Family Psychology: Research and Practice* 1(4):323.
- Donnellan, M. B., K. J. Conger, K. K. McAdams, and T. K. Neppl. 2009. "Personal Characteristics and Resilience to Economic Hardship and Its Consequences: Conceptual Issues and Empirical Illustrations." *Journal of Personality* 77(6):1645–76.
- Dooley, David, Joann Prause, and Kathleen A. Ham-Rowbottom. 2000.
 "Underemployment and Depression: Longitudinal Relationships." *Journal of Health and Social Behavior* 41(4):421–36.
- Eaton, William W. 2001. The Sociology of Mental Disorders. 3rd ed. New York: Praeger.
- Edin, Kathryn, and Rebecca Joyce Kissane. 2010. "Poverty and the American Family: A Decade in Review." *Journal of Marriage and Family* 72(3):460–79.
- Elder, Glen H. Jr. 1998. "The Life Course as Developmental Theory." *Child Development* 69(1):1–12.
- Elder, Glen H. Jr., Monica Kirkpatrick Johnson, and Robert Crosnoe. 2003. "The Emergence and Development of Life Course Theory." in *Handbook of the Life Course*. New York: Kluwer Academic.

- Elliott, Marta. 2001. "Gender Differences in Causes of Depression." *Women & Health* 33(3-4):183–98.
- Enders, C. K., and D. L. Bandalos. 2001. "The Relative Performance of Full Information Maximum Likelihood Estimation for Missing Data in Structural Equation Models." *Structural Equation Modeling* 8(3):430–57.
- Ennis, Nicole E., Stevan E. Hobfoll, and Kerstin E. E. Schroder. 2000. "Money Doesn't Talk, It Swears: How Economic Stress and Resistance Resources Impact Inner-City Women's Depressive Mood." American Journal of Community Psychology 28(2):149–73.
- Erickson, Rebecca J. 2005. "Why Emotion Work Matters: Sex, Gender, and the Division of Household Labor." *Journal of Marriage and Family* 67(2):337–51.
- Evenson, Ranae J., and Robin W. Simon. 2005. "Clarifying the Relationship between Parenthood and Depression." *Journal of Health and Social Behavior* 46(4):341– 58.
- Fiske, Veronica, and Christopher Peterson. 1991. "Love and Depression: The Nature of Depressive Romantic Relationships." *Journal of Social and Clinical Psychology* 10(1):75–90.
- Fox, Greer Litton, Michael L. Benson, Alfred A. DeMaris, and Judy Wyk. 2002. "Economic Distress and Intimate Violence: Testing Family Stress and Resources Theories." *Journal of Marriage and Family* 64(3):793–807.
- Fox, Greer Litton, and Dudley Chancey. 1998. "Sources of Economic Distress Individual and Family Outcomes." *Journal of Family Issues* 19(6):725–49.
- Gallo, Linda C., and Karen A. Matthews. 2003. "Understanding the Association between Socioeconomic Status and Physical Health: Do Negative Emotions Play a Role?" *Psychological Bulletin* 129(1):10.
- Gibson-Davis, Christina M., Kathryn Edin, and Sara McLanahan. 2005. "High Hopes but Even Higher Expectations: The Retreat from Marriage among Low-Income Couples." *Journal of Marriage and Family* 67(5):1301–12.
- Goosby, Bridget J. 2007. "Poverty Duration, Maternal Psychological Resources, and Adolescent Socioemotional Outcomes." *Journal of Family Issues* 28(8):1113–34.
- Gudmunson, Clinton, Ivan Beutler, Craig Israelsen, J. McCoy, and E. Hill. 2007. "Linking Financial Strain to Marital Instability: Examining the Roles of Emotional Distress and Marital Interaction." *Journal of Family and Economic Issues* 28(3):357–76.

- Gupta, Anjali E., and Aletha C. Huston. 2009. "Depressive Symptoms and Economic Outcomes of Low-Income Women: A Review of the Social Causation, Social Selection, and Interactionist Hypotheses." Social Issues and Policy Review 3(1):103–40.
- Halliday Hardie, Jessica, and Amy Lucas. 2010. "Economic Factors and Relationship Quality Among Young Couples: Comparing Cohabitation and Marriage." *Journal* of Marriage and Family 72(5):1141–54.
- Hardie, Jessica Halliday, and Amy Lucas. 2010. "Economic Factors and Relationship Quality among Young Couples: Comparing Cohabitation and Marriage." *Journal* of Marriage and Family 72(5):1141–54.
- Hatfield, Elaine, John T. Cacioppo, and Richard L. Rapson. 1994. *Emotional Contagion: Studies in Emotion and Social Interaction*.
- Heavey, Christopher L., Christopher Layne, and Andrew Christensen. 1993. "Gender and Conflict Structure in Marital Interaction: A Replication and Extension." *Journal of Consulting and Clinical P* 61(1):16.
- Heflin, Colleen M., and John Iceland. 2009. "Poverty, Material Hardship, and Depression." *Social Science Quarterly* 90(5):1051–71.
- Hochschild, Arlie Russell. 1979. "Emotion Work, Feeling Rules, and Social Structure." *American Journal of Sociology* 551–75.
- Huston, Ted L. 2000. "The Social Ecology of Marriage and Other Intimate Unions." *Journal of Marriage and Family* 62(2):298–320.
- Iceland, John, and Kurt J. Bauman. 2007. "Income Poverty and Material Hardship: How Strong Is the Association?" *The Journal of Socio-Economics* 36(3):376–96.
- Johnson, Matthew D. 2012. "Healthy Marriage Initiatives: On the Need for Empiricism in Policy Implementation." *American Psychologist* 67(4):296–308.
- Johnson, Michael P., and Kathleen J. Ferraro. 2000. "Research on Domestic Violence in the 1990s: Making Distinctions." *Journal of Marriage and Family* 62(4):948–63.
- Kahn, Joan R., and Leonard I. Pearlin. 2006. "Financial Strain over the Life Course and Health among Older Adults." *Journal of Health and Social Behavior* 47(1):17–31.
- Kalmijn, Matthijs, Anneke Loeve, and Dorien Manting. 2007. "Income Dynamics in Couples and the Dissolution of Marriage and Cohabitation." *Demography* 44(1):159–79.

- Karney, Benjamin R., and Thomas N. Bradbury. 1995. "The Longitudinal Course of Marital Quality and Stability: A Review of Theory, Method, and Research." *Psychological Bulletin* 118(1):3–34.
- Karney, Benjamin R., and Thomas N. Bradbury. 2005. "Contextual Influences on Marriage Implications for Policy and Intervention." *Current Directions in Psychological Science* 14(4):171–74.
- Kennedy, Sheela, and Larry Bumpass. 2008. "Cohabitation and Children's Living Arrangements: New Estimates from the United States." *Demographic Research* 19:1663.
- Kenny, David A., Deborah A. Kashy, and William L. Cook. 2006. *Dyadic Data Analysis*. New York: The Guilford Press.
- Kessler R.C, Berglund P. Demler O. Jin R. Merikangas K. R. Walters E. E. 2005. "LIfetime Prevalence and Age-of-Onset Distributions of Dsm-Iv Disorders in the National Comorbidity Survey Replication." Archives of General Psychiatry 62(6):593–602.
- Kessler, Ronald C. 2003. "Epidemiology of Women and Depression." *Journal of Affective Disorders* 74(1):5–13.
- Kinnunen, Ulla, and Taru Feldt. 2004. "Economic Stress and Marital Adjustment among Couples: Analyses at the Dyadic Level." *European Journal of Social Psychology* 34(5):519–32.
- Kouros, C. D., and E. Mark Cummings. 2011. "Transactional Relations between Marital Functioning and Depressive Symptoms." *American Journal of Orthopsychiatry* 81(1):128–38.
- Larson, Reed W., and David M. Almeida. 1999a. "Emotional Transmission in the Daily Lives of Families: A New Paradigm for Studying Family Process." *Journal of Marriage and the Family*.
- Larson, Reed W., and David M. Almeida. 1999b. "Emotional Transmission in the Daily Lives of Families: A New Paradigm for Studying Family Process." *Journal of Marriage and the Family* 61(1).
- Lavee, Yoav. 2013. "Stress Processes in Families and Couples." Pp. 159–76 in Handbook of Marriage and the Family. Springer.
- Lin, A. C., and D. R. Harris, eds. 2008. *The Colors of Poverty: Why Racial and Ethnic Disparities Exist*. Russell Sage Foundation Publications.
- Lorant, Vincent et al. 2003. "Socioeconomic Inequalities in Depression: A Meta-Analysis." *American Journal of Epidemiology* 157(2):98–112.

- Lynch, John W., George A. Kaplan, and Sarah J. Shema. 1997. "Cumulative Impact of Sustained Economic Hardship on Physical, Cognitive, Psychological, and Social Functioning." *New England Journal of Medicine* 337(26):1889–95.
- Manning, Wendy D., and Jessica A. Cohen. 2012. "Premarital Cohabitation and Marital Dissolution: An Examination of Recent Marriages." *Journal of Marriage and Family* 74(2):377–87.
- Manuel, Jennifer I., Melissa L. Martinson, Sarah E. Bledsoe-Mansori, and Jennifer L. Bellamy. 2012. "The Influence of Stress and Social Support on Depressive Symptoms in Mothers with Young Children." *Social Science and Medicine* 75(11):2013–20.
- McLanahan, Sara. 2004. "Diverging Destinies: How Children Are Faring under the Second Demographic Transition." *Demography* 41(4):607–27.
- McLanahan, Sara. 2009. "Fragile Families and the Reproduction of Poverty." *The Annals* of the American Academy of Political and Social Science 621(1):111–31.
- McLanahan, Sara, and Julia Adams. 1987. "Parenthood and Psychological Well-Being." Annual Review of Sociology 13(5):237–57.
- McLanahan, Sara, and Audrey N. Beck. 2010. "Parental Relationships in Fragile Families." *The Future of Children* 20(2):17.
- McLanahan, Sara, and Christine Percheski. 2008. "Family Structure and the Reproduction of Inequalities." *Annual Review of Sociology* 34(1):257–76.
- McLanahan, Sara S. 2006. Fragile Families and the Marriage Agenda. NY: Springer.
- McLoyd, Vonnie C. 1990. "The Impact of Economic Hardship on Black Families and Children: Psychological Distress, Parenting, and Socioemotional Development." *Child Development* 61(2):311–46.
- McLoyd, Vonnie C. 1998. "Socioeconomic Disadvantage and Child Development." *American Psychologist* 53(2):185–204.
- McLoyd, Vonnie C., Ana Mari Cauce, David Takeuchi, and Leon Wilson. 2000. "Marital Processes and Parental Socialization in Families of Color: A Decade Review of Research." *Journal of Marriage and Family* 62(4):1070–93.
- Miech, Richard Allen, and Michael J. Shanahan. 2000. "Socioeconomic Status and Depression over the Life Course." *Journal of Health and Social Behavior* 41(2):162–76.

- Milkie, Melissa A. 2010. "The Stress Process Model: Some Family-Level Considerations." Pp. 93–108 in *Advances in the Conceptualization of the Stress Process*. Springer.
- Milkie, Melissa A., Alex Bierman, and Scott Schieman. 2008. "How Adult Children Influence Older Parents' Mental Health: Integrating Stress-Process and Life-Course Perspectives." *Social Psychology Quarterly* 71(1):86–105.
- Milkie, Melissa A., Marybeth J. Mattingly, Kei M. Nomaguchi, Suzanne M. Bianchi, and John P. Robinson. 2004. "The Time Squeeze: Parental Statuses and Feelings about Time with Children." *Journal of Marriage and Family* 66(3):739–61.
- Mirowsky, John, and Catherine Ross. 2002. "Depression, Parenthood, and Age at First Birth." *Social Science and Medicine* 54(8):1281–98.
- Mirowsky, John, and Catherine E. Ross. 1999. "Economic Hardship across the Life Course." *American Sociological Review* 548–69.
- Mirowsky, John, and Catherine E. Ross. 2001. "Age and the Effect of Economic Hardship on Depression." *Journal of Health and Social Behavior* 42(2):132–50.
- Muntaner, Carles, William W. Eaton, Richard Miech, and Patricia O'Campo. 2004. "Socioeconomic Position and Major Mental Disorders." *Epidemiologic Reviews* 26(1):53–62.
- Muntaner, Carles, Edwin Ng, Christophe Vanroelen, Sharon Christ, and William W. Eaton. 2013. "Social Stratification, Social Closure, and Social Class as Determinants of Mental Health Disparities." Pp. 205–27 in *Handbook of the Sociology of Mental Health*. Springer.
- Muthen, L., and B. O. Muthen. 2010. *Mplus User's Guide. Sixth Edition*. Sixth. Los Angeles, CA: Muthen & Muthen.
- National Institute of Child Health, and Human Development Early Child Care Research Network. 2005. "Duration and Developmental Timing of Poverty and Children's Cognitive and Social Development from Birth through Third Grade." *Child Development* 76(4):795–810.
- Neff, Lisa A., and Benjamin R. Karney. 2007. "Stress Crossover in Newlywed Marriage: A Longitudinal and Dyadic Perspective." *Journal of Marriage and Family* 69(3):594–607.
- Nelson, S. Katherine, Kostadin Kushlev, Tammy English, Elizabeth W. Dunn, and Sonja Lyubomirsky. 2013. "In Defense of Parenthood Children Are Associated With More Joy Than Misery." *Psychological Science* 24(1):3–10.

- Nolen-Hoeksema, Susan, Judith Larson, and Carla Grayson. 1999. "Explaining the Gender Difference in Depressive Symptoms." *Journal of Personality and Social Psychology* 77(5):1061–72.
- Nomaguchi, Kei, and Amanda N. House. 2013. "Racial-Ethnic Disparities in Maternal Parenting Stress The Role of Structural Disadvantages and Parenting Values." *Journal of Health and Social Behavior* 54(3):386–404.
- Nomaguchi, Kei M., and Melissa A. Milkie. 2003. "Costs and Rewards of Children: The Effects of Becoming a Parent on Adults' Lives." *Journal of Marriage and Family* 65(2):356–74.
- O'Brien, Marion. 2005. "Studying Individual and Family Development: Linking Theory and Research." *Journal of Marriage and Family* 67(4):880–90.
- Olesen, Sarah C., Peter Butterworth, Liana S. Leach, Margaret Kelaher, and Jane Pirkis. 2013. "Mental Health Affects Future Employment as Job Loss Affects Mental Health: Findings from a Longitudinal Population Study." *BMC Psychiatry* 13(1):144.
- Paat, Yok-Fong. 2011. "The Link Between Financial Strain, Interpersonal Discord and Children's Antisocial Behaviors." *Journal of Family Violence* 26(3):195–210.
- Papp, Lauren M., E. Mark Cummings, and Marcie C. Goeke-Morey. 2009. "For Richer, for Poorer: Money as a Topic of Marital Conflict in the Home." *Family Relations* 58(1):91–103.
- Parke, R. D. et al. 2004. "Economic Stress, Parenting, and Child Adjustment in Mexican American and European American Families." *Child Development* 75(6):1632–56.
- Parke, Ross D. et al. 2004. "Economic Stress, Parenting, and Child Adjustment in Mexican American and European American Families." *Child Development* 75(6):1632–56.
- Paulson, J. F., S. Dauber, and J. A. Leiferman. 2006. "Individual and Combined Effects of Postpartum Depression in Mothers and Fathers on Parenting Behavior." *Pediatrics* 118(2):659–68.
- Paulson, James F., and Sharnail Bazemore. 2010. "Prenatal and Postpartum Depression in Fathers and Its Association with Maternal Depression: A Meta-Analysis." JAMA: The Journal of the American Medical Association 303(19):1961–69.
- Pearlin, Leonard I., Elizabeth G. Menaghan, A. Lieberman Morton, and Joseph T. Mullan. 1981. "The Stress Process." *Journal of Health and Social Behavior* 22(4):337–56.

- Pearlin, Leonard I., Scott Schieman, Elena M. Fazio, and Stephen C. Meersman. 2005. "Stress, Health, and the Life Course: Some Conceptual Perspectives." *Journal of Health and Social Behavior* 46(2):205–19.
- Petterson, Stephen M., and Alison Burke Albers. 2001. "Effects of Poverty and Maternal Depression on Early Child Development." *Child Development* 72(6):1794–1813.
- Proulx, C. M., H. M. Helms, and C. Buehler. 2007. "Marital Quality and Personal Well-Being: A Meta-Analysis." *Journal of Marriage and Family* 69(3):576–93.
- Pruchno, R., M. Wilson-Genderson, and F. P. Cartwright. 2009. "Depressive Symptoms and Marital Satisfaction in the Context of Chronic Disease: A Longitudinal Dyadic Analysis." *Journal of Family Psychology; Journal of Family Psychology* 23(4):573.
- Putnick, Diane L. et al. 2010. "Stability, Continuity, and Similarity of Parenting Stress in European American Mothers and Fathers across Their Child's Transition to Adolescence." *Parenting: Science and Practice* 10(1):60–77.
- Raikes, H. Abigail, and Ross A. Thompson. 2005. "Efficacy and Social Support as Predictors of Parenting Stress among Families in Poverty." *Infant Mental Health Journal* 26(3):177–90.
- Ramchandani, Paul G. et al. 2008. "Depression in Men in the Postnatal Period and Later Child Psychopathology: A Population Cohort Study." *Journal of the American Academy of Child & Adolescent Psychiatry* 47(4):390–98.
- Randall, Ashley K., and Guy Bodenmann. 2009. "The Role of Stress on Close Relationships and Marital Satisfaction." *Clinical Psychology Review* 29(2):105– 15.
- Reichman, Nancy E., Hope Corman, and Kelly Noonan. 2004. "Effects of Child Health on Parents' Relationship Status." *Demography* 41(3):569–84.
- Reichman, Nancy E., Julien O. Teitler, Irwin Garfinkel, and Sara S. McLanahan. 2001. "Fragile Families: Sample and Design." *Children and Youth Services Review* 23(4/5):303–26.
- Ridgeway, Cecilia L. 2011. Framed by Gender: How Gender Inequality Persists in the Modern World. Oxford University Press.
- Du Rocher Schudlich, T. D., L. M. Papp, and E. M. Cummings. 2011. "Relations between Spouses' Depressive Symptoms and Marital Conflict: A Longitudinal Investigation of the Role of Conflict Resolution Styles." *Journal of Family Psychology* 25(4):531.

- Rosenfield, Sarah, and Dawne Mouzon. 2013. "Gender and Mental Health." in *Handbook* of the Sociology of Mental Health, vol. 2nd, edited by Carol S. Aneshensel, Jo C. Phelan, and Alex Bierman. New York: Springer.
- Ross, Catherine E., and Joan Huber. 1985. "Hardship and Depression." *Journal of Health and Social Behavior* 312–27.
- Rubin, D. B., and R. J. A. Little. 2002. "Statistical Analysis with Missing Data." *Hoboken, NJ: J Wiley & Sons.*
- Sassler, Sharon. 2010. "Partnering across the Life Course: Sex, Relationships, and Mate Selection." *Journal of Marriage and Family* 72(3):557–75.
- Schoppe-Sullivan, Sarah J., Sarah C. Mangelsdorf, Cynthia A. Frosch, and Jean L. McHale. 2004. "Associations between Coparenting and Marital Behavior from Infancy to the Preschool Years." *Journal of Family Psychology* 18(1):194.
- Smock, Pamela J. 2000. "Cohabitation in the United States: An Appraisal of Research Themes, Findings, and Implications." *Annual Review of Sociology* 26(1):1–20.
- Sobolewski, Juliana M., and Paul R. Amato. 2005. "Economic Hardship in the Family of Origin and Children's Psychological Well-Being in Adulthood." *Journal of Marriage and Family* 67(1):141–56.
- Sohr-Preston, Sara L., and Laura V. Scaramella. 2006. "Implications of Timing of Maternal Depressive Symptoms for Early Cognitive and Language Development." *Clinical Child and Family Psychology Review* 9(1):65–83.
- Solantaus, Tytti, Jenni Leinonen, and Raija-Leena Punamaki. 2004. "Children's Mental Health in Times of Economic Recession: Replication and Extension of the Family Economic Stress Model in Finland." *Developmental Psychology* 40(3):412–29.
- Song, Zhaoli, Maw-Der Foo, and Marilyn A. Uy. 2008. "Mood Spillover and Crossover among Dual-Earner Couples: A Cell Phone Event Sampling Study." *Journal of Applied Psychology* 93(2):443–52.
- Thoits, Peggy A. 2010. "Stress and Health: Major Findings and Policy Implications." *Journal of Health and Social Behavior* 51(1 suppl):S41–S53.
- Thoits, Peggy A. 2011. "Mechanisms Linking Social Ties and Support to Physical and Mental Health." *Journal of Health and Social Behavior* 52(2):145–61.
- Thompson, A., and N. Bolger. 1999. "Emotional Transmission in Couples under Stress." *Journal of Marriage and the Family* 61(1):38–48.

- Thomson, Elizabeth, Thomas L. Hanson, and Sara S. McLanahan. 1994. "Family Structure and Child Well-Being: Economic Resources vs. Parental Behaviors." *Social Forces* 73(1):221–42.
- Tucker, Jenna N., Joseph G. Grzywacz, Iris Leng, C. Randall Clinch, and Thomas A. Arcury. 2010. "Return to Work, Economic Hardship, and Women's Postpartum Health." Women & Health 50(7):618–38.
- Turney, K., and M. J. Carlson. 2011. "Multipartnered Fertility and Depression among Fragile Families." *Journal of Marriage and Family* 73(3):570–87.
- Umberson, Debra, Meichu D. Chen, James S. House, Kristine Hopkins, and Ellen Slaten. 1996. "The Effect of Social Relationships on Psychological Well-Being: Are Men and Women Really so Different?" *American Sociological Review* 837–57.
- Umberson, Debra, and Jennifer Karas Montez. 2010. "Social Relationships and Health: A Flashpoint for Health Policy." *Journal of Health and Social Behavior* 51(1 suppl):S54–S66.
- Umberson, Debra, Tetyana Pudrovska, and Corinne Reczek. 2010. "Parenthood, Childlessness, and Well-Being: A Life Course Perspective." *Journal of Marriage and Family* 72(3):612–29.
- Umberson, Debra, and Corinne Reczek. 2007. "Interactive Stress and Coping Around Parenting: Explaining Trajectories of Change in Intimate Relationships Over the Life Course." Advances in Life Course Research 12:87–121.
- Umberson, Debra, Kristi Williams, Daniel A. Powers, Meichu D. Chen, and Anna M. Campbell. 2005. "As Good as It Gets? A Life Course Perspective on Marital Quality." Social Forces 84(1):493–511.
- Vogel, David L., and Benjamin R. Karney. 2002. "Demands and Withdrawal in Newlyweds: Elaborating on the Social Structure Hypothesis." *Journal of Social* and Personal Relationships 19(5):685–701.
- Wadsworth, Martha E., and Thomas M. Achenbach. 2005. "Explaining the Link between Low Socioeconomic Status and Psychopathology: Testing Two Mechanisms of the Social Causation Hypothesis." *Journal of Consulting and Clinical Psychology* 73(6):1146.
- Waller, Maureen R., and Sara S. McLanahan. 2005. "'His' and 'her' Marriage Expectations: Determinants and Consequences." *Journal of Marriage and Family* 67(1):53–67.
- Webb, David A. et al. 2008. "Postpartum Physical Symptoms in New Mothers: Their Relationship to Functional Limitations and Emotional Well-Being." *Birth* 35(3):179–87.

- Wee, Kim Yiong, Helen Skouteris, Ciaran Pier, Ben Richardson, and Jeannette Milgrom. 2011. "Correlates of Ante- and Postnatal Depression in Fathers: A Systematic Review." *Journal of Affective Disorders* 130(3):358–77.
- Western, Bruce, Leonard M. Lopoo, and Sara McLanahan. 2004. "Incarceration and the Bonds between Parents in Fragile Families." Pp. 21–45 in *Imprisoning America: The Social Effects of Mass Incarceration*.
- Westman, Mina. 2001. "Stress and Strain Crossover." Human Relations 54(6):717-51.
- Westman, Mina, and Dalia Etzion. 1995. "Crossover of Stress, Strain and Resources from One Spouse to Another." *Journal of Organizational Behavior* 16(2):169–81.
- Westman, Mina, and Amiram D. Vinokur. 1998. "Unraveling the Relationship of Distress Levels within Couples: Common Stressors, Empathic Reactions, or Crossover via Social Interactions?" *Human Relations* 51(2):137–56.
- Whisman, Mark A., and Lisa A. Uebelacker. 2009. "Prospective Associations between Marital Discord and Depressive Symptoms in Middle-Aged and Older Adults." *Psychology and Aging* 24(1):184.
- Whooley, Mary A. et al. 2002. "Depressive Symptoms, Unemployment, and Loss of Income: The CARDIA Study." *Archives of Internal Medicine* 162(22):2614–20.
- Wickrama, K. A. S., Kyung Hwa Kwag, Federick O. Lorenz, Rand D. Conger, and Florensia F. Surjadi. 2010. "Dynamics of Family Economic Hardship and the Progression of Health Problems of Husbands and Wives During the Middle Years: A Perspective From Rural Mid-West." *Journal of Aging and Health* 22(8):1132–57.
- Wickrama, K. A. S., Florensia F. Surjadi, Frederick O. Lorenz, Rand D. Conger, and Catherine Walker O'Neal. 2012. "Family Economic Hardship and Progression of Poor Mental Health in Middle-Aged Husbands and Wives." *Family Relations* 61(2):297–312.
- Williams, Deadric T., Jacob E. Cheadle, and Bridget J. Goosby. 2013. "Hard Times and Heart Break Linking Economic Hardship and Relationship Distress." *Journal of Family Issues*.
- Winship, Christopher, and Larry Radbill. 1994. "Sampling Weights and Regression Analysis." *Sociological Methods & Research* 23(2):230–57.
- Wu, Zheng, and Michael S. Pollard. 2000. "Economic Circumstances and the Stability of Nonmarital Cohabitation." *Journal of Family Issues* 21(3):303–28.

- Yorgason, Jeremy B., David Almeida, and Shevaun D. Neupert. 2006. "A Dyadic Examination of Daily Health Symptoms and Emotional Well-Being in Late-Life Couples." *Family Relations* 55(5):613–24.
- Zilanawala, A., and N. V. Pilkauskas. 2012. "Material Hardship and Child Socioemotional Behaviors: Differences by Types of Hardship, Timing, and Duration." *Children and Youth Services Review*.
- Zimmerman, Frederick J., and Wayne Katon. 2005. "Socioeconomic Status, Depression Disparities, and Financial Strain: What Lies behind the Income-Depression Relationship?" *Health Economics* 14(12):1197–1215.



Figure 2.1: Conceptual Model for the Longitudinal and Reciprocal Effects of Economic Hardship and Depressive Symptoms among Couples



Figure 2.2: Conceptual Model Linking Economic Hardship, Parents' Depressive Symptoms, and Parents' Relationship Distress

	All Coup	les	Married Co	uples	Cohabiting Co	ouples
	Mean or %	SD	Mean or %	SD	Mean or %	SD
Family Structure						
Married	72.61					
Cohabiting	27.39					
Age at child's birth						
Mother	28.66	5.66	29.89	5.11	25.38	5.75
Father	31.01	6.38	32.02	5.91	28.33	6.80
Mother's race/ethnicity						
White non-Hispanic	42.70		53.30		14.50	
Black non-Hispanic	19.16		11.30		40.00	
Hispanic	30.12		25.70		41.90	
Other non-Hispanic	8.03		9.68		3.62	
Parents are a different race/ethnicity	11.71		12.30		10.20	
Mother's Education						
Less than High School	19.10		14.40		31.60	
High school diploma or equivalent	27.30		20.20		46.20	
Some college	19.55		19.10		20.70	
Bachelor's degree or higher	34.04		46.30		1.43	
Father's Education						
Less than High School	20.24		14.30		36.00	
High school diploma or equivalent	20.22		15.30		33.10	
Some college	26.10		26.60		24.70	
Bachelor's degree or higher	33.44		43.70		6.21	
Health Status (1-Year)						
Mother's Health	4.04	0.88	4.16	0.82	3.70	0.95
Father's Health	4.05	0.92	4.10	0.89	3.91	0.98
Child Health	4.59	0.68	4.64	0.62	4.47	0.82
Employment Status (Year 1)						
Mother has $job = 1$	55.95		57.40		52.00	
Father has $job = 1$	90.28		95.10		77.50	
Social Support (Year 1)						
Mother received support	23.81		21.10		31.10	
Father received support	29.10		27.10		34.50	
Incarceration History (1-Year)						
Father	9.33		4.66		21.90	
Domestic Violent						
Father	3.23		3.06		3.69	
Number of children in home < 18 years old	1.01	1.26	1.01	1.25	1.01	1.28

Table 2.1: Sample Statistics (Means, Percentages, and Standard Deviations) among Couples, by Marital Status

Table 2.1 continues on next page

	All Couples	Married Couples	Cohabiting Couples
	Mean or % SD	Mean or % SD	Mean or % SD
Fertility History			
Couple Higher Order Birth (Yes=1)	50.98	54.50	41.70
Multipartnered Fertility			
Neither parent has a child by another partner	73.15	83.2	46.20
Father has a child by another partner	9.73	7.21	16.50
Mother has a child by another partner	11.08	6.77	22.60
Both parents have a child by another partner	6.04	2.77	14.80
Below poverty line = 1 (1-Year)	26.01	18.50	46.00
N (Unweighted)	1218	631	587

Table 2.1: Sample Statistics (Means, Percentages, and Standard Deviations) among Couples, by Marital Status

Note: Variables are from the baseline and 1-Year survey. All means are weighted using city sampling weights. Numbers of cases are unweighted.

Table 2.2:	Weighted Means for Economic	Hardship, D	epressive Symptoms,	and Relationship	Distress, by	/ Marital
Status						

	All Co	uples	Married 0	Couples	Cohabiting	Couples	Mean
	Mean	SD	Mean	SD	Mean	SD	Difference
Economic Hardship (Mother Reports)							
Year 1	0.60	1.10	0.51	1.08	0.86	1.10	-0.35
Year 3	0.52	0.96	0.39	0.80	0.85	1.24	-0.46 **
Year 5	0.54	1.08	0.41	0.94	0.90	1.33	-0.49 *
Depressive Symptoms (Mothers)							
Year 1	0.38	1.23	0.26	1.00	0.72	1.65	-0.46 *
Year 3	0.53	1.38	0.44	1.26	0.79 ^a	1.64	-0.35 *
Year 5	0.42	1.27	0.41	1.23	0.46	1.35	-0.05
Depressive Symptoms (Fathers)							
Year 1	0.41	1.21	0.38	1.14	0.49	1.37	-0.11
Year 3	0.53	1.43	0.58	1.49	0.38 ^a	1.23	0.20
Year 5	0.34	1.15	0.34	1.09	0.36	1.28	-0.03
Relationship Distress							
Year 5 (Mothers)	1.31	0.50	1.24	0.45	1.48	0.59	-0.24 **
Year 5 (Fathers)	1.21	0.39	1.16	0.36	1.32	0.44	-0.16

Note: All means are weighted using city sampling weights. Mean differences by marital status are tested using two-tailed *t* tests. Two-sample *t* tests were used to test differences within families, and means with identical superscripts denote differences between mothers and fathers. ***p < .001; **p < .01; *p < .05; †p < .10

	All Couples					Married Couples					Cohabiting Couples					
	Moth	ner	Father	Father		ther	Fath	er	Mo	Mother		her				
	exp(b)	t	exp(b)	t	exp(b)	t	exp(b)	t	exp(b)	se	exp(b)	t				
Direct Effect																
Econ Hardship 1 -> Dep. Symptoms 3	1.27**	3.42	1.22**	2.60	1.34**	2.61	1.11	0.76	1.30**	2.91	1.22*	2.19				
Econ Hardship 3 -> Dep. Symptoms 5	1.21**	2.60	1.04	0.52	1.60** ^a	3.39	1.39*	1.99	1.15†ª	1.90	1.00	0.05				
Reciprocal Effect																
Dep Symptoms 1 -> Econ Hardship 3	1.09**	3.15	1.00	0.03	1.16**	2.69	1.05	0.79	1.07*	1.10	0.98	-0.43				
Dep Symptoms 3 -> Econ Hardship 5	0.99	-0.10	1.02	0.67	1.02	0.36	1.06	1.13	0.97	-0.88	1.00	0.04				

Table 2.3: Exponentiated Beta Coefficients for the Longitudinal and Reciprocal Effects on Economic Hardship and Depressive Symptoms, by Martial Status

Note: Parameter estimates with identical superscripts denote statistical differences. Dep = Depressive Symptoms ***p < .001; **p < .01; *p < .05; †p < .10

		All Couples						Married	Couples				Cohabiting Couples			
	Moth	ner		Fatl	her		Mothe	er	Fath	er		Mot	Mother		Father	
	В	SE		В	SE		В	SE	В	SE		В	SE		В	SE
Economic Hardship (5-Yr)	0.20***	0.03		0.06*	0.03		0.25*** ^a	0.04	0.11*** ^a	0.04		0.18*** ^b	0.04		0.03 ^b	0.04
Actor Effects Depressive Symptoms (5- Yr)	0.16***	0.03		0.26***	0.03		0.09*	0.04	0.18***c	0.04		0.19***	0.04	0.	.31*** ^d	0.04
Partner Effects Depressive Symptoms (5- Yr)	0.11*	0.03		0.05†	0.03		0.06	0.04	0.07†°	0.041		0.14***	0.04		0.03 ^d	0.04
Model Fit:	x ²	df	CFI	RMSEA	SRMR				x ²	df	CFI	RMSEA	SRMR			
	7.78	22.00	1.00	0.000	0.005				5.61	44.00	1.00	0.000	0.007			
							x ² diffe	erence test:	7.82	6.00	р	= 0.33				

Table 2.4: Standardized Regression Coefficients for the Effects of Economic Hardship and Depressive Symptoms on Relationship Distress

Note: Parameter estimates with identical superscripts denote statistical differences. ***p < .001; **p < .01; *p < .05; †p < .10

	Moth	ers	Fathe	rs
	exp(b)	t	exp(b)	t
Variables				
Depressive Symptoms	1.06	0.99	1.01	0.17
Economic Hardship	0.89	-1.42	0.97	-0.44
Marital Status				
Married (reference)				
Cohabiting	1.21	0.88	1.94**	3.19
Parent's Age	1.01	0.88	1.00	-0.42
Race/Ethnicity				
Non-Hispanic White (reference)				
Non-Hispanic Black	1.22	0.79	0.82	-0.84
Hispanic	1.83*	2.39	1.45	1.62
Other	1.68	1.19	0.72	-0.66
Parent's Education				
Less than H.S. (reference)				
High School Diploma	0.90	-0.48	0.80	-1.16
Some College	0.85	-0.67	0.66+	-1.77
Bachelor's Degree or higher	0.57	-1.56	0.64	-1.30

Table 2.5: Attrition for Mothers and Fathers Betwen Years 1 - 3

***p < .001; **p < .01; *p < .05; †p < .10

Table 2.6: Attrition for Mothers and Fathers Between Years 3 - 5

	Moth	ers	Fathe	ers
	exp(b)	t	exp(b)	t
Variables	_			
Depressive Symptoms	1.00	0.07	0.89	-1.62
Economic Hardship	0.88	-1.34	1.01	0.14
Marital Status				
Married (reference)				
Cohabiting	1.20	0.83	1.11	0.50
Parent's Age	1.02	0.90	0.98	-1.19
Race/Ethnicity				
Non-Hispanic White (reference)				
Non-Hispanic Black	0.88	-0.51	1.12	0.46
Hispanic	1.38	1.29	1.01	0.04
Other	1.89†	1.68	0.87	-0.26
Parent's Education				
Less than H.S. (reference)				
High School Diploma	0.67	-1.62	1.40	1.59
Some College	1.00	-0.01	0.56*	-2.06
Bachelor's Degree or higher	0.68	-1.18	0.58	-1.49

Table 2.7: Relationship Dissolution Between Years 1 - 3

	Mothers	3	Fathers	8
	exp(b)	t	exp(b)	t
Variables				
Depressive Symptoms	1.04	1.01	1.09+	1.78
Economic Hardship	1.207***	3.52	1.19***	3.39
Marital Status				
Married (reference)				
Cohabiting	2.693***	5.17	2.69***	5.26
Parent's Age	0.940***	-4.44	0.97**	-2.96
Race/Ethnicity				
Non-Hispanic White (reference)				
Non-Hispanic Black	1.779***	3.37	1.86**	3.28
Hispanic	0.524**	-3.05	0.67+	-1.78
Other	1.01	0.03	1.87+	1.81
Parent's Education				
Less than H.S. (reference)				
High School Diploma	1.14	0.80	1.27	1.27
Some College	1.08	0.43	0.77	0,77
Bachelor's Degree or higher	0.503+	-1.93	0.33**	-2.85

Table 2.8: Relationship Dissolution Between Years 3 - 5

	Mother	S	Fathers	8
	exp(b)	t	exp(b)	t
Variables				
Depressive Symptoms	1.02	0.49	1.12*	2.45
Economic Hardship	1.14*	2.25	1.11+	1.75
Marital Status				
Married (reference)				
Cohabiting	3.27***	6.00	3.10***	5.80
Parent's Age	0.98+	-1.70	0.98*	-2.04
Race/Ethnicity				
Non-Hispanic White (reference)				
Non-Hispanic Black	1.60*	2.57	2.03***	3.44
Hispanic	0.77	-1.25	1.16	0.66
Other	1.13	0.31	0.66	-0.76
Parent's Education				
Less than H.S. (reference)				
High School Diploma	0.90	-0.65	1.10	0.54
Some College	0.73+	-1.65	0.89	-0.57
Bachelor's Degree or higher	0.58+	-1.76	0.80	-0.71

	Table 2.9: Control Variables on the Longitudinal and Reciprocal Effects
--	---

	Econ. Hardshi	o (3-Year)	Econ. Hard	ship (5-Year)		Dep. Syn	n. (3-Year)		Dep. Sym. (5-Year)				
	Mother R	eport	Mothe	r Report	Moth	iers	Father	rs	Moth	ers	Father	s	
Control Variables	b	se	b	se	b	se	b	se	b	se	b	se	
Dep. Sym (Lagged)					0.37***	-0.97	0.428***	0.05	0.32***	0.04	0.497***	0.05	
Econ. Hardship (Lagged)	0.416***	0.04	0.48	0.037***									
Family Structure													
Married (reference)													
Cohabiting	0.24*	0.12	0.251*	0.12	0.12	0.20			0.34	0.25	0.33	0.27	
Age at child's birth													
Mothers	-0.02†	0.01	-0.01	0.01	0.00	0.02			0.00	0.02			
Fathers	-0.01	0.01	0.01	0.01			-0.04*	0.02			0.03	0.02	
Mother's race/ethnicity													
White non-Hispanic (reference)													
Black non-Hispanic	-0.11	0.12	0.09	0.12	-0.02	0.20	-0.38†	0.22	-0.44†	0.24	0.06	0.27	
Hispanic	-0.16	0.13	-0.17	0.14	-0.29	0.23	0.00	0.23	-0.31	0.26	0.19	0.31	
Other non-Hispanic	-0.01	0.24	0.23	0.25	-0.41	0.56	-0.12	0.44	-0.82†	0.48	-1.087+	0.60	
Parents are a different race/ethnicity	0.12	0.14	0.09	0.13	0.31	0.25	0.01	0.24	0.33	0.31	0.45	0.35	
Mother's Education													
Less than High School (reference)													
High school diploma or equivalent	0.20	0.13	0.246†	-0.86	0.09	0.23			0.62*	0.26			
Some college	0.50***	0.13	0.21	0.16	0.25	0.24			0.64*	0.26			
Bachelor's degree or higher	0.12	0.20	-0.10	0.25	-0.20	0.32			0.76*	0.34			
Father's Education	0.12	0.20	0.10	0.25	0.20	0.52			0.70	0.54			
Less than High School (reference)													
High school diploma or equivalent	0.23*	0.13	-0.11	0.13			0.26	0.22			-0.19	0.27	
Some college	0.10	0.15	-0.07	0.15			-0.27	0.22			-0.27	0.35	
Bachelor's degree or higher	-0.54*	0.14	-0.618*	0.27			-0.26	0.25			-0.02	0.43	
Health Status (1-Vear)	-0.54	0.25	-0.010	0.27			-0.20	0.55			-0.02	0.45	
Mother's Health	0.11*	0.05	0.10+	0.05	0.07	0.00			0.15	0.10			
Fether's Health	-0.11	0.05	-0.10	0.05	-0.07	0.09	0.10	0.00	-0.15	0.10	0 242**	0.10	
Child Health	0.00	0.00	-0.02	0.00	0.27**	0.11	-0.10	0.09	0.27*	0.12	-0.343	0.10	
Employment Status (Veer 1)	-0.07	0.00	0.04	0.07	-0.27	0.11	0.05	0.12	-0.27	0.15	-0.21	0.14	
Mother has ich = 1	0.10	0.10	0.16	0.10	0.264	0.16			0.40*	0.10			
Notice has $job = 1$	-0.10	0.10	-0.16	0.10	-0.201	0.16	0.50*	1.02	-0.40*	0.19	0.424+	0.26	
Father has $job = 1$	0.01	0.12	-0.06	0.12			-0.59**	1.23			-0.4341	0.26	
Social Support (Tear 1)	0.24**	0.10	0.244*	0.10	0.22+	0.17			0 66**	0.10			
Father received support	0.34**	0.10	0.244*	0.10	0.32	0.17	0.00	0.20	0.00**	0.19	0.20	0.21	
Father received support	0.21*	0.10	0.15	0.10			0.09	0.20			0.30	0.21	
Incarceration History (1- rear)	0.00*	0.11	0.02	0.12	0.12	0.00	0.22	0.00	0.04	0.00	0.5001	0.04	
Father	0.28*	0.11	0.02	0.12	-0.13	0.20	0.33	0.22	0.36	0.23	0.502*	0.26	
Domestic Violent	0.4444								0.48		4.05		
Father	-0.61**	0.21	-0.03	0.20	-0.02	0.42	-0.35	0.46	0.47	0.52	-1.07	0.66	
Number of children in home < 18 years old	0.15***	0.04	-0.02	0.05	-0.14†	0.07	0.17	0.09	0.12	0.08	0.04	0.09	
Fertility History	0.04					0.40	0.4444		0.00				
Couple Higher Order Birth (Yes= 1)	-0.01	0.11	0.07	0.11	0.36†	0.19	0.61**	0.21	0.00	0.22	0.03	0.25	
Multipartnered Fertility	- ·												
Neither parent has a child by another partner (reterence)										0.40	0.00	
Father has a child by another partner	0.17	0.14	0.06	0.14	0.41†	0.22	0.35	0.26	0.34	0.27	0.10	0.29	
Mother has a child by another partner	-0.08	0.14	0.00	0.15	0.17	0.24	-0.08	0.24	-0.41	0.25	-0.45	0.32	
Both parents have a child by another partner	0.17	0.17	0.10	0.16	0.44†	0.25	0.11	0.32	-0.09	0.33	0.36	0.37	
Below poverty line = $1 (1-Year)$	0.04	0.11	0.04	0.12	0.03	0.20	-0.46*	0.19	-0.15	0.23	-0.03	0.24	

	Econ. Hardship (3-Year) Mother Report		Econ. Hardship (5-Year) Mother Report			Dep. Sym	. (3-Year)		Dep. Sym. (5-Year)				
					Mothers		Fathers		Mothers		Fathers		
Control Variables	b	se	b	se	b	se	b	se	b	se	b	se	
Dep. Sym (Lagged)					0.445***	0.06	0.508***	0.09	0.329***	0.07	0.486***	0.08	
Econ. Hardship (Lagged)	0.472***	0.06	0.693***	0.08									
Age at child's birth													
Mothers	-0.01	0.02	-0.01	0.02	0.03	0.02			0.048+	0.03			
Fathers	-0.03	0.02	-0.01	0.02			-0.076*	0.03			0.082*	0.03	
Mother's race/ethnicity													
White non-Hispanic (reference)													
Black non-Hispanic	0.13	0.19	0.17	0.23	-0.04	0.30	-0.15	0.32	-0.762†	0.41	-0.36	0.37	
Hispanic	-0.15	0.22	-0.04	0.22	-0.40	0.45	-0.09	0.35	0.06	0.41	-0.21	0.41	
Other non-Hispanic	0.07	0.30	0.21	0.37	-0.27	0.61	-0.31	0.50	-0.90	0.56	-1.599*	0.68	
Parents are a different race/ethnicity	0.18	0.20	0.12	0.23	-0.20	0.48	-0.20	0.37	0.81	0.45	0.822+	0.44	
Mother's Education													
Less than High School (reference)													
High school diploma or equivalent	0.15	0.28	0.43	0.32	0.14	0.51			0.67	0.46			
Some college	0.837**	0.30	-0.02	0.34	0.15	0.51			0.52	0.46			
Bachelor's degree or higher	0.601+	0.32	-0.08	0.39	-0.24	0.52			0.19	0.45			
Father's Education													
Less than High School (reference)													
High school diploma or equivalent	0.45	0.28	-0.536†	0.29			0.54	0.41			0.28	0.57	
Some college	0.03	0.28	-0.706*	0.28			-0.39	0.42			0.24	0.59	
Bachelor's degree or higher	-0.51	0.34	-1.319***	0.34			0.15	0.49			0.59	0.60	
Health Status (1-Year)													
Mother's Health	-0.266**	0.10	-0.184+	0.10	-0.18	0.15			-0.22	0.18			
Father's Health	-0.11	0.10	0.07	0.10			-0.22	0.13			-0.686***	0.18	
Child Health	0.02	0.13	0.11	0.16	0.15	0.20	-0.05	0.22	-0.452*	0.22	-0.425†	0.24	
Employment Status (Year 1)	0.02	0.15	0.11	0.10	0.125	0.20	0.02	0.22	0.102	0.22	0.120	0.21	
Mother has job = 1	-0.19	0.18	-0.17	0.18	-0.06	0.25			-0.341†	0.31			
Father has job $= 1$	-0.01	0.23	0.18	0.10	0.00	0.25	-0.27	0.43	0.541	0.51	-0.27	0.44	
Social Support (Vear 1)	0.01	0.25	0.10	0.52			0.27	0.45			0.27	0.44	
Mother received support	0.26	0.17	0.487*	0.19	0.46†	0.27			0 828**	0.30			
Eather received support	0.430*	0.17	0.437	0.15	0.401	0.27	0.00	0.34	0.020	0.50	0.14	0.33	
Incarceration History (1-Vear)	0.450	0.17	0.17	0.21			0.00	0.54			0.14	0.55	
Eather	0.30	0.27	-0.08	0.34	-0.30	0.43	0.27	0.38	0.42	0.51	0.22	0.60	
Domestic Violent	0.50	0.27	-0.00	0.54	-0.50	0.45	0.27	0.56	0.42	0.51	0.22	0.00	
Father	-0.25	0.43	0.04	0.48	-2 32**	0.86	0.04	0.76	-1.09	0.74	-13 098***	1.28	
Number of children in home < 18 years old	0.186*	0.08	0.04	0.43	-0.21	0.12	0.281*	0.14	0.03	0.13	-13.098	0.17	
Fortility History	0.100	0.00	0.05	0.10	-0.21	0.12	0.201	0.14	0.05	0.15	0.04	0.17	
Couple Higher Order Birth (Ves-1)	-0.04	0.20	0.25	0.23	0.38	0.30	0.38	0.32	0.20	0.35	0.627‡	0.36	
Multiportpored Fortility	-0.04	0.20	0.23	0.25	0.56	0.50	0.50	0.52	0.20	0.55	-0.027	0.50	
Neither parent has a child by another partner (reference)												
Fother has a shild by another partner	0.252+	0.21	0.21	0.20	0 652*	0.22	0.07	0.20	0.966*	0.42	0.44	0.46	
Mother has a shild by another partner	0.5521	0.21	-0.31	0.29	0.035*	0.35	-0.07	0.39	-0.800*	0.45	-0.44	0.40	
Notifier has a child by another partner	0.15	0.22	-0.736*	0.52	0.0571	0.39	-0.14	0.45	-0.51	0.44	0.10	0.39	
Bolow payerty line = 1 (1 Vace)	0.04	0.48	-0.52	0.37	-0.20	0.48	-0.00	0.00	0.05	0.08	1.24	0.95	
below poverty line = $1(1 - 1 \text{ ear})$	0.21	0.21	-0.55	0.24	0.07	0.57	-0.02	0.34	-0.04	0.45	0.00	0.41	

Table 2.10: Control Variables for the Longitudinal and Reciprocal Effects (Marrieds Only)

	Econ. Hardship (3-Year) Mother Report		Econ. Hardship (5-Year) Mother Report			Dep. Svm	1. (3-Year)	Dep. Svm. (5-Year)				
					Mothers		Fathers		Mothers		Fathers	
Control Variables	b	se	b	se	b	se	b	se	b	se	b	se
Dep. Sym (Lagged)					0.311***	0.04	0.442***	0.08	0.354***	0.06	0.540***	0.08
Econ. Hardship (Lagged)	0.363***	0.05	0.401***	0.04								
Age at child's birth												
Mothers	-0.02	0.01	-0.02	0.02	-0.03	0.02			0.01	0.03		
Fathers	0.00	0.01	0.02	0.01			-0.01	0.02			0.01	0.02
Mother's race/ethnicity												
White non-Hispanic (reference)	0.0101				0.00		0.45		0.40			
Black non-Hispanic	-0.313*	0.16	-0.05	0.14	0.08	0.29	-0.47	0.31	-0.49	0.31	0.29	0.37
Hispanic	-0.322*	0.16	-0.325*	0.15	-0.28	0.28	0.11	0.32	-0.781*	0.32	0.23	0.44
Other non-Hispanic	-0.07	0.35	0.28	0.31	-1.26	0.93	0.34	0.74	-0.21	0.81	-1.23	0.75
Parents are a different race/ethnicity	0.04	0.17	0.02	0.14	0.60*	0.27	0.01	0.31	-0.07	0.35	-0.01	0.39
Mother's Education												
Less than High School (reference)	0.12	0.14	0.15	0.14	0.01	0.04			0 < 15*	0.22		
High school diploma or equivalent	0.12	0.14	0.15	0.14	0.01	0.26			0.645*	0.32		
Some college	0.324*	0.16	0.24	0.16	0.27	0.27			0.5997	0.36		
Bachelor's degree or higher	-0.189	0.40	0.06	0.35	-1.23	0.91			2.127***	0.57		
Father's Education												
Less than High School (reference)	0.16	0.14	0.02	0.14			0.20	0.27			0.401	0.21
High school diploma or equivalent	0.16	0.14	0.02	0.14			0.30	0.27			-0.401	0.31
Some college	0.14	0.17	0.23	0.16			0.22	0.31			-0.57	0.44
Bachelor's degree or higher	-0.42	0.37	0.16	0.39			-13.30***	1.10			-10.90*	5.51
Health Status (1-Year)	0.02	0.07	0.00	0.07	0.01	0.11			0.11	0.12		
Mother's Health	-0.03	0.06	-0.09	0.06	0.01	0.11	0.01	0.12	-0.11	0.12	0.11	0.14
Father's Health	0.05	0.07	-0.05	0.06	0.272**	0.12	-0.01	0.13	0.10	0.16	-0.11	0.14
Child Health	-0.121	0.07	0.04	0.08	-0.3/3**	0.13	0.10	0.15	-0.18	0.16	-0.03	0.17
Employment Status (Year 1)	0.04	0.10	0.14	0.12	0.24	0.01			0.000*	0.07		
Mother has $job = 1$	-0.06	0.12	-0.14	0.12	-0.26	0.21	0.015**	0.27	-0.692*	0.27	0.12	0.20
Father has $job = 1$	-0.01	0.13	-0.205 T	0.12			-0.815**	0.27			-0.13	0.30
Social Support (Year 1)	0.0444		0.45		0.000							
Mother received support	0.34**	0.12	0.17	0.11	0.098	0.22			0.448†	0.24		
Father received support	0.14	0.12	0.15	0.11			0.28	0.23			0.44	0.28
Incarceration History (1-Year)	0.05.4*	0.10	0.01	0.12	0.24	0.22	0.27	0.04	0.25	0.05	0.000*	0.00
Father Demostic Michael	0.254*	0.12	0.01	0.13	-0.24	0.22	0.27	0.26	0.35	0.25	0.603*	0.29
Domestic Violent	0 (0**	0.22	0.00	0.00	0.04	0.44	0.61	0.50	1 0214	0.61	1 0071	0.62
Father	-0.60**	0.22	0.08	0.22	0.26	0.44	-0.61	0.58	1.021+	0.61	-1.097	0.63
Number of children in home < 18 years old	0.14*	0.05	-0.05	0.06	-0.16	0.10	0.11	0.12	0.14	0.12	0.16	0.12
Fertility History	0.02	0.12	0.02	0.12	0.04	0.25	0.000*	0.27	0.07	0.20	0.4001	0.20
Couple Higher Order Birth (Yes= 1)	0.02	0.13	-0.03	0.13	0.34	0.25	0.680*	0.27	0.06	0.30	0.499+	0.30
Multipartnered Fertility	c)											
Neither parent has a child by another partner (r	eference)	0.17	0.20	0.14	0.155	0.22	0.57	0.24	0.744	0.25	0.04	0.40
Father has a child by another partner	-0.044	0.17	0.20	0.16	-0.155	0.32	0.57	0.36	0.766*	0.35	0.34	0.40
Nother has a child by another partner	-0.26	0.17	0.16	0.17	0.21	0.29	-0.15	0.31	-0.46	0.30	-0.5/	0.37
Both parents have a child by another partner	0.07	0.19	0.21	0.17	0.6/3*	0.29	0.15	0.36	-0.25	0.31	0.15	0.38
Below poverty line = $1(1 - Y ear)$	0.00	0.12	0.09	0.13	-0.12	0.25	-0.57	0.24	-0.12	0.27	-0.29	0.29

Table 2.11: Control Variables for the Longitudinal and Reciprocal Effects (Cohabitors Only)

Table 2.12: Control Variables on Relationship Distress

		All Co	uples		Married	Couples			Cohabiting Couples				
	Mother		Father		Moth	Mother		Father		Mother		Father	
Control Variables	В	SE	В	SE	В	SE	В	SE	В	SE	В	SE	
Dep. Sym. (Lagged)	.06*	0.03	0.028	0.03	0.085*	0.04	0.04	0.04	0.05	0.04	0.03	0.04	
Family Structure													
Married (reference)													
Cohabiting	0.02	0.04	0.12**	0.04	-	-	-	-	-	-	-	-	
Parent's Age at child's birth	-0.04	0.03	-0.07*	0.03	0.00	0.04	-0.02	0.04	-0.05	0.04	-0.093*	0.05	
Race/ethnicity													
White non-Hispanic (reference)													
Black non-Hispanic	0.11**	0.03	0.10**	0.04	0.04	0.04	0.15**	0.05	0.12**	0.05	0.05	0.06	
Hispanic	0.03	0.03	0.02	0.03	0.03	0.04	0.03	0.05	0.03	0.05	-0.01	0.06	
Other non-Hispanic	-0.02	0.03	0.07*	0.03	0.00	0.04	0.11**	0.04	-0.03	0.04	0.06	0.04	
Education													
Less than High School (reference)													
High school diploma or equivalent	0.02	0.03	-0.01	0.04	0.01	0.06	-0.13*	0.06	0.04	0.04	0.03	0.05	
Some college	0.03	0.04	0.00	0.04	0.03	0.07	-0.10	0.07	0.04	0.05	0.03	0.05	
Bachelor's degree or higher	0.04	0.04	-0.02	0.04	0.03	0.08	-0.12	0.08	0.07	0.04	-0.03	0.04	
Parent's Health Status (1-Year)	0.02	0.03	-0.02	0.03	-0.02	0.04	-0.01	0.04	0.03	0.04	-0.04		
Child's Health (1-Year)	-0.09**	0.03	-0.04	0.03	0.00	0.04	0.00	0.04	-0.14***	0.04	-0.07†	0.04	
Employment Status (Year 1)	0.01	0.03	0.03	0.03	0.05	0.04	0.01	0.04	-0.01	0.04	0.04	0.04	
Social Support (Year 1)	0.04	0.03	0.03	0.03	0.01	0.04	0.08*	0.04	0.06	0.04	-0.01	0.04	
Incarceration History (1-Year)													
Father	0.11***	0.03	0.07*	0.03	0.04	0.04	-0.01	0.05	0.14***	0.04	0.09*	0.04	
Domestic Violent (1-Year)													
Father	0.08**	0.03	0.07*	0.03	0.06	0.04	0.03	0.04	0.09*	0.04	0.09*	0.04	
Fertility History													
Couple Higher Order Birth (Yes=1)	-0.03	0.03	-0.03	0.03	-0.01	0.04	-0.04	0.04	-0.03	0.04	-0.05	0.04	
Multipartnered Fertility													
Neither parent has a child by another partner (reference)													
Father has a child by another partner	0.07**	0.03	-0.01	0.03	0.01	0.04	0.00	0.05	0.11**	0.04	-0.03	0.04	
Mother has a child by another partner	0.06*	0.03	0.05	0.03	0.08+	0.04	0.05	0.04	0.07+	0.04	0.05	0.04	
Both parents have a child by another partner	0.04	0.03	0.03	0.03	0.10*	0.04	0.03	0.04	0.03	0.04	0.03	0.05	
Below poverty line = 1 (1-Year)	0.01	0.03	-0.03	0.03	0.10*	0.04	-0.01	0.04	-0.03	0.04	-0.03	0.04	

CHAPTER III

Parents' Depressive Symptoms and Coparenting: Does Race & Ethnicity Matter?

INTRODUCTION

Supportive and cooperative coparenting relationships have several beneficial outcomes for parents and children. For example, coparenting is associated with children's adjustment (Baril, Crouter, and McHale 2007; Feinberg and Kan 2008) and parenting behavior (Carlson, McLanahan, and Brooks-Gunn 2008; Margolin, Gordis, and John 2001). Cooperative coparenting refers to the extent to which parents respect and support each other's parenting efforts (Feinberg 2003). Coparenting is especially important after the birth of a child as mothers and fathers begin to take on the joint enterprise of sharing parental responsibilities and duties, and learn to work together as a team to ensure the optimal well-being for their child and family. At the same time, however, the demands of having a young child increases parents' day-to-day stressors which often leads to higher levels of psychological distress, such as depressive symptoms (Umberson, Pudrovska, and Reczek 2010), which may infringe on supportive coparenting between parents. Conversely, supportive coparenting may reduce parents' psychological distress as partners work together and support one another in the parenting process. Although scholars have suggested that mental health is important to the coparenting relationship between parents (Feinberg 2003), few prior studies have empirically done so. Thus, the purpose of this chapter is to examine depressive symptoms and coparenting as a longitudinal and dyadic process among couples with a young child.

Family systems theory has long argued that individuals within families are interdependent (Cox and Paley 1997; O'Brien 2005), and families are often structured in

ways that parents serve as joint caregivers who are responsible for nurturing children—a line of reasoning that gave rise to coparenting research (McHale and Lindahl 2011). A more recent family theory such as the stress-spillover/crossover (hereafter stresscrossover) builds on the interdependence of individuals within families and focuses on the ways in which stress and distress compromises positive family functioning. Stresscrossover researchers also recognize that positive family processes can reduce stress and distress among family members (Larson and Almeida 1999; Neff and Karney 2007; Westman 2001). Stress crossover happens in two ways: (1) partner A's own stress or distress can negatively affect her/his own family functioning (*spillover*); and (2) partner B's stress can also adversely affect Partner A's family functioning (crossover). The fundamental crux of stress-crossover research highlights the importance of dyadic relationships holistically, illuminating stress and distress as both an intrapersonal and intra-dyadic phenomenon. Thus, taking a stress-crossover approach to the study of parents' post-birth psychological distress and coparenting can be fruitful for stress and family research.

Further, there are reasons to believe that the extent to which stress and distress affects family functioning may vary across race/ethnicity. Prior research suggests that race and ethnic minorities, when compared to Whites, are exposed to a greater number of social stressors (i.e., racism, discrimination, poverty, etc.), which have been linked to higher levels of psychological distress (Sternthal, Slopen, and Williams 2011). As a consequence, the effects of psychological distress may be more detrimental to the coparenting relationship for minority couples than for Whites. Moreover, given that minority parents also report higher levels of stress in the parenting role than Whites (Nomaguchi and House 2013), having a supportive co-partner may be more meaningful for race/ethnic minorities than Whites in reducing depressive symptoms. Although scholars have noted the potential differences between race and ethnic groups in coparenting (Feinberg 2003; McHale and Lindahl 2011), many studies have used homogeneous samples, making the ability to examine variability across race and ethnicity impossible (for an exception, see Carlson and Hognas 2011). Even more, the association between psychological distress and coparenting may be a functioning of the child's age because the needs of children change over time, and thus, parental distress and coparenting may also change (Feinberg 2003; McHale, Kuersten-Hogan, and Rao 2004). Thus, the current study addresses the following questions: (1) Do depressive symptoms lead to lower levels of supportive coparenting? (2) Is supportive coparenting associated with lower levels of depressive symptoms for both parents? Finally, (3) do the associations vary by race/ethnicity?

The current study extends previous research on the association between parents' depressive symptoms and supportive coparenting in several ways. First, data are used from three time points for parents over their child's early developmental years: infant, toddler, and preschool. In addition, dyadic methods are used to understand (a) how each parent's own depressive symptoms are linked to their own report of supportive coparenting (*spillover*) and (b) how each parent's depressive symptoms are linked to their partner's report of supportive coparenting (*crossover*). Second, this study investigates the reverse association between coparenting and depressive symptoms to understand whether coparenting lowers parents' psychological distress. Third, the analyses are executed separately by race/ethnicity to examine whether the aforementioned associations vary

between groups. Last, this study uses data from a diverse, nationally representative sample of urban births in the late 1990s, and thus the results reflect urban couples (married and cohabiting) during the first five-years after the birth of their child. All in all, this study addresses the longitudinal association between parents' depressive symptoms and coparenting as children develop in years 0 through 5 of their lives.

LITERATURE REVIEW

Theoretical Perspective

The present study draws on several theoretical perspectives and additional empirical studies that link psychological distress and dyadic functioning as a reciprocal and dyadic process. The association between factors are outlined in Figure 3.1. First, as noted with Path A, the expectation is that parents' depressive symptoms are associated with lower levels of coparenting for both parents, which draws on stress-spillover perspective (Neff and Karney 2007). One of the key assumptions of stress crossover research is that relationships do not exist in a vacuum. Rather, individuals within relationships are interdependent, and thus parental stressors play a vital part in hindering dyadic functioning (Larson and Almeida 1999; Neff and Karney 2007). Conversely, Path B shows that parents' cooperative coparenting is associated with lower levels of depressive symptoms for both parents. This path relies on research that suggests that being in a positive and supportive relationships provides emotional sustenance for parents, which leads to lower distress (e.g., Umberson and Montez 2010).

Paths C and D reflect the potential moderating factor of race/ethnicity (e.g., Helms 2013). These pathways suggest that given that racial and ethnic minorities are at an elevated risk for psychological distress, the effects of depressive symptoms on coparenting may be exacerbated for racial and ethnic minorities relative to Whites (Path C). Based on a similar logic, coparening may be more meaningful for minorities than for Whites in reducing psychological distress (Path D). The model also takes into account the extent to which these processes are stable or change over time as a function of children's developmental stages. Indeed, scholars have argued that parental stressors and the relationship between partners change as the needs of the child changes (e.g., Nomaguchi and Milkie 2003). Taken together, the current study captures the complexity of family life by highlighting psychological distress and coparenting as a dyadic process while simultaneously addressing variations across race and ethnicity, and across time. The conceptual model guides both the review of research (see below) and the statistical analyses.

Depressive Symptoms and Coparenting

Studies have well-documented that parents with young children are at an elevated risk for psychological distress such as depression (Evenson and Simon 2005; Umberson et al. 2010). The general premise is that the emotional and physical demands of parenting a young child generates elevated levels of stress that ultimately affects parents' mental health (Umberson et al. 2010). These consistent findings have important implications for the stability and quality of intimate relationships because depressed parents are more likely to display anger, and less likely to show support which consequently leads to conflict and distress between parents (Conger, Conger, and Martin 2010). Indeed, prior studies show that parents' depression affects family functioning such as marital quality (Beach et al. 2003). Limited attention, however, has been given to mental health and coparenting. Proponents of coparenting research suggest that parents' mental health may serve as an important factor that alters effective coparenting (Feinberg 2003). As a result,

scholars are beginning to pay attention to the empirical association between the effects of depressive symptoms on coparenting.

Prior studies give some support for the association between parents' depression and the coparenting relationship between partners. For example, Cabrera, Shannon, and La Taillade (2009), using data from the Early Childhood Longitudinal Study-Birth Cohort (ECLS-B), found that fathers' depressive symptoms were significantly related to higher levels of coparenting conflict (i.e., conflict with their partner about issues about their children). The effects, however, were not significant for mothers. In two studies using different samples of resident fathers, Bronte-Tinkew et al. (2007) and Bronte-Tinkew, Horowitz, and Scott (2009) found that paternal depression was significantly associated with lower levels of fathers' report of coparenting support. In another study of 115 married couples, Elliston et al. (2008) found that fathers' depressive symptoms were associated with withdrawal in coparenting; the effects, however, were not significant for mothers. Carlson and Hognas (2011), using data from the Fragile Families study, found mothers' and fathers' risk of depression was significantly associated with mothers' report of coparenting. Despite the few studies on depression and coparenting, the studies above point to the growing efforts to shine light on this important issue. Thus, I offer the following hypothesis for each parent: *higher depressive symptom levels will be* associated with lower cooperative coparenting levels for mothers and fathers.

In addition to the direct association between parents' own depressive symptoms and coparenting, there is reason to believe that depression in one parent may affect their partner's view of the coparenting relationship. This line of reasoning is rooted in the idea that stress or distress in one partner can influence the other partner's functioning in close relationships (Neff and Karney 2007). Thus, cross-partner association between one parent's depressive symptoms and the other parent's cooperative coparenting is not improbable. Few prior studies exist on the cross-partner association between depression and coparenting: one study found that fathers' depression decreased mothers' coparenting reports (Carlson and Hognas 2011). Additionally, another study found that mothers' depression was not significantly related to fathers' report of coparenting (Bronte-Tinkew et al. 2007). Notably, these prior studies did not explore depression and coparenting as a dyadic process; rather, partner's depression was used as a control variable in the regression-based analyses. Indeed, more empirical work is needed to understand the cross-partner association between depression and coparenting. All in all, I hypothesize that *higher depressive symptom levels of one partner will be associated with lower levels of cooperative coparenting from the other parent.*

<INSERT FIGURE 3.1>

Coparenting and Depressive Symptoms

Many scholars have painted a clear picture that psychological distress and other family processes have mutually influencing or reciprocal effects (e.g., depression and marital quality; Kouros and Cummings 2011). Thus, depressive symptoms and supportive coparenting may operate in similar ways. Although studies on the association between depressive symptoms and coparenting are increasing, few studies have examined the reverse association even though there are good reasons for doing so. For example, scholars argue that supportive and cooperative coparenting reflects parents working together to take care of the responsibilities for their child. As such, if mother-father coparenting relationship is supportive, and parents view each other as part of a team that contributes to the well-being of their child, each parent may feel a sense of appreciation, respect, and love (Cutrona 1996; Don and Mickelson 2012) leading to better psychological outcomes and parental adjustment (Thoits 2011; Umberson and Montez 2010) for couples with a young child.

In many ways, coparenting is similar to spousal/partner support in intimate relationships given the emotional sustenance and the subsequent positive health outcomes that partner support provides (Feinberg 2002, 2003). For example, in an intervention study of 169 heterosexual couples, Feinberg and Kan (2008) found that mothers (but not fathers) who participated in the intervention program that introduced couples to coparenting (i.e., mutual support strategies for positive joint parenting) had a significant decrease in depressive symptoms compared to the control group (who were not introduced to coparenting). As previously mentioned, prior studies have not examined the effects of coparenting on depressive symptoms explicitly; however, previous research highlights the importance of positive mother-father relationships (e.g., spousal/partner support) on parental psychological well-being (Edwards et al. 2012). Inferences from these studies are used to build a foundation for understanding how coparenting may affect parents' depressive symptoms. In a study of African American mothers, Edwards et al. (2012) found that greater support from fathers was associated with fewer maternal depressive symptoms. Other studies give support that social support lowers depressive symptoms (Bielawska-Batorowicz and Kossakowska-Petrycka 2006; Don and Mickelson 2012; Gremigni et al. 2011), and low levels of social support were significantly associated with higher levels of depression in mothers (Horwitz et al. 2007; O'Hara and Swain 1996). Thus, I offer the following hypothesis for each parent: *higher cooperative* coparenting levels will be associated with lower depressive symptom levels.
<INSERT FIGURE 3.2>

Race/Ethnicity, Parental Depression, and Coparenting

Race and ethnic variations in both mental health and family functioning have pushed scholars to address the complexity of such differences (and similarities). For instance, given that race and ethnic minorities are disproportionately exposed to social stressors (i.e., socioeconomic inequality, discrimination, etc.) relative to Whites, minorities tend to display higher levels of psychological distress (Brown et al. 2000; George and Lynch 2003; Williams, Neighbors, and Jackson 2003). Race and ethnic variation in parental depression is less clear. For example, studies on race and ethnic variations in parental depression after the birth of a child are inconclusive with some studies showing that minority parents, when compared to White parents, have lower rates of depression (Wei et al. 2008), higher rates (Howell et al. 2005; Liu and Tronick 2013), or no differences (Yonkers et al. 2001). Many of these studies, however, have focused largely on postpartum depression or small, non-representative samples.

Increasingly, with the availability of longitudinal population-based data, studies are beginning to address the prevalence and correlates of parents' depressive symptoms following the birth of a child through the child's preschool years. For example, in a study of depressive symptomology among fathers with infants, Bronte-Tinkew et al. (2007) found that African American and Hispanic fathers displayed higher levels of depressive symptoms compared to White fathers. In addition, in a sample of mothers, Turney (2012) found race/ethnic differences in depression chronicity such that Whites and African Americans were less likely to report no depression or intermittent depression compared to persistent depression; and, Hispanic mothers, compared to White mothers, were less likely to report persistent depression than no depression. Notably, the extent to which race and ethnic variations exist in depression or depressive symptoms is largely contingent upon the analytic sample and the measures used. Still, questions about race and ethnic variations in the association between depressive symptoms and coparenting remained unanswered.

In regards to race/ethnicity and family outcomes, studies show, compared to Whites, race and ethnic minorities are less likely to marry (McLoyd et al. 2000), more likely to divorce (Bryant et al. 2010), and display lower levels of relationship quality (Bulanda and Brown 2007). Given that a range of structural and interpersonal factors contribute to these disparities, scholars argue for a more culturally sensitive approach to the study of minority families to capture the lived experiences of these unions independently rather than comparisons vis-à-vis White families (Bryant et al. 2010; Burton 2010; Few 2007). Increased attention has been given to the need for a more culturally diverse understanding of coparenting (Cabrera, Shannon, and Jolley-Mitchell 2013; Feinberg 2003) as prior research has focused disproportionately on White, middle class couples (McHale and Lindahl 2011).

Although the investigation of coparenting among minority families have increased, these studies have focused largely on single-mother families and extended kinship (Dorsey, Forehand, and Brody 2007; Jones and Lindahl 2011; Jones et al. 2007). Yet, we know less about coparenting in two-parent minority families—although research is emerging. For example, in a study of 735 low-income Mexican American families, Cabrera et al. (2009) found that couple conflict was significantly related to coparenting conflict; and, coparenting conflict among mothers was associated with lowered parentchild interactions and lower level of parental warmth among fathers. In another study of 192 married African American couples, Riina and McHale (2012) found that stressors such as economic strain and discrimination lowered coparenting satisfaction. These studies lay important groundwork for the study of coparenting in two-parent minority families. Thus, the current study builds on these prior studies by taking a longitudinal and dyadic approach to understand psychological distress and coparenting across Black, Hispanic, and White couples.

The above review gives some indication that race/ethnicity matters in both parental depressive symptoms and coparenting. The extent to which the race and ethnic variations exist in the effects of depressive symptoms and coparenting is less known. Based on the logic that race/ethnic minorities experience a range of interpersonal stressors such as economic inequality and discrimination, these stressors may manifest themselves in mental health and subsequently affect family processes more so than Whites; thus, creating a difference in the association between groups. Alternatively, psychological distress can be detrimental to coparenting similarly across race and ethnicity, and thus no differences may emerge which suggest that Black, Hispanic, and White couples do not vary in the effects of psychological distress and cooperative coparenting. Given the limited research, the current analyses are largely exploratory and thus no hypothesis is offered.

Coparenting Relationships Over Time

As a child ages, the coparenting relationship between parents also change. The needs of a child during certain developmental stages may dictate to parents the nature of the coparenting relationship (Feinberg 2003; McHale et al. 2004). For example, the infant years may require parents to negotiate the child care duties (e.g., feeding, changing diapers, etc.) whereas the toddler to preschool years may be a time when parents begin to

set limits and rules for their child's behavior (Mangelsdorf, Laxman, and Jessee 2011). Relatedly, the extent to which parents work together as coparents may be contingent upon the coparenting division of labor (e.g., childcare duties; Feinberg 2003). For instance, mothers, when compared to fathers, are more involved in early child care responsibilities (Bianchi and Milkie 2010). Although mothers continue to outperform fathers in child care, fathers tend to increase their levels of involvement as their child gets older (Woodworth, Belsky, and Crnic 1996). Thus, a parent's mental health may affect the coparenting relationship more during certain developmental stages. That is, mothers' mental health may hinder cooperative coparenting during the infant to toddler years as mothers are usually the parent that gives primary care.

For fathers, poor mental health may affect fathers during the toddler to preschool years since fathers are more involved during this time period. Alternatively, the negative impact of depressive symptoms on cooperative coparenting may affect parents regardless of the child developmental stage. Indeed, the availability of longitudinal data allows for researchers to capture change and stability over time. The current study builds on prior longitudinal studies on coparenting (Schoppe-Sullivan et al. 2004) by examining the association between depressive symptoms and coparenting for parents when their child ages from infants to toddlers, and from toddlers to preschoolers.

Additional Factors

The analysis includes several control variables that are expected to be associated with both depressive symptoms and coparenting. For example, prior research has documented *marital status* differences (married versus cohabitors) in depression (Brown 2000) and coparenting (Hohmann-Marriott 2011). Parental *age* is associated with lower levels of depression (Mirowsky and Ross 2002) and increased coparenting (Bronte-Tinkew et al. 2007). Parent's *levels of education* is associated with fewer depressive symptoms (Lorant et al. 2003) and higher levels of cooperative coparenting (Stright and Bales 2003). Immigrant status is associated with depression (Wilmoth and Chen 2003) and coparenting (Carlson and Hognas 2011). Parents' positive *relationship quality* is associated with lower levels of depression and higher levels of coparenting (Carlson and Hognas 2010). Parent's *physical health* has been linked to poor mental health (Webb et al. 2008) and affects coparenting (Carlson et al. 2008). *Poverty* and *employment* increases depression (Dooley, Prause, and Ham-Rowbottom 2000) and lowers coparenting (Lindsey, Caldera, and Colwell 2005).

Social support is associated with higher levels of coparenting (Lindsey et al. 2005) and lower levels of depression (Thoits 2010). Fathers' incarceration history is associated with depression (Turney, Wildeman, and Schnittker 2012) and less coparenting (Carlson and Hognas 2011); parental *impulsivity* or self-control is related to lower levels of coparenting (Talbot and McHale 2004) and increases in depressive symptoms (Carlson et al. 2008). Studies have documented that *multipartnered fertility* is associated with increased depression (Bronte-Tinkew et al. 2009; Turney and Carlson 2011), and *the number of children* parents have together is also associated with depression (McLanahan and Adams 1987; Turney and Carlson 2011) and coparenting (Lindsey et al. 2005). Parents' *first birth* affects parents' mental health (Mirowsky and Ross 2002) as well as coparenting (Mchale 2007). Child characteristics such as *child's sex* matters in coparenting (McHale 1995) and *child's temperament* and is associated with less coparenting (Davis et al. 2009) and higher levels of parental depression (Hanington,

Ramchandani, and Stein 2010). Accordingly, these factors are adjusted for in the statistical analyses.

METHODS

Data

Data for this study are from the Fragile Families and Child Well-being Study (FFCW). The FFCW is a nationally representative, longitudinal study that follows an urban birth cohort of 4,898 children and their parents (3,712 unmarried and 1,186 married births) in 20 U.S. cities with populations of 200,000 or more. The FFCW is based on a stratified, multistage probability sample with an oversample of unmarried births in urban cities. The FFCW began 1998-2000 and contains 4,898 mothers and 3,830 fathers. At baseline, mothers were interviewed in person while in the hospital within 48 hours of the birth, and fathers were interviewed in person or by phone once he was located (for more detailed information, see Reichman et al. 2001). Parents were re-interviewed when the child was one, three, and five years of age. The response rate for eligible mothers and fathers at baseline was 86% and 78%, respectively. Subsequent 1-, 3-, and 5-year follow-ups yielded 90%, 88%, and 87% response rates for eligible mothers, and 74%, 72%, and 70% for eligible fathers (Bendheim-Thomas Center on Child Wellbeing 2008).

The sample includes couples (biological mothers and fathers of the focal child) who were living together (either married or cohabiting) at the baseline survey and participated in the subsequent surveys and has no missing values on the focal variables. All survey waves are used; however, the main analyses are based on the follow-up waves because the focal variables (i.e., depressive symptoms and coparenting) were measured at the 1-, 3-, and 5-year waves only. The selection criteria of the sample for the data analyses are parents who were either married or cohabiting from the 1-year to the 3-year surveys, and from the 3-year to the 5-year follow-up waves. This is done in order to maintain couples who were romantically involved consistently over all survey waves.

As a result of the selection criteria, of the couples living together at the 1-year follow-up (N = 2, 341), 347 cases were dropped (15%) from the sample because either the mother or father was not interviewed at the 3-year survey, 317 cases (14%) were dropped because the parents ended their relationship between the 1- and 3-year surveys, and 433 cases (18%) were dropped because information on depressive symptoms or coparenting were missing. This resulted in a sample of 1,244 couples (53%). For couples who were living together at the 3-year follow-up (N=2,032), 324 cases (16%) were dropped because either mother or father were not interviewed at the 5-year follow-up, 288 cases (14%) because the relationship ended between the 3- and 5-year waves, and 32 cases (2%) because of missing data on depressive symptoms and coparenting. This resulted in a sample of 1,388 (68%) couples who were romantically involved and consistently living together over 3- and 5-year follow-up surveys.

In addition, the analytic sample was separated into couples with the same race/ethnicity, namely Non-Hispanic Black, Hispanic, and Non-Hispanic White. For the couples in the 1- and 3-year follow surveys, 192 couples were either considered as "other" or were in mixed race relationships. These cases were ambiguous and so were removed from the analysis, resulting in a sample of 1,052 couples. For the couples in the 3- and 5-year surveys, 207 couples were either considered as "other" or were in mixed race relationships. These cases were deleted as "other" or were in mixed race relationships. These cases were deleted and resulted in a sample of 1,181 couples. For a more detailed description of the breakdown of race and ethnicity, see Figures 3.3 through 3.6.

In analyses of attrition, the excluded cases were more likely to be racial/ethnic minorities, somewhat younger, and had lower levels of education compared to parents who remain in the sample (see Tables 3.6 through 3.9). Comments about potential implications of attrition are in the Discussion section. To maximize sample size, multiple imputation was employed to impute missing data on the covariates, but not the focal variables for which Full Information Maximum Likelihood (FIML) was used (Acock et al. 2005; Allison 2002; Enders and Bandalos 2001). Ten multiple imputation data sets were constructed using imputation by chained equations in Stata and then the analyses were conducted and combined using Rubin's rules (Little and Rubin 2002) in Mplus. City sampling weights are used for the descriptive statistics to adjust for oversampling of nonmarital births but not for the analyses because the study controls for key characteristics associated with the weights (e.g., marital status at the birth of the child, age, race, and education; see Winship and Radbill 1994).

Measures

Depressive Symptoms

Depressive symptoms at each wave were assessed using the Composite International Diagnostic Interview-Short Form for Major Depression (CIDI-SF), which is a comprehensive, standardized instrument used to assess the presence of mental disorders as specified by the Diagnostic and Statistical Manual of Mental Disorders (DSM-III-R; American Psychiatric Association 1994). Respondents were asked the following stem questions: (a) "During the past 12 months, has there ever been a time when you felt sad, blue, or depressed for two or more weeks in a row? (b) "During the past 12 months, has there ever been a time lasting two weeks or more when you lost interest in most things like hobbies, work, or activities that usually give you pleasure?" Respondents who affirmed these questions were asked about the following symptoms: (1) "losing interest," (2) "feeling tired," (3) "changes in weight," (4) "trouble sleeping," (5) "trouble concentrating," (6) "feeling down," and (7) "thoughts about death." Each symptom is a dichotomous variable with the value of 1 indicating the presence of a symptom. For mothers, α reliability scores at the 1-, 3-, and 5-year follow-up waves were .90, .86, and .93, respectively. For fathers, α reliability scores at the 1-, 3-, and 5-year follow-up waves were .82, .86, and .86, respectively. The items were summed creating a range from 0 to 8 symptoms. Prior research using the FFCW study have examined depressive symptoms for both mothers and fathers (Bronte-Tinkew et al. 2009; Heflin and Iceland 2009).

Coparenting

Coparenting at the 1-, 3, and 5-year follow-up surveys was measured by asking each parent five items about how the parents work together in raising their child. These items are (1) "When (father/mother) is with (child), he/she acts like the father/mother you want for your child," (2) "You can trust (father/mother) to take good care of (child)," (3) "He/She respects the schedules and rules you make for (child)," (4) "He/She supports you in the way you want to raise (child)," and, (5) "You and (father/mother) talk about problems that come up with raising (child)," Responses are (1) "rarely true", (2) "sometimes true", and (3) "always true." For mothers, α reliability scores at the 1-, 3-, and 5-year follow-up waves were .58, .62, and .70, respectively. The items were summed with higher scores reflecting higher levels of cooperative coparenting.

Race/Ethnicity

Race and ethnicity was measured using mothers and fathers' own reports to the following question: "Which of these categories best describes your race?" There were four response categories: (1) Non-Hispanic White, (2) Non-Hispanic Black, (3) Hispanic, and (4) other. Both parents' report of race/ethnicity was combined to identify same race couples; many of the parental relationships were race/ethnic homogenous. As mentioned earlier, couples who were either considered as "other" or were in mixed race relationships were dropped from the sample which resulted in the construction of the following dummy variables: *White* (reference), *Black, and Hispanic*.

Additional Factors

Control variables were included in all statistical models (described above) that are expected to be associated with depressive symptoms and coparenting. Identical measures were used for both mothers and fathers. Each control variable is represented for both mothers and fathers at the baseline survey (unless otherwise specified). *Marital status* was measured by a dichotomous variable with (0) indicating married (reference group) and (1) indicating cohabitation. Mothers and fathers' *age* was measured (in years) as continuous variables. Mothers and fathers' *education* level was measured using four categories: (1) less than high school (reference), (2) high school or equivalent, (3) some college or tech training, and (4) college graduate or more. *Physical health* (measured at the 1-year follow-up for parents and child) was measured by asking parents the following question: "In general, how is your health?" Mothers reported child's health with responses ranging from (1) poor to (5) excellent. *Employment status* was measured with a dichotomous item indicating whether each parent "Did any regular work for pay last week?" Response were (0) no and (1) yes.

Social support was measured with a dichotomous question (0=no, 1=yes) asking both parents "since child was born, have you received any financial help or money from anyone other than [partner]?" Mothers reported on fathers' *incarceration history* (at the 1-year follow-up) indicating whether fathers have ever been in jail or prison (0=no, 1=yes). Parent's *impulsivity* was gauged using Dickman's (1990) impulsivity scale (6items) to capture the ability to have self-control; responses range from (1) strongly disagree to (4) strongly agree, with higher scores reflecting higher impulsivity. "Mothers reported the number of children in the household at the 1-year follow-up. Parents' fertility history was gauged with two separated measures: First, a measured was created to indicate whether the focal child is a higher order birth or first birth (0=first birth, 1=higher order birth). Second, a measured was created to indicate a series of dummy variables indicating *multipartnered fertility* (at 1-year follow-up) which reflects whether mothers and fathers reported having a child with another partner: neither parent has a child by another partner (reference), father has child by another partner only, mother has child by another partner only, and both parents has a child by another partner. Poverty status (at the baseline survey) was measured using the household income-to-needs ratio based on the official U.S. poverty thresholds from the Census Bureau (adjusted for household composition and year). The variable was dichotomized to indicate that a ratio of 1 or less reflects a family lived in poverty, and a ratio above 1 reflects a family lived above the poverty line (0=no poverty, 1=poverty). The study also takes into account additional child characteristics such as child's sex (boy=1) and temperament—which is measured by 6-items that gauges the difficulty of a child's temperament with higher scores indicating higher levels of difficult temperament.

Analytic Strategy

Structural equation modeling (SEM; Bollen 1989) was employed using Mplus 6.11 (Muthen and Muthen 2010) to estimate the hypothesized paths in Figures 3.1 and 3.2. SEM is a useful statistical technique for handling dyadic data within the context of Actor-Partner Interdependent Model (APIM) with distinguishable dyads (i.e., mothers and fathers; (Kenny, Kashy, and Cook 2006). In Figure 3.1, the analyses examine the direct effect of each parent's depressive symptoms on their own cooperative coparenting, and the effect of the other partner's depressive symptoms on their own cooperative coparenting. The outcome variables are standardized, and thus reflect the following interpretation: a unit change in the independent variable is associated with a standard deviation change in the endogenous variable. For the APIM for the effects of coparenting on depressive symptoms (Figure 3.2), the model was estimated using Negative Binomial Regression with Maximum Likelihood Estimation with robust standard errors. This was done to take into account the over-dispersion of both depressive symptoms as count variables. The parameter estimates are exponentiated; thus, the interpretation is as follows: a unit change in the independent variable is associated with a multiplicative change in the expected count in the dependent variable. To address whether the association between depressive symptoms and coparenting varies by race/ethnicity, the APIMs were both reestimated using multi-group SEM to allow the estimates for each group to be tested independently. This approach allows the parameter estimates to be analyzed separately for each race and ethnic group.

RESULTS

Descriptive Statistics

Table 3.1 presents the mean, percentages, and standard deviations for the demographic characteristics for couples by race and ethnicity (weighted using the city sampling weights). White couples were more likely to be married (89%) compared to African American (39%) and Hispanic (59%) couples. A larger percentage of white couples have a bachelor's degree (53% for mothers; 49% for fathers) whereas a larger portion African American couples have a high school diploma (47% for mothers; 42% for fathers), and a larger portion of Hispanic couples have less than a high school diploma (45% for mothers; 54% for fathers). On average, white couples were older than both African American and Hispanic couples. In addition, a majority of the Hispanic parents were born outside the U.S. (60%) while Black parents and White parents (over 80%) were born in the U.S.

<INSERT TABLE 3.1>

Table 3.2 shows the means for depressive symptoms and cooperative coparenting (weighted using city sampling weights) with significant differences evaluated by race and ethnicity using *t-tests* (noted by subscripts). Overall, both parents' depressive symptoms were similar at Year-1 and Year-5 but increased at the Year-3 follow-up. Coparenting remained relatively stable across survey waves for both mothers and fathers. On average, Black and Hispanic mothers displayed higher levels of depressive symptomology than white mothers at the 1- and 3-Year survey. Only Black mothers' and White mothers', however, were statistically different (subscript a). For fathers, depressive symptoms were significant different between Black fathers and White fathers at the 1-Year (subscript d) with Black fathers displaying higher depressive symptoms levels. At the 3-Year survey,

Black fathers and White fathers remained statistically different (subscript e), and Hispanic fathers were statistically different from White fathers (subscript f); in both cases, White fathers depressive symptoms levels were higher. In regards to coparenting, white mothers reported higher levels than Black and Hispanic mothers at Year-1 with White mothers reporting significantly higher coparenting levels (subscripts b and c). Hispanic fathers reported significantly higher levels of cooperative coparenting than Black fathers (subscript g) and White fathers (subscript c) at the 1-Year survey. The significant differences were consistent at the 3-Year survey (subscripts h and i). At the 5-Year survey, only Hispanic fathers and Black fathers were significantly different (subscript j) with Hispanic fathers reporting higher coparenting levels.

<INSERT TABLE 3.2>

Actor-Partner Interdependent Models

Depressive Symptoms and Coparenting

The first research question was whether parents' depressive symptoms were associated with lower levels of cooperative coparenting for couples during their child's early years (i.e., infant, toddler, and preschool). To address this question, the analyses were executed using actor-partner interdependence models. As Table 3.3, Panel A shows, with respect to depressive symptoms at Year-1 and coparenting at Year-3, a one-standard deviation increase in depressive symptoms was significantly associated with .09-standard deviation decrease in coparenting for mothers (actor effect). For fathers, a one-standard deviation increase in depressive symptoms was significantly but modestly associated with .06-standard deviation decrease in coparenting (actor effect). There were no significant partner effects. From Year-3 to Year-5, only fathers' depressive symptoms were significantly associated with .06-standard deviation decrease in coparenting for mothers (actor effect).

(partner effect). Taken together, the results demonstrate that depressive symptoms are associated with copariting among mothers and fathers. Also, the effects are consistent for mothers over Years 1 to 3 and Years 3 to 5 after the birth of a child. All in all, the results support the research hypothesis that depressive symptoms lower cooperative coparenting.

Coparenting and Depressive Symptoms

The second question addressed whether coparenting was associated with lower levels of depressive symptoms in couple dyads. Table 3.3, Panel B shows the association between coparenting and depressive symptoms over time. The results indicated that as coparenting (mothers' report) at Year 1 increases, depressive symptoms for mothers modestly decreased by a factor of .54 (p < .10; actor effects). The partner effects, however, yielded no statistical significant results. For fathers, there were significant partner and actor effects: when fathers' report higher levels of cooperative coparenting at Year-1, depressive symptoms decreased by a factor of .41 at Year-3 (p < .05); and, mothers' report of coparenting at Year-1was significantly associated with a reduction in fathers' depressive symptoms at Year-3 by a factor of .46 (p < .05). From Years 3 to 5, no statistically significant association emerged. These findings give some support for the research hypothesis that cooperative coparenting is associated with lower levels of depressive symptoms. Moreover, the results show that cooperative coparenting may be beneficial for fathers' mental health over Years 1 to 3 after a child's birth.

Race/Ethnicity, Depressive Symptoms, and Coparenting

To address the third question—whether the association between depressive symptoms and coparenting varies by race and ethnicity—the analyses were executed using multigroup SEM for the models encompassing depressive symptoms to cooperative coparenting, and cooperative coparenting to depressive symptoms (see Tables 3.4 and 3.5, respectively). In Table 3.4, the results show the effect of depressive symptoms on coparenting—for Years 1 to 3 and Years 3 to 5—across race and ethnicity. For Black couples, no statistically significant results emerged for the association between depressive symptoms and coparenting from Years 1 to 3. In regards to Years 3 to 5, fathers' depressive symptoms is associated with .09 standard deviation decrease in coparenting for mothers (partner effect) and a .09 standard deviation decrease in coparenting and depressive symptoms is presented. The results show that, for Black couples, as fathers' report of cooperative coparenting increases, depressive symptoms decrease by a factor of .23 (p < .05) from Year 1 to Year 3. In regards to Year-3 to Year-5, no significant association between coparenting and depressive symptoms emerged.

For Hispanic couples (see Table 3.4), mothers' depressive symptoms were significantly associated with a .16-standard deviation decrease in coparenting (actor effect). Similarly, fathers' depressive symptoms were significantly associated with .11 standard deviations in coparenting from Years 1 to 3 (actor effect). For Years 3 to 5, no significant effects emerged, however. For the association between cooperative coparenting and depressive symptoms (Table 3.5), as fathers' report of cooperative coparenting increases at Year-1, depressive symptoms for mothers at Year-3 marginally increase by a factor of 3.52 (p < .10; partner effect)—which is contrary to the research hypothesis. From Years 3 to 5, the findings were not statistically significant for Hispanic couples.

Turning to White couples, Table 3.4 shows that mothers' with higher levels of depressive symptoms was associated with a .16 standard deviation decrease in

cooperative coparenting. No significant effects emerged from Years 3 to 5. In Table 3.5, as mothers' report of higher levels of cooperative coparenting increases, mothers' depressive symptoms decrease by a factor of .31 (actor effect; p < .10); and, mothers' report of higher levels of cooperative coparenting was significantly associated with a decrease in depressive symptoms for fathers by a factor of .17 (partner effect; p < .001). The results for Years 3 to 5 were not statically significant.

In sum, the results show that race and ethnic variations exist in the longitudinal and dyadic association between depressive symptoms and cooperative coparenting. The variations, however, are largely contingent upon the age of the child. More specifically, depressive symptoms tend to lower coparenting for Hispanic and White couples from Year-1 to Year-3 whereas depressive symptoms affect coparenting for Black couples from Year-3 to Year-5. Moreover, the association between coparenting and depressive symptoms tends to matter more for Black and White couples (particularly fathers) than Hispanic couples.

DISCUSSION

The purpose of the present study was to examine the longitudinal association between (1) depressive symptoms and cooperative coparenting, (2) coparenting and depressive symptoms, and (3) whether race/ethnic variations emerged in the associations among couples living in urban cities after their child's birth ages one, three, and five. The findings corroborate prior studies that depressive symptoms are associated with levels of coparenting among parents with young children (Cabrera et al. 2009; Carlson and Hognas 2011; Elliston et al. 2008). The current study extends this line of research by examining the longitudinal association between parents' mental health and the extent to which mothers and fathers cooperatively co-parent with each other as children grow from

infants to toddlers; and, from toddlers to preschoolers. Overall, the results support the notion that stress-crossover in which parents' psychological distress has adverse effects on family functioning (Neff and Karney 2007). Specifically, higher levels of depressive symptoms were associated with decreased cooperative coparenting between parents.

With regards to the association between cooperative coparenting and parents' depressive symptoms, the findings give credence to the idea that when parents work together in supportive ways in the context of child rearing, parents' psychological distress decreases. Although the findings from the current study show a modest association for mothers, the finding for fathers are quite telling: that is, fathers' depressive symptoms are reduced when fathers give and receive coparenting support, especially as children age from infant to toddler. Indeed, prior research shows that giving and receiving support can be beneficial for mental health (Ko and Lewis 2011). These findings are particularly striking because mothers are more likely to be involved in early child care responsibilities than fathers (Bianchi and Milkie 2010). Being an active participant in early child rearing tasks can be beneficial to fathers' health, however (Eggebeen and Knoester 2001). Although scholars have suggested the potential association between coparenting and depression (Feinberg 2003), little empirical investigation has explored the association. The present study is one of the first to explicitly address the association between cooperative coparenting and depression. Moreover, the findings show evidence that coparenting can be helpful to parents' psychological well-being, especially fathers.

Turning to race and ethnicity, the relationship between depressive symptoms and coparenting varies between race and ethnic groups. The analyses were performed separately for each race and ethnic group because many families are organized around many cultural norms and scripts, and as a result, may respond differently (or similarly) to certain stressors and family processes (Dilworth-Anderson, Burton, and Johnson 1993). Although the results show that depressive symptomology matters in coparenting similarly for each group, the pattern of results indicates that the variation is contingent upon the developmental stage of the focal child. For example, depressive symptoms were associated with cooperative coparenting for Hispanic couples (mothers and fathers) and White couples (mothers only) during their child's infant to toddler years. For Black couples (particularly fathers), the association emerged from the toddler to preschool years.

One explanation for the variations in the effects for race and ethnic groups during certain developmental ages could be because some parents are more engaged with their children during later stages. For example, Black fathers are involved with preschoolers more than infants (Black, Dubowitz, and Starr 1999), monitor their children more (Hofferth 2003), and tend to hold more traditional family and gender ideologies (e.g., separate spheres; Toth and Xu 1999) than fathers from other race and ethnic groups. Thus, the temporal variation in the effects may reflect the following: as a child grows older, Black fathers want (and may be expected) to be more involved in the childrearing process but mental health interferes with actively engaging with their preschool-age child, especially in the context of coparenting. Simply put, depressive symptoms may be a hindrance in coparenting relationship during the preschool years—a time when Black fathers are more engaged with their children. Although there has been a limited amount of studies on race and ethnic variations in the association between mental health and coparenting, the present study corroborates a prior study that investigated depression and

coparenting among Mexican American families (Cabrera et al. 2009) as well as providing an extension to other racial groups.

Moreover, race and ethnic differences also emerged for couples in regards to the association between coparenting and depressive symptoms for the Year-1 to Year-3 analysis only. More specifically, cooperative coparenting plays an important role in decreasing psychological distress for both Black and White fathers (and modestly for White mothers). These findings point to the importance of social support in the parental role for some families. Contrary to expectations, Hispanic fathers' report of coparenting was associated with higher levels of mothers' depressive symptoms, though the association was modest. To speculate, this association may suggests that although Hispanic fathers are reporting high levels of coparenting, mothers may be involved in the lion's share of the parenting duties, and subsequently affecting mothers' psychological health, or, given that the expectation in certain Hispanic groups is that mothers take on much of the parenting role and thus fathers increase in coparenting may lead to psychological distress among mothers who see this as part of their sphere of influence. Nevertheless, the overall findings points to the importance of coparenting education (Fagan 2008)—not only for parents' relationships and children's well-being but also for parents' mental health. Moreover, the findings also highlight the importance of a culturally relevant approach to the study of families given the variability in the association between depressive symptoms and coparenting for different race and ethnic groups during particular child developmental stages.

Implications

In regards to implications for research, the results highlight family systems theory in general (Cox and Paley 1997), and stress-crossover association of psychological distress

112

and family functioning among couple dyads in particular (Neff and Karney 2007). The results also point to the importance of examining whether or not couples of different race and ethnic groups vary in the association between psychological distress and family processes, and in this case, cooperative coparenting—which points to the importance of culturally relevant research. Future research can move this line of inquiry forward by examining how depression affects coparenting as children get older. Moreover, other factors such as marital status, birth-order, and socioeconomic status may moderate the extent to which depressive symptoms affect supportive coparenting.

The results may also have policy implications. For example, in light of public policy that focuses on strengthening relationship quality among low-income couples to promote father involvement and favorable outcomes for children, the notion of coparenting has been somewhat ignored (McHale 2010). Yet, studies show that cooperative coparenting is beneficial to children's well-being (Palkovitz, Fagan, and Hull 2013) and paternal involvement (Carlson et al. 2008). The current study provides some evidence that cooperative coparenting may also improve parents' mental health over critical periods of child development. Thus, public policy efforts that focus on strengthening relationship quality between parents may also find it valuable to include ways to increase cooperative coparenting in an efforts to help parents work together to mutually care for their child.

Limitations

Although the current study provides valuable insight on the association between depression and coparenting, there are notable limitations. First, the analyses are based on a sample of urban couples after the birth of a child in the late 1990s; therefore, the inferences cannot be generalized to the larger population. Moreover, the sample was restricted to couples living together across survey years; therefore, less is known about couples who separate over time (i.e., single mothers and nonresident fathers). Second, as in many studies using survey data, the potential for missing variable bias should be noted. Even though several control variables are employed in the current study (see Tables 3.10 through 3.15), and the analysis takes into account the longitudinal design of these data, the threat of unobserved variables may bias the parameter estimates. Ideally, observational studies and studies with an experimental design would be beneficial in this regard. In addition, missing data is not random; for example, minorities and respondents with lower levels of socioeconomic status are more likely drop out of the study over time (see Tables 3.6 and 3.7). As such, coparenting may be overestimated and depressive symptoms may be underestimated as respondents who dropped out of the study tend to have lower levels of coparenting and higher levels of depressive symptomology. Lastly, given that coparenting is a multi-dimensional construct (Van Egeren and Hawkins 2004; Feinberg 2003), the present study only uses cooperative coparenting. Future studies may benefit using additional measures of the coparental relationship that tap into negative aspects of coparenting such as parental undermining.

Strengths

Despite the noted limitations, there are valuable strengths to the current analyses. First, the present study uses a large, diverse longitudinal sample of parents living in urban cities with a sizeable number of couples from the three largest race and ethnic groups in the U.S. Second, one of the strengths of the Fragile Families data is that both parents were interviewed, and thus, this study leverages data from both parents to understand the how depression and coparenting is linked for mothers and fathers which extends prior studies using reports from only one parent (Bronte-Tinkew et al. 2007; Carlson and Hognas

2011). Moreover, using Actor-Partner Interdependent Models provide an empirical approach to test the theoretical perspective that parenting is inherently dyadic (e.g., stress-crossover). Last, although coparenting researchers have noted that cooperative coparenting may affect parents' mental health (Feinberg 2003), the current study is one of the first to empirically test the association.

CONCLUSION

In conclusion, the current study explored the association between depressive symptoms and coparenting, coparenting and depressive symptoms, and whether the associations varied by race and ethnicity among an urban parents after a child's birth in the late 1990s. Using Actor-Partner Interdependent Models, the results show that the longitudinal association between depressive symptoms and coparenting matters in three distinct ways: (1) depressive symptoms is detrimental for coparenting, particularly for mothers, and (2) the link between coparenting and depressive symptoms may be more beneficial for fathers, and (3) race and ethnic variations emerged in the associations, particularly during specific child developmental stages. Future research can move this line of research further by understanding the association between psychological distress and coparenting as children enter into middle and high school.

REFERENCES

- Acock, Alan, Manfred M. H. van Dulmen, Katherine R. Allen, and Fred P. Piercy. 2005.
 "Contemporary and Emerging Research Methods in Studying Families." Pp. 59– 89 in *Sourcebook of Family Theory & Research*, edited by A. Acock, K. Allen, P. Dilworth-Anderson, and D. Klein V. 1. Bengtson. Thousand Oaks, CA: Sage.
- Allison, Paul D. 2002. "Missing Data: Quantitative Applications in the Social Sciences." British Journal of Mathematical and Statistical Psychology 55(1):193–96.
- Association, American Psychiatric. 1994. *Diagnostic and Statistical Manual of Mental Disorders*. Washington, DC: Author.
- Baril, Megan E., Ann C. Crouter, and Susan M. McHale. 2007. "Processes Linking Adolescent Well-Being, Marital Love, and Coparenting." *Journal of Family Psychology* 21(4):645.
- Beach, Steven RH, Jennifer Katz, Sooyeon Kim, and Gene H. Brody. 2003. "Prospective Effects of Marital Satisfaction on Depressive Symptoms in Established Marriages: A Dyadic Model." *Journal of Social and Personal Relationships* 20(3):355–71.
- Bendheim-Thomas Center on Child Wellbeing. 2008. Introduction to the Fragile Families Public Use Data. Princeton University.
- Bianchi, Suzanne M., and Melissa A. Milkie. 2010. "Work and Family Research in the First Decade of the 21st Century." *Journal of Marriage and Family* 72(3):705–25.
- Bielawska-Batorowicz, Eleonora, and Karolina Kossakowska-Petrycka. 2006.
 "Depressive Mood in Men after the Birth of Their Offspring in Relation to a Partner's Depression, Social Support, Fathers' Personality and Prenatal Expectations." *Journal of Reproductive and Infant Psychology* 24(1):21–29.
- Black, Maureen M., Howard Dubowitz, and Raymond H. Starr Jr. 1999. "African American Fathers in Low Income, Urban Families: Development, Behavior, and Home Environment of Their Three-Year-Old Children." *Child Development* 70(4):967–78.
- Bollen, Kenneth A. 1989. Structural Equations with Latent Variable. New York: Wiley.
- Bronte-Tinkew, J., A. Horowitz, and M. E. Scott. 2009. "Fathering With Multiple Partners: Links to Children's Well-Being in Early Childhood." *Journal of Marriage and Family* 71(3):608–31.
- Bronte-Tinkew, Jacinta, Kristin A. Moore, Gregory Matthews, and Jennifer Carrano. 2007. "Symptoms of Major Depression in a Sample of Fathers of Infants: Sociodemographic Correlates and Links to Father Involvement." *Journal of Family Issues* 28(1):61–99.

- Brown, Susan L. 2000. "The Effect of Union Type on Psychological Well-Being: Depression among Cohabitors versus Marrieds." *Journal of Health and Social Behavior* 41(3):241–55.
- Brown, Tony N. et al. 2000. "Being Black and Feeling Blue': The Mental Health Consequences of Racial Discrimination." *Race and Society* 2(2):117–31.
- Bryant, Chalandra M. et al. 2010. "Race Matters, Even in Marriage: Identifying Factors Linked to Marital Outcomes for African Americans." *Journal of Family Theory & Review* 2(3):157–74.
- Bulanda, Jennifer Roebuck, and Susan L. Brown. 2007. "Race-Ethnic Differences in Marital Quality and Divorce." *Social Science Research* 36(3):945–67.
- Burton, Linda M., Eduardo Bonilla-Silva, Victor Ray, Rose Buckelew, and Elizabeth Hordge Freeman. 2010. "Critical Race Theories, Colorism, and the Decade's Research on Families of Color." *Journal of Marriage and Family* 72(3):440–59.
- Cabrera, Natasha J., Jacqueline D. Shannon, and Stephanie Jolley-Mitchell. 2013. "Coparenting in Latino Families." Pp. 9–25 in *Gender Roles in Immigrant Families*. Springer.
- Cabrera, Natasha J., Jacqueline D. Shannon, and Jaslean J. La Taillade. 2009. "Predictors of Coparenting in Mexican American Families and Links to Parenting and Child Social Emotional Development." *Infant mental health journal* 30(5):523–48.
- Carlson, Marcia J., and Robin S. Hognas. 2011. "Coparenting in Fragile Families: Understanding How Parents Work Together after a Nonmarital Birth." Pp. 81– 103 in *Coparenting: A conceptual and clinical examination of family systems*, edited by James P. McHale and Kristin M. Lindahl. Washington, DC: APA.
- Carlson, Marcia J., Sara S. McLanahan, and Jeanne Brooks-Gunn. 2008. "Coparenting and Nonresident Fathers' Involvement with Young Children after a Nonmarital Birth." *Demography* 45(2):461–88.
- Conger, Rand D., Katherine J. Conger, and Monica J. Martin. 2010. "Socioeconomic Status, Family Processes, and Individual Development." *Journal of Marriage and Family* 72(3):685–704.
- Cox, Martha J., and Blair Paley. 1997. "Families as Systems." Annual Review of Psychology 48(1):243–67.
- Cutrona, Carolyn E. 1996. Social Support in Couples: Marriage as a Resource in Times of Stress. Sage Publications, Inc.
- Davis, Evan F., Sarah J. Schoppe-Sullivan, Sarah C. Mangelsdorf, and Geoffrey L. Brown. 2009. "The Role of Infant Temperament in Stability and Change in

Coparenting across the First Year of Life." *Parenting: Science and Practice* 9(1-2):143–59.

- Dilworth-Anderson, Peggye, Linda M. Burton, and Leanor Boulin Johnson. 1993. "Reframing Theories for Understanding Race, Ethnicity, and Families." Pp. 627– 49 in *Sourcebook of family theories and methods*. Springer.
- Don, Brian Paul, and Kristin D. Mickelson. 2012. "Paternal Postpartum Depression: The Role of Maternal Postpartum Depression, Spousal Support, and Relationship Satisfaction." *Couple and Family Psychology: Research and Practice* 1(4):323.
- Dooley, David, Joann Prause, and Kathleen A. Ham-Rowbottom. 2000.
 "Underemployment and Depression: Longitudinal Relationships." *Journal of Health and Social Behavior* 41(4):421–36.
- Dorsey, Shannon, Rex Forehand, and Gene Brody. 2007. "Coparenting Conflict and Parenting Behavior in Economically Disadvantaged Single Parent African American Families: The Role of Maternal Psychological Distress." *Journal of Family Violence* 22(7):621–30.
- Edwards, Renee C. et al. 2012. "Supportive Relationships and the Trajectory of Depressive Symptoms among Young, African American Mothers." *Journal of Family Psychology* 26(4):585.
- Van Egeren, Laurie A., and Dyane P. Hawkins. 2004. "Coming to Terms with Coparenting: Implications of Definition and Measurement." *Journal of Adult Development* 11(3):165–78.
- Eggebeen, David J., and Chris Knoester. 2001. "Does Fatherhood Matter for Men?" *Journal of Marriage and Family* 63(2):381–93.
- Elliston, Donna, James McHale, Jean Talbot, Meagan Parmley, and Regina Kuersten-Hogan. 2008. "Withdrawal from Coparenting Interactions during Early Infancy." *Family process* 47(4):481–99.
- Enders, Craig K., and Deborah L. Bandalos. 2001. "The Relative Performance of Full Information Maximum Likelihood Estimation for Missing Data in Structural Equation Models." *Structural Equation Modeling* 8(3):430–57.
- Evenson, Ranae J., and Robin W. Simon. 2005. "Clarifying the Relationship between Parenthood and Depression." *Journal of Health and Social Behavior* 46(4):341– 58.
- Fagan, Jay. 2008. "Randomized Study of a Prebirth Coparenting Intervention with Adolescent and Young Fathers." *Family Relations* 57(3):309–23.
- Feinberg, Mark E. 2002. "Coparenting and the Transition to Parenthood: A Framework for Prevention." *Clinical Child and Family Psychology Review* 5(3):173–95.

- Feinberg, Mark E. 2003. "The Internal Structure and Ecological Context of Coparenting: A Framework for Research and Intervention." *Parenting: Science and Practice* 3(2):95–131.
- Feinberg, Mark E., and Marni L. Kan. 2008. "Establishing Family Foundations: Intervention Effects on Coparenting, Parent/infant Well-Being, and Parent-Child Relations." *Journal of Family Psychology* 22(2):253.
- Few, April L. 2007. "Integrating Black Consciousness and Critical Race Feminism into Family Studies Research." *Journal of Family Issues* 28(4):452–73.
- George, Linda K., and Scott M. Lynch. 2003. "Race Differences in Depressive Symptoms: A Dynamic Perspective on Stress Exposure and Vulnerability." *Journal of health and social behavior* 353–69.
- Gremigni, Paola, Laura Mariani, Valentina Marracino, Andrea Luigi Tranquilli, and Angelo Turi. 2011. "Partner Support and Postpartum Depressive Symptoms." *Journal of Psychosomatic Obstetrics & Gynecology* 32(3):135–40.
- Hanington, Lucy, Paul Ramchandani, and Alan Stein. 2010. "Parental Depression and Child Temperament: Assessing Child to Parent Effects in a Longitudinal Population Study." *Infant Behavior and Development* 33(1):88–95.
- Heflin, Colleen M., and John Iceland. 2009. "Poverty, Material Hardship, and Depression." *Social Science Quarterly* 90(5):1051–71.
- Helms, Heather M. 2013. "Marital Relationships in the Twenty-First Century." Pp. 233– 54 in *Handbook of Marriage and the Family*. Springer. Retrieved October 7, 2014 (http://link.springer.com/chapter/10.1007/978-1-4614-3987-5_11).
- Hofferth, Sandra L. 2003. "Race/Ethnic Differences in Father Involvement in Two-Parent Families Culture, Context, or Economy?" *Journal of Family Issues* 24(2):185–216.
- Hohmann-Marriott, Bryndl. 2011. "Coparenting and Father Involvement in Married and Unmarried Coresident Couples." *Journal of Marriage and Family* 73(1):296–309.
- Horwitz, Sarah McCue, Margaret J. Briggs-Gowan, Amy Storfer-Isser, and Alice S. Carter. 2007. "Prevalence, Correlates, and Persistence of Maternal Depression." *Journal of Women's Health* 16(5):678–91.
- Howell, Elizabeth A., Pablo A. Mora, Carol R. Horowitz, and Howard Leventhal. 2005.
 "Racial and Ethnic Differences in Factors Associated with Early Postpartum Depressive Symptoms." *Obstetrics & Gynecology* 105(6):1442–50.
- Jones, Deborah J., and Kristin M. Lindahl. 2011. "Coparenting in Extended Kinship Systems: African American, Hispanic, Asian Heritage, and Native American Families."

- Jones, Deborah J., Alecia A. Zalot, Sarah E. Foster, Emma Sterrett, and Charlene Chester. 2007. "A Review of Childrearing in African American Single Mother Families: The Relevance of a Coparenting Framework." *Journal of Child and Family Studies* 16(5):671–83.
- Kenny, David A., Deborah A. Kashy, and William L. Cook. 2006. *Dyadic Data Analysis*. New York: The Guilford Press.
- Ko, Linda K., and Megan A. Lewis. 2011. "The Role of Giving and Receiving Emotional Support in Depressive Symptomatology among Older Couples: An Application of the Actor-Partner Interdependence Model." *Journal of social and personal relationships* 28(1):83–99.
- Kouros, Chrystyna D., and E. Mark Cummings. 2011. "Transactional Relations between Marital Functioning and Depressive Symptoms." *American Journal of Orthopsychiatry* 81(1):128–38.
- Larson, Reed W., and David M. Almeida. 1999. "Emotional Transmission in the Daily Lives of Families: A New Paradigm for Studying Family Process." *Journal of Marriage and the Family* 61(1).
- Lindsey, Eric W., Yvonne Caldera, and Malinda Colwell. 2005. "Correlates of Coparenting During Infancy"." *Family Relations* 54(3):346–59.
- Little, Roderick J. A., and Donald B. Rubin. 2002. *Statistical Analysis with Missing Data*. Hoboken, NJ: Wiley.
- Liu, Cindy H., and Ed Tronick. 2013. "Rates and Predictors of Postpartum Depression by Race and Ethnicity: Results from the 2004 to 2007 New York City PRAMS Survey (Pregnancy Risk Assessment Monitoring System)." *Maternal and child health journal* 17(9):1599–1610.
- Lorant, Vincent et al. 2003. "Socioeconomic Inequalities in Depression: A Meta-Analysis." *American journal of epidemiology* 157(2):98–112.
- Mangelsdorf, Sarah C., Daniel J. Laxman, and Allison Jessee. 2011. "Coparenting in Two-Parent Nuclear Families."
- Margolin, Gayla, Elana B. Gordis, and Richard S. John. 2001. "Coparenting: A Link between Marital Conflict and Parenting in Two-Parent Families." *Journal of Family Psychology* 15(1):3.
- Mchale, James P. 2007. "When Infants Grow up in Multiperson Relationship Systems." Infant mental health journal 28(4):370–92.
- McHale, James P. 1995. "Coparenting and Triadic Interactions during Infancy: The Roles of Marital Distress and Child Gender." *Developmental psychology* 31(6):985.

- McHale, James P. 2010. "Shared Child Rearing in Nuclear, Fragile, and Kinship Family Systems: Evolution, Dilemmas, and Promise of a Coparenting Framework."
- McHale, James P., Regina Kuersten-Hogan, and Nirmala Rao. 2004. "Growing Points for Coparenting Theory and Research." *Journal of Adult Development* 11(3):221–34.
- McHale, James P., and Kristin M. Lindahl. 2011. *Coparenting: A Conceptual and Clinical Examination of Family Systems*. American Psychological Association.
- McLanahan, Sara, and Julia Adams. 1987. "Parenthood and Psychological Well-Being." Annual Review of Sociology 13(5):237–57.
- McLoyd, Vonnie C., Ana Mari Cauce, David Takeuchi, and Leon Wilson. 2000. "Marital Processes and Parental Socialization in Families of Color: A Decade Review of Research." *Journal of Marriage and Family* 62(4):1070–93.
- Mirowsky, John, and Catherine Ross. 2002. "Depression, Parenthood, and Age at First Birth." *Social Science and Medicine* 54(8):1281–98.
- Muthen, Linda K., and Bengt O. Muthen. 2010. *Mplus User's Guide. Sixth Edition*. Los Angeles, CA: Muthen & Muthen.
- Neff, Lisa A., and Benjamin R. Karney. 2007. "Stress Crossover in Newlywed Marriage: A Longitudinal and Dyadic Perspective." *Journal of Marriage and Family* 69(3):594–607.
- Nomaguchi, Kei, and Amanda N. House. 2013. "Racial-Ethnic Disparities in Maternal Parenting Stress The Role of Structural Disadvantages and Parenting Values." *Journal of health and social behavior* 54(3):386–404.
- Nomaguchi, Kei M., and Melissa A. Milkie. 2003. "Costs and Rewards of Children: The Effects of Becoming a Parent on Adults' Lives." *Journal of Marriage & Family* 65(2):356–74.
- O'Brien, Marion. 2005. "Studying Individual and Family Development: Linking Theory and Research." *Journal of Marriage and Family* 67(4):880–90.
- O'Hara, Michael W., and Annette M. Swain. 1996. "Rates and Risk of Postpartum Depression-a Meta-Analysis." *International review of psychiatry* 8(1):37–54.
- Palkovitz, Rob, Jay Fagan, and James Hull. 2013. "Coparenting and Children's Well-Being." *Handbook of Father Involvement: Multidisciplinary Perspectives* 202.
- Reichman, Nancy E., Julien O. Teitler, Irwin Garfinkel, and Sara S. McLanahan. 2001. "Fragile Families: Sample and Design." *Children and Youth Services Review* 23(4/5):303–26.

- Riina, Elizabeth M., and Susan M. McHale. 2012. "The Trajectory of Coparenting Satisfaction in African American Families: The Impact of Sociocultural Stressors and Supports." *Journal of Family Psychology* 26(6):896.
- Schoppe-Sullivan, Sarah J., Sarah C. Mangelsdorf, Cynthia A. Frosch, and Jean L. McHale. 2004. "Associations between Coparenting and Marital Behavior from Infancy to the Preschool Years." *Journal of Family Psychology* 18(1):194.
- Sternthal, Michelle J., Natalie Slopen, and David R. Williams. 2011. "Racial Disparities in Health." *Du Bois Review: Social Science Research on Race* 8(01):95–113.
- Stright, Anne Dopkins, and Stephanie Stigler Bales. 2003. "Coparenting Quality: Contributions of Child and Parent Characteristics." *Family Relations* 52(3):232–40.
- Talbot, Jean A., and James P. McHale. 2004. "Individual Parental Adjustment Moderates the Relationship between Marital and Coparenting Quality." *Journal of Adult Development* 11(3):191–205.
- Thoits, Peggy A. 2010. "Stress and Health: Major Findings and Policy Implications." *Journal of Health and Social Behavior* 51(1 suppl):S41–S53.
- Thoits, Peggy A. 2011. "Mechanisms Linking Social Ties and Support to Physical and Mental Health." *Journal of Health and Social Behavior* 52(2):145–61.
- Toth, John F., and Xiaohe Xu. 1999. "Ethnic and Cultural Diversity in Fathers' Involvement A Racial/Ethnic Comparison of African American, Hispanic, and White Fathers." *Youth & Society* 31(1):76–99.
- Turney, Kristen, and Marcia J. Carlson. 2011. "Multipartnered Fertility and Depression among Fragile Families." *Journal of Marriage and Family* 73(3):570–87.
- Turney, Kristin. 2012. "Prevalence and Correlates of Stability and Change in Maternal Depression: Evidence from the Fragile Families and Child Wellbeing Study." *PLoS ONE* 7(9):1–9.
- Turney, Kristin, Christopher Wildeman, and Jason Schnittker. 2012. "As Fathers and Felons Explaining the Effects of Current and Recent Incarceration on Major Depression." *Journal of health and social behavior* 53(4):465–81.
- Umberson, Debra, and Jennifer Karas Montez. 2010. "Social Relationships and Health: A Flashpoint for Health Policy." *Journal of Health and Social Behavior* 51(1 suppl):S54–S66.
- Umberson, Debra, Tetyana Pudrovska, and Corinne Reczek. 2010. "Parenthood, Childlessness, and Well-Being: A Life Course Perspective." *Journal of Marriage and Family* 72(3):612–29.

- Webb, David A. et al. 2008. "Postpartum Physical Symptoms in New Mothers: Their Relationship to Functional Limitations and Emotional Well-Being." *Birth* 35(3):179–87.
- Wei, Guo, Linda B. Greaver, Stephen M. Marson, Cynthia H. Herndon, and James Rogers. 2008. "Postpartum Depression: Racial Differences and Ethnic Disparities in a Tri-Racial and Bi-Ethnic Population." *Maternal and Child Health Journal* 12(6):699–707.
- Westman, Mina. 2001. "Stress and Strain Crossover." Human Relations 54(6):717-51.
- Williams, David R., Harold W. Neighbors, and James S. Jackson. 2003. "Racial/ethnic Discrimination and Health: Findings from Community Studies." *American Journal of Public Health* 93(2).
- Wilmoth, Janet M., and Pei-Chun Chen. 2003. "Immigrant Status, Living Arrangements, and Depressive Symptoms among Middle-Aged and Older Adults." *The Journals* of Gerontology Series B: Psychological Sciences and Social Sciences 58(5):S305–S313.
- Winship, Christopher, and Larry Radbill. 1994. "Sampling Weights and Regression Analysis." *Sociological Methods & Research* 23(2):230–57.
- Woodworth, Sharon, Jay Belsky, and Keith Crnic. 1996. "The Determinants of Fathering during the Child's Second and Third Years of Life: A Developmental Analysis." *Journal of Marriage and the Family* 679–92.
- Yonkers, Kimberly A. et al. 2001. "Onset and Persistence of Postpartum Depression in an Inner-City Maternal Health Clinic System." *American Journal of Psychiatry* 158(11):1856–63.





Figure 3.1: Conceptual Model Linking Depressive Symptoms, Coparenting, and Race/Ethnicity over Time



Figure 3.2: Latino Ancestry among Mothers



Figure 3.3: Latino Ancestry among Fathers



Figure 3.4: Geographic Location for Mothers, by Race/Ethnicity



Figure 3.5: Geographic Location for Fathers, by Race/Ethnicity

	All Couples		Black Couples		Hispanic Couples		White Couples		
	Mean or %	SD	Mean or %	% SD Mean or % SD		SD	Mean or % SD		
Marital Status (Baseline)									
Married	67.02	38.63		58.78		89.92			
Cohabiting	32.98	61.40			41.20		10.10		
Parent's Age at birth									
Mother's Age	27.95	5.75	26.79	5.86	25.88	4.79	30.29	5.57	
Father's Age	30.40	6.67	29.76	6.87	28.35	6.12	32.43	6.41	
Mother's Education									
Less than HS	22.44		19.70		45.30		5.49		
HS Diploma	33.45		47.00		33.70	33.70		25.60	
Some College	18.41		26.40		16.20		15.50		
Bachelor's Degree or higher	25.70		6.91		4.77		53.40		
Father's Education									
Less than HS	25.35	25.35 21.60			53.90		4.35		
HS Diploma	26.56		41.90		20.90		22.30		
Some College	22.88		30.00		16.50		24.00		
Bachelor's Degree or higher	25.21		6.43	6.43		8.66		49.30	
Immigrant Status (Born in U.S.)									
Mother's Immigrant Status	0.74		0.85		0.39		0.96		
Father's Immigrant Status	0.69		0.83		0.40		0.85		
Relationship Quality									
Mothers Relationship Quality	4.24	0.84	4.09	0.81	4.04	0.98	4.48	0.66	
Fathers Relationship Quality	4.27	0.88	4.08	0.97	4.14	0.89	4.49	0.77	
Health Status									
Mother's Health	3.99	0.93	3.90	1.03	3.65	0.84	4.30	0.82	
Father's Health	4.04	0.92	4.09	0.82	3.74	1.03	4.26	0.81	
Child Health (1-Year)	4.53	0.74	4.58	0.72	4.27	0.90	4.71	0.52	
Employment Status									
Mother has job (1-Year)	50.70		59.80		38.40		55.40		
Father has job (1-Year)	89.43 77		77.30	77.30 9		92.90		93.60	
Social Support									
Mother social support	24.62		36.40		15.20		25.40		
Father social support	31.26		41.10		14.00	14.00 39.60			
Father's Incarceration History									
Eather Incarcerated	8.01		14.2		9.75		3.11		

Table 3.1: Weighted Descriptive Statistics by Race/Ethnicity (N = 1052)

TABLE 3.1 continued on next page

Table 3.1: Weighted I	Descriptive Statistics	by Race/Ethnicit	v (N = 1052)
radie din in eignieu z	esemptive statisties	of Itaco, Danner	J (1 1002)

	All Couples		Black Couples		Hispanic Couples		White Couples	
	Mean or %	SD	Mean or %	SD	Mean or % SD		Mean or %	SD
Parent's Impulsivity ^a								
Mother Impulsivity	1.92	0.57	1.96	0.55	2.01	0.64	1.82	0.50
Father Impulsivity	1.86	0.62	1.84	0.67	2.00	0.63	1.77	0.55
# of children in HH	1.12	1.30	1.50	1.48	1.21	1.31	0.83	1.11
Fertility History								
Higher order birth	54.12		55.80		54.90		52.50	
No MFP ^b	69.13		37.70		66.00		89.60	
Father only MFP	9.89		18.30		9.08		5.78	
Mother only MFP	10.97		19.00		14.90		3.07	
Both MFP	10.01		25.00		9.97		1.54	
Poverty Status								
Lives in poverty	19.74		30.90 34.60			1.36		
Child Characteristics								
Boy=1	60.38		57.60		68.30		55.60	
Child Temperament	2.46	0.69	2.51	0.80	2.62	0.68	2.31	0.59
Ν	1052		379		315		358	

Note: Variables are from baseline survey or 1-year follow-up survey unless otherwise noted. All variables are weighted by city sampling weights; however, the number of cases (N) is unweighted. ^aMulti-partnered fertility; ^bmeasured at the 3-year follow-up survey.

Table 3.2: Weighted Mean Differences

	All Couples		Black Couples		Hispanic Couples		White Couples	
	Mothers	Fathers	Mothers	Fathers	Mothers	Fathers	Mothers	Fathers
Variables	<u>.</u>							
Dep. Sym. Year 1	0.43	0.36	0.48	0.53 ^d	0.51	0.28 ^d	0.34	0.34
Dep. Sym. Year 3	0.70	0.52	1.03 ^a	0.32 ^e	0.77	0.43 ^f	0.45 ^a	0.71 ^{ef}
Dep. Sym. Year 5	0.40	0.32	0.41	0.34	0.35	0.31	0.43	0.33
Coparenting Year 1	2.86	2.89	2.83 ^b	2.87 ^g	2.83 ^c	2.91 ^g	2.90 ^{bc}	2.89
Coparenting Year 3	2.82	2.86	2.80	2.83 ⁱ	2.83	2.92 ^{hi}	2.83	2.84 ^h
Coparenting Year 5	2.81	2.90	2.83	2.86 ^j	2.81	2.92 ^j	2.81	2.90

Means with identical superscripts denote significant difference.
		Year 1 to	Year 3		Year 3 to Year 5					
Panel A: Dep. Symptoms> Coparenting ^a										
	Mothe	ers	Fathe	ers	Moth	ers	Fathers			
	b se		b	se	b	se	b	se		
Actor Effects	-0.09** 0.03		-0.05†	0.03	-0.02	0.03	-0.02	0.03		
Partner Effects	-0.004	0.03	0.02	0.03	-0.06*	0.03	-0.04	0.03		
		Year 1 to	Year 3		Year 3 to Year 5					
Panel B: Coparenting> Dep. Symptoms ^b										
	Mothe	ers	Fathe	ers	Moth	ers	Fatl	ners		
	exp(b)	t	exp(b)	t	exp(b)	t	exp(b)	t		
Actor Effects	0.54 †	-1.83	0.41*	-2.05	0.66	-1.47	0.84	-0.43		
Partner Effects	0.94	-0.16	0.46*	-2.19	0.80	0.58	0.66	-1.28		

Table 3.3: Parameter Estimates for the Association between Depressive Symptoms and Coparenting

***p < .001; **p < .01; *p < .05; †p < .10

Table 3.4: Standardized Coefficients Linking Depressive Symptoms to Coparenting, by Race/Ethnicity

Panel A: Dep. Symptoms --> Coparenting

Year 1 to Year 3

	Black Couples					Hispanic	Couples		White Couples					
	Mothers Fathers		Mothe	Mothers Fathers			Mothe	rs	Fathers					
	b	se	b	se	b	se	b	se	b	se	1	b	se	
Actor Effects	-0.02	0.05	-0.08	0.05	-0.15**	0.05	-0.11*	0.05	-0.16***	0.04	0.	02	0.05	
Partner Effects	-0.03	0.05	0.01	0.05	0.00	0.05	0.03	0.05	0.00	0.04	0.	04	0.05	

Panel B: Dep. Symptoms --> Coparenting

Year 3 to Year 5

	Black Couples					Hispanic	Couples	White Couples					
	Mothers		Mothers Fathers		Moth	Mothers		ers	Mother	rs	Fath	Fathers	
	b	se	b	se	b	se	b	se	b	se	b	se	
Actor Effects	0.00	0.05	-0.09†	0.05	-0.03	0.05	-0.01	0.05	-0.04	0.04	-0.03	0.05	
Partner Effects	-0.09*	0.04	-0.05	0.05	-0.06	0.05	-0.03	0.05	-0.02	0.04	-0.01	0.05	

***p < .001; **p < .01; *p < .05; †p < .10

Table 3.5: Exponentiated Estimates Linking Coparenting to Depressive Symptoms, by Race/Ethnicity

Panel A: Coparenting --> Dep. Symptoms

Year 1 to Year 3

	Black Couples				Hispanic Couples						White Couples					
	Mothers Fathers		Mot	Mothers		Fath	ers		Mot	hers		Fathers				
	exp(b)	t	exp(b)	t	exp(b)	t		exp(b)	t	exp	(b)	t		exp(b)	t	
Actor Effects	0.64	-1.07	0.23*	-2.00	0.36	-1.47		1.26	0.30	0.3	6†	-1.86		0.54	-0.63	
Partner Effects	0.57	-1.06	0.65	-1.10	3.56†	-1.10		0.59	-0.80	1.0)3	0.03		0.13***	-2.93	

Panel B: Coparenting --> Dep. Symptoms

Year 3 to Year 5

	Black Couples					Hispani	ic Couples		White Couples				
	Mothers		Fathers		Moth	ners	Fath	iers	Moth	ners	Fat	hers	
	exp(b)	t	exp(b)	t	exp(b)	t	exp(b)	t	exp(b)	t	exp(b)	t	
Actor Effects	0.64	-1.39	0.71	-1.34	1.14	0.61	0.76	-0.75	0.90	-0.15	0.87	-0.68	
Partner Effects	1.60	0.16	1.01	-0.03	0.89	-0.55	0.62	-1.58	0.86	-0.43	0.84	-0.76	

***p < .001; **p < .01; *p < .05; †p < .10

	Moth	ers	Fathe	ers
Variables	exp(b)	t	exp(b)	t
Coparenting	1.27	0.67	0.95	-0.14
Depressive Symptoms	1.05	0.76	1.00	0.02
Marital Status				
Married (reference)				
Cohabiting	1.10	0.43	1.89**	2.63
Parent's Age	1.02	0.89	0.99	-0.69
Race/Ethnicity				
Non-Hispanic White (reference)				
Non-Hispanic Black	1.21	0.72	0.82	-0.79
Hispanic	1.92*	2.47	1.46	1.52
Other	1.98	1.54	0.50	-1.12
Parent's Education				
Less than H.S. (reference)				
High School Diploma	0.78	-1.07	0.94	-0.28
Some College	0.78	-0.98	0.85	-0.64
Bachelor's Degree or higher	0.53+	-1.66	0.99	-0.02

Table 3.6: Attrition for Mothers and Fathers Between Years 1 - 3

Exponentiated coefficients

	Moth	ers	Fathers			
Variables	exp(b)	t	exp(b)	t		
Coparenting	1.16	0.49	0.64	-1.33		
Depressive Symptoms	0.99	-0.25	0.86 +	-1.94		
Marital Status						
Married (reference)						
Cohabiting	1.17	0.73	1.11	0.48		
Parent's Age	1.02	0.97	0.98	0.48		
Race/Ethnicity						
Non-Hispanic White (reference)						
Non-Hispanic Black	0.88	-0.51	1.09	0.36		
Hispanic	1.39	1.34	1.02	0.07		
Other	1.89 +	1.69	0.91	-0.17		
Parent's Education						
Less than H.S. (reference)						
High School Diploma	0.66+	-1.65	1.46+	1.78		
Some College	0.99	-0.06	0.59+	-1.87		
Bachelor's Degree or higher	0.69	-1.11	0.61	-1.34		

Table 3.7: Attrition for Mothers and Fathers Between Years 3 - 5

Exponentiated coefficients

	Mothe	ers	Fathers			
Variables	exp(b)	t	exp(b)	t		
Coparenting	0.40***	-3.83	0.45**	-2.72		
Depressive Symptoms	1.05	1.07	1.08	1.32		
Marital Status						
Married (reference)						
Cohabiting	2.82***	5.08	2.35***	3.88		
Parent's Age	0.95***	-3.86	0.97*	-2.24		
Race/Ethnicity						
Non-Hispanic White (reference)						
Non-Hispanic Black	1.70**	2.99	1.68*	2.45		
Hispanic	0.57*	-2.52	0.73	-1.31		
Other	0.97	-0.06	1.96+	1.66		
Parent's Education						
Less than H.S. (reference)						
High School Diploma	1.10	0.53	1.27	1.31		
Some College	1.04	0.18	0.75	-1.26		
Bachelor's Degree or higher	0.50+	-1.87	0.12***	-3.40		

Table 3.8: Relationship Ended Between Years 1 - 3

Exponentiated coefficients

	Mothe	ers	Father	rs
Variables	exp(b)	t	exp(b)	t
Coparenting	0.60*	-2.48	0.61+	-1.88
Depressive Symptoms	1.03	0.70	1.11*	2.18
Marital Status				
Married (reference)				
Cohabiting	3.35***	6.12	3.02***	5.64
Parent's Age	0.977+	-1.71	0.98*	-2.02
Race/Ethnicity				
Non-Hispanic White (reference)				
Non-Hispanic Black	1.57*	2.45	1.96**	3.25
Hispanic	0.74	-1.46	1.15	0.60
Other	1.15	0.35	0.50	-1.10
Parent's Education				
Less than H.S. (reference)				
High School Diploma	0.91	-0.54	1.13	0.69
Some College	0.73	-1.63	0.86	-0.75
Bachelor's Degree or higher	0.56+	-1.85	0.78	-0.78

Table 3.9: Relationship Ended Between Years 3 - 5

Exponentiated coefficients

	Y	Year-1 to Yea	ar-3 (N=1052)		Y	Year-3 to Ye	ar-5 (N=1181)	
_	Mothe	r	Father	r	Mothe	r	Father	
Control Variables	b	se	b	se	b	se	b	se
Coparenting (Lagged)	0.43***	0.03	0.31***	0.03	0.45***	0.03	0.36***	0.03
Cohabitation = 1	-0.01	0.04	-0.01*	0.04	-0.04	0.03	-0.01	0.04
Black	0.02	0.04	0.02	0.04	0.07†	0.03	-0.02	0.04
Hispanic	0.01	0.04	0.07†	0.04	0.03	0.04	-0.02	0.04
U.S. Citizen $= 1$	-0.09**	0.03	-0.03	0.03	0.03	0.03	-0.06†	0.03
HS Diploma	0.02	0.04	-0.08*	0.04	0.00	0.03	0.01	0.03
Some College	0.01	0.04	-0.10*	0.04	-0.03	0.04	-0.05	0.04
Bachelor's Degree or higher	-0.01	0.04	-0.09*	0.04	-0.02	0.04	-0.06	0.04
Parent's Age	0.00	0.03	0.04	0.04	-0.06†	0.03	-0.01	0.03
Employed = 1	0.03	0.03	-0.04	0.03	0.00	0.03	0.01	0.03
Social Support = 1	0.01	0.03	-0.03	0.03	-0.04†	0.03	0.00	0.03
Father has a child by another partner	-0.01	0.03	-0.03	0.03	0.02	0.03	0.02	0.03
Mother has a child by another partner	-0.01	0.03	0.04	0.03	-0.01	0.03	-0.02	0.03
Both parents have a child by another partner	0.00	0.03	0.02	0.03	-0.03	0.03	-0.08*	0.03
Father ever been incarcerated =1	-0.01	0.03	-0.04	0.03	-0.07*	0.03	0.00	0.03
Parent's Health	0.03	0.03	0.04	0.03	-0.05†	0.03	-0.01	0.03
Parent's Impulsivity	-0.02	0.03	-0.04	0.03	0.00	0.03	-0.05	0.03
Relationship Quality	0.12***	0.03	0.07*	0.03	0.12***	0.03	0.16***	0.03
Child's Health	0.07*	0.03	0.01	0.03	0.01	0.03	0.05†	0.03
Child is male = 1	0.02	0.03	0.02	0.03	0.02	0.03	-0.02	0.03
Child Temperament	-0.03	0.03	0.01	0.03	-0.01	0.03	-0.01	0.03
Poverty = 1	-0.01	0.03	-0.01	0.03	0.03	0.03	0.01	0.03

Table 3.10: Standardized Estimates for Control Variables on Coparenting at Year-3 and Year 5 (All Couples)

	Yea	ar-1 to Ye	ar-3 (N=1052))	Yea	Year-3 to Year-5 (N=1181)					
	Mothe	er	Fath	ner	Mothe	er	Fathe	r			
Control Variables	b	se	b	se	b	se	b	se			
Dep. Symptoms (Lagged)	0.38***	0.04	0.36***	0.06	0.38***	0.04	0.44***	0.06			
Cohabitation = 1	0.10	0.23	-0.11	0.23	0.14	0.26	0.33	0.27			
Black	0.24	0.24	-0.33	0.24	-0.37	0.25	0.01	0.28			
Hispanic	0.02	0.31	-0.13	0.29	-0.07	0.30	0.19	0.36			
U.S. Citizen = 1	0.41	0.30	0.05	0.30	-0.01	0.28	0.50	0.37			
HS Diploma	-0.16	0.26	0.54*	0.25	0.81**	0.26	0.15	0.28			
Some College	0.13	0.29	0.16	0.27	1.09***	0.28	-0.11	0.37			
Bachelor's Degree or higher	-0.07	0.41	-0.33	0.40	1.24**	0.37	0.23	0.45			
Parent's Age	0.01	0.02	-0.03ł	0.02	0.01	0.02	0.02	0.02			
Employed = 1	-0.51**	0.18	-0.58*	0.28	-0.33ł	0.20	-0.13	0.27			
Social Support = 1	0.22	0.18	0.28	0.20	0.81***	0.19	0.31	0.22			
Higher order birth = 1	0.08	0.19	0.82***	0.19	0.19	0.21	0.22	0.25			
Father has a child by another partner	0.82**	0.24	0.18	0.28	0.59*	0.25	0.03	0.30			
Mother has a child by another partner	0.40	0.25	0.25	0.27	-0.18	0.25	-0.53	0.40			
Both parents have a child by another partner	0.64*	0.26	0.12	0.30	0.48	0.31	0.53	0.34			
Father ever been incarcerated =1	-0.45ł	0.26	0.08	0.23	0.53*	0.23	0.40	0.27			
Parent's Health	0.60	0.13	-0.09	0.10	0.01	0.10	-0.33**	0.11			
Parent's Impulsivity	0.60***	0.13	0.55***	0.15	0.54**	0.16	0.08	0.19			
Relationship Quality	-0.01	0.10	0.25*	0.11	0.03	0.11	-0.32**	0.12			
Child's Health	-0.34**	0.11	0.15	0.15	-0.29*	0.11	-0.09	0.13			
Child Temperament	-0.15	0.12	-0.16	0.13	0.03	0.12	-0.18	0.15			
Poverty = 1	-0.02	0.22	-0.50*	0.23	0.47**	0.22	0.00	0.27			

Table 3.11: Negative Binomial Estimates for Control Variables on Depressive Symptoms at Year-3 and Year 5 (All Couples)

	Black Couples					Hispar	nic Couples			Whit	e Couples	
	Moth	ers	Fathe	rs	Moth	ers	Fath	ers	Mothe	ers	Fathe	ers
Variables	b	se	b	se	b	se	b	se	b	se	b	se
Coparenting (Lagged)	0.47***	0.05	0.29***	0.05	0.27***	0.05	0.24***	0.05	0.50***	0.04	0.37***	0.05
Cohabitation = 1	0.04	0.05	-0.06	0.06	-0.04	0.05	-0.11ł	0.06	0.03	0.06	-0.09	0.07
U.S. Citizen = 1	-0.10ł	0.05	0.02	0.05	-0.08	0.05	-0.07	0.06	-0.03	0.04	0.01	0.05
HS Diploma	0.00	0.06	-0.14*	0.06	0.06	0.05	-0.06	0.06	0.10	0.07	0.02	0.08
Some College	0.05	0.07	-0.10	0.07	0.03	0.06	-0.07	0.06	0.07	0.08	-0.07	0.09
Bachelor's Degree or higher	0.07	0.06	-0.02	0.06	-0.03	0.05	-0.12*	0.06	0.06	0.09	-0.09	0.10
Parent's Age	0.00	0.05	-0.09	0.06	0.12*	0.06	0.20**	0.06	-0.05	0.05	0.05	0.06
Employed = 1	0.00	0.05	-0.06	0.05	0.04	0.05	-0.02	0.05	0.06	0.04	0.00	0.05
Social Support = 1	-0.01	0.05	-0.03	0.05	0.02	0.05	0.03	0.06	0.03	0.05	-0.06	0.05
Higher order birth = 1	0.05	0.05	-0.03	0.05	-0.06	0.05	-0.05	0.06	0.01	0.05	0.02	0.05
Father has a child by another partner	-0.03	0.05	-0.01	0.06	0.00	0.05	-0.08	0.05	0.03	0.04	0.00	0.05
Mother has a child by another partner	0.00	0.05	0.06	0.06	0.00	0.06	0.07	0.06	-0.05	0.05	0.00	0.05
Both parents have a child by another partner	0.04	0.05	0.06	0.06	-0.08	0.05	0.00	0.06	-0.04	0.05	0.03	0.05
Father ever been incarcerated =1	0.02	0.05	0.02	0.05	-0.12*	0.05	-0.06	0.05	0.10*	0.05	-0.14**	0.05
Parent's Health	0.10*	0.05	0.05	0.06	-0.10ł	0.05	0.11*	0.05	0.01	0.05	-0.01	0.05
Parent's Impulsivity	-0.02	0.05	-0.08	0.05	-0.03	0.05	-0.02	0.05	-0.01	0.04	-0.03	0.05
Relationship Quality	0.03	0.05	0.07	0.05	0.20***	0.05	0.03	0.06	0.21***	0.05	0.11ł	0.06
Child's Health	0.11*	0.05	-0.03	0.05	0.05	0.05	0.08	0.05	-0.04	0.04	0.05	0.05
Child is male = 1	-0.03	0.05	-0.04	0.05	-0.02	0.05	0.10*	0.05	0.14**	0.04	0.00	0.05
Child Temperament	-0.01	0.05	0.01	0.05	-0.10*	0.05	-0.03	0.05	-0.01	0.04	0.03	0.05
Poverty = 1	0.01	0.05	-0.10ł	0.06	-0.08	0.05	0.04	0.05	0.03	0.05	0.11*	0.05

	Black Couples					Hispanic Couples					White Couples			
	Mothers		Fathers		N	Mothers		Fathers		Moth	Mothers		Fathers	
Variables	b	se	b	se	b	s	se	b	se	b	se	b	se	
Coparenting (Lagged)	0.46***	0.05	0.36***	0.05	0.33*	** 0.	.07	0.27***	0.06	0.54***	0.07	0.41***	0.11	
Cohabitation = 1	-0.04	0.06	-0.01	0.05	-0.0	4 0.	.05	0.06	0.05	-0.02	0.06	-0.05	0.06	
U.S. Citizen = 1	-0.03	0.03	-0.02	0.05	0.0	2 0.	.05	-0.08	0.07	0.00	0.04	-0.01	0.04	
HS Diploma	-0.03	0.06	0.09	0.06	-0.0	1 0.	.05	-0.07	0.05	0.00	0.08	0.07	0.08	
Some College	-0.10	0.06	0.04	0.06	0.0) 0.	.05	-0.04	0.06	0.01	0.10	-0.01	0.10	
Bachelor's Degree or higher	-0.09	0.07	-0.03	0.06	0.04	4 0.	.04	0.01	0.04	-0.01	0.12	-0.02	0.12	
Parent's Age	-0.02	0.06	-0.04	0.06	-0.1	lł 0.	.06	0.07	0.06	-0.02	0.06	-0.01	0.05	
Employed = 1	-0.07	0.05	-0.13*	0.05	0.0	0.	.05	0.15*	0.07	0.07ł	0.04	0.02	0.04	
Social Support = 1	-0.01	0.05	0.02	0.05	-0.15	i* 0.	.06	0.01	0.06	-0.02	0.05	-0.02	0.05	
Higher order birth = 1	0.08	0.05	0.09ł	0.05	-0.0	7 0.	.05	0.04	0.06	0.03	0.05	-0.02	0.05	
Father has a child by another partner	-0.01	0.05	0.08ł	0.05	0.0	0.	.05	-0.04	0.06	-0.09ł	0.05	-0.02	0.06	
Mother has a child by another partner	0.00	0.05	0.01	0.05	0.04	4 0.	.05	-0.02	0.05	0.02	0.04	-0.06	0.05	
Both parents have a child by another partner	-0.06	0.05	-0.04	0.06	0.0	3 0.	.06	-0.11	0.07	0.01	0.04	-0.08	0.06	
Father ever been incarcerated =1	-0.09	0.05	-0.06	0.05	-0.0	9 0.	.06	0.02	0.05	-0.08	0.06	0.02	0.06	
Parent's Health	-0.05	0.04	0.00	0.05	0.0) 0.	.06	0.01	0.05	-0.08*	0.04	-0.05	0.05	
Parent's Impulsivity	0.05	0.05	-0.05	0.04	-0.0	8 0.	.05	-0.09ł	0.05	0.01	0.05	0.02	0.06	
Relationship Quality	0.10*	0.05	0.16***	0.05	0.16	** 0.	.06	0.12*	0.06	0.05	0.06	0.21***	0.06	
Child's Health	0.02	0.04	0.08	0.05	0.0	2 0.	.05	0.07	0.06	-0.01	0.04	0.02	0.04	
Child is male = 1	0.02	0.04	-0.06	0.04	0.0	0.	.05	0.03	0.05	0.04	0.04	-0.01	0.04	
Child Temperament	-0.06	0.05	0.00	0.05	0.0	2 0.	.05	0.01	0.05	0.02	0.04	-0.04	0.04	
Poverty = 1	0.04	0.04	-0.09	0.05	0.02	2 0.	.05	0.06	0.05	-0.03	0.06	0.09*	0.04	

Table 3.13: Standardize Estimates for Control Variables on Coparenting by Race/Ethnicity (Year-5)

	Black Couples					Hispani	c Couples		White Couples				
	Mothers		Fathers		Moth	Mothers		Fathers		Mothers		Fathers	
Variables	b	se	b	se	b	se	b	se	b	se	b	se	
Dep. Symptoms (Lagged)	0.31***	0.07	0.56***	0.12	0.42***	0.09	0.20ł	0.11	0.53***	0.08	0.50***	0.11	
Cohabitation = 1	-0.34	0.30	-0.23	0.34	0.00	0.47	-0.33	0.36	0.53	0.45	0.62	0.52	
U.S. Citizen = 1	1.28	0.82	1.84ł	0.99	0.47	0.38	0.20	0.40	0.22	0.64	-0.45	0.52	
HS Diploma	0.00	0.38	0.80*	0.35	-0.08	0.40	0.40	0.39	-0.70	0.84	0.76	0.50	
Some College	-0.08	0.42	0.39	0.41	-0.43	0.55	1.12*	0.46	0.34	0.79	-0.05	0.50	
Bachelor's Degree or higher	-0.72	0.68	-1.25	0.93	2.00**	0.73	0.98	0.73	-0.70	0.90	-0.27	0.59	
Parent's Age	-0.03	0.03	-0.03	0.02	0.02	0.03	0.00	0.03	0.07ł	0.04	-0.05	0.03	
Employed = 1	-0.19	0.26	-0.21	0.32	-1.12**	0.35	-0.43	0.55	-0.60*	0.29	-1.67*	0.69	
Social Support = 1	0.28	0.24	-0.04	0.26	0.09	0.35	0.74ł	0.39	0.12	0.30	0.84*	0.35	
Higher order birth = 1	0.23	0.26	0.34	0.28	-0.83*	0.36	0.64ł	0.33	0.39	0.36	1.823***	0.36	
Father has a child by another partner	0.61ł	0.36	0.12	0.44	1.26ł	0.40	0.44	0.54	0.78	0.55	-0.75	0.50	
Mother has a child by another partner	0.26	0.36	-0.53	0.38	1.30**	0.52	0.68	0.47	-0.07	0.48	0.74	0.50	
Both parents have a child by another partner	0.96*	0.38	-0.35	0.42	0.95ł	0.54	-0.48	0.66	0.15	0.75	-0.12	0.60	
Father ever been incarcerated =1	-0.89**	0.31	0.17	0.34	0.44	0.40	0.90ł	0.46	-0.77	0.95	0.42	0.44	
Parent's Health	-0.252ł	0.13	0.26	0.15	-0.26	0.17	-0.12	0.19	0.01	0.21	-0.56**	0.18	
Parent's Impulsivity	0.60***	0.17	0.76***	0.21	0.49*	0.25	0.50*	0.25	0.70*	0.24	0.40	0.30	
Relationship Quality	0.02	0.13	0.18	0.13	0.04	0.17	0.47*	0.22	0.03	0.24	0.37	0.28	
Child's Health	-0.28ł	0.16	0.53**	0.19	-0.32*	0.15	0.19	0.20	-0.54*	0.25	0.17	0.35	
Child Temperament	-0.05	0.18	0.22	0.18	-0.16	0.20	-0.04	0.23	-0.20	0.22	-0.88**	0.27	
Poverty = 1	-0.12	0.28	0.00	0.31	-0.27	0.34	-0.25	0.34	0.99	0.78	-2.85***	0.69	

Table 3.14: Negative Binomial Coefficients for Control Variables on Depressive Symptoms by Race/Ethnicity (Year-3)

	Black Couples				Hispa	nic Couples		_	White Couples				
	Mothers		Fathers		M	Mothers		Fathers		Mothers		Fathers	
Variables	b	se	b	se	b	se	b	se	b	se	b	se	
Dep. Symptoms (Lagged)	0.24*	0.10	0.19	0.12	0.23	0.15	0.21*	0.10	0.36	0.20	0.28	0.18	
Cohabitation = 1	0.08	0.19	0.13	0.17	-0.03	0.18	0.08	0.13	0.41	0.56	-0.10	0.26	
U.S. Citizen = 1	-0.21	0.24	0.08	0.12	-0.17	0.17	0.27	0.18	0.31	0.28	-0.11	0.23	
HS Diploma	0.11	0.24	-0.02	0.18	0.20	0.20	-0.15	0.12	-0.16	0.30	0.11	0.25	
Some College	0.33	0.24	-0.10	0.20	0.08	0.19	0.10	0.22	-0.01	0.30	0.17	0.25	
Bachelor's Degree or higher	0.29	0.37	-0.09	0.20	1.04	0.69	-0.07	0.22	-0.11	0.26	0.16	0.35	
Parent's Age	-0.02	0.08	0.02	0.03	-0.05	0.19	0.02	0.01	0.05	0.08	0.02	0.01	
Employed $= 1$	-0.20	0.22	-0.14	0.25	-0.01	0.20	0.24	0.17	-0.07	0.14	-0.63†	0.38	
Social Support = 1	0.25	0.19	0.42	0.52	0.58	0.30	0.10	0.14	0.23	0.22	-0.01	0.11	
Higher order birth $= 1$	0.09	0.32	0.11	0.16	0.08	0.17	0.07	0.15	-0.12	0.16	-0.11	0.10	
Father has a child by another partner	0.01	0.21	-0.13	0.18	0.35	0.26	-0.21	0.21	0.23	0.44	0.47	0.29	
Mother has a child by another partner	0.02	0.19	-0.14	0.16	-0.10	0.48	-0.08	0.13	-0.28	0.29	-0.09	0.17	
Both parents have a child by another partner	0.01	0.25	0.29	0.24	0.68	0.49	-0.19	0.23	-0.24	0.44	-0.02	0.31	
Father ever been incarcerated =1	0.68†	0.40	0.20	0.22	-0.27	0.24	-0.19	0.15	-0.14	0.31	0.14	0.30	
Parent's Health	-0.12	0.10	-0.09	0.15	0.11	0.09	-0.03	0.05	0.02	0.10	-0.18*	0.09	
Parent's Impulsivity	0.21	0.13	-0.02	0.13	0.17	0.20	0.21†	0.12	-0.05	0.20	-0.01	0.12	
Relationship Quality	0.01	0.10	-0.05	0.09	0.01	0.10	-0.11	0.08	-0.03	0.18	0.01	0.69	
Child's Health	-0.08	0.20	-0.09	0.10	-0.09	0.10	0.07	0.09	-0.40	¢ 0.17	-0.10	0.12	
Child Temperament	0.04	0.13	0.03	0.09	0.10	0.10	-0.01	0.11	-0.05	0.10	-0.08	0.14	
Poverty = 1	-0.08	0.18	-0.11	0.17	0.28	0.21	0.05	0.11	0.43	0.50	-0.27	0.27	

Table 3.15: Negative Binomial Coefficients for Control Variables on Depressive Symptoms by Race/Ethnicity (Year-5)

CHAPTER IV

Parents' Parenting Stress and Relationship Quality Over Time: The Role of Poverty

INTRODUCTION

The birth of a child is a joyous and exciting occasion for many parents, and having a child contributes to parents' overall life satisfaction and happiness (Nelson et al. 2013). Yet, extant research suggests that the demands of parenting young children result in two general outcomes: (1) an increase in parental stress or distress (Crnic, Low, and Bornstein 2002; Evenson and Simon 2005; Umberson, Pudrovska, and Reczek 2010) and (2) a decrease in relationship quality between intimate partners (Lavee 2013; Randall and Bodenmann 2009; Umberson and Reczek 2007). The general argument is that the day-today hassles of parenting result in elevated levels of stress and strain in the parenting role, and as parents adjust to the needs of their child, it is often at the expense of the motherfather relationship leading to declines in relationship quality and satisfaction. Both parenting stress and relationship quality are critically important given the impact these factors have on broader family processes and children's well-being (Benzies, Harrison, and Magill-Evans 2004; Crnic, Gaze, and Hoffman 2005; Cui, Donnellan, and Conger 2007). Although prior studies have examined the parenting stress and relationship quality nexus (Lavee, Sharlin, and Katz 1996; Quittner, Glueckauf, and Jackson 1990), these studies have been limited to small, homogenous samples. Moreover, prior research has either examined the link between stress in the parenting role and its effect on the quality of intimate relationships, or how supportive partners help to reduce parenting stress. As a consequence, the direction of the association remains unclear. Thus, the purpose of the

current study is to examine parenting stress and couple's relationship quality as a longitudinal and reciprocal process among couples five years after the birth of a child.

The longitudinal and reciprocal association between parenting stress and couple's relationship quality may change as children develop over time. Developmental and lifecourse perspectives highlights that the birth of a child tend to set the stage for complex family dynamics whereby parental stress, intimate relationships, and children change in tandem (e.g., Umberson et al. 2010). Specifically, prior research suggests that, when children are young, parenthood tends to be more stressful and often create strain between partners because early childhood is more demanding; however, as children grow older, both stress in parenting and relationship strain gradually dissipates (Nomaguchi and Milkie 2003; Umberson, Williams, Powers, Chen, et al. 2005).

Parenting stress refers to the extent to which parents perceive that their parental demands exceed familial and social resources to meet those demands (Abidin 1992). As such, economic disadvantage makes the parenting role especially challenging by further exacerbating stressful conditions. Factors such as poverty and economic hardship reflect the lack of financial and material resources that hinder families from accessing basic needs (e.g. food, housing, health care, etc.). As a result, many parents become frustrated and overwhelmed leading to negative parenting behaviors (McLeod and Shanahan 1993; McLoyd 1995) and conflict between partners (Hardie and Lucas 2010). Prior research has highlighted the ways in which poverty affects both parenting stress and relationship quality (Cunradi et al. 2000; Middlemiss 2003; Raikes and Thompson 2005). Many of these studies, however, have examined families in poverty using cross-sectional data. Consequently, prior research inherently conceals poverty fluctuations or histories (i.e.,

poverty entries and exits) that many families experience over time (e.g., Edin and Kissane 2010). Understanding the association between parenting stress and couple's relationship quality longitudinally in the context of families' continual and temporal poverty experiences as children develop paints a more holistic picture of stress in the parenting role, couples, and economic inequality. Thus, the present study, using data from the Fragile Families and Child Well-being Study (FFCW), addresses the following questions: (1) is there a longitudinal and reciprocal association between parenting stress and relationship quality; (2) does the association change over time as children age from infant to toddler, and from toddler to preschooler; and (3) does the association vary across families' poverty histories?

LITERATURE REVIEW

Theoretical Perspective

The current study draws on two theoretical perspectives and additional empirical studies that highlight the association between parenting stress and couple's relationship quality. First, I rely on the stress spillover hypothesis (Bolger et al. 1989; Larson and Almeida 1999; Neff and Karney 2007) to understand the how stress in one familial domain (e.g., parenting) affects another domain (e.g., mother-father relationship quality). Stress spillover contends that family roles can generate stress, subsequently creating a contagion of stressful experiences that affects the well-being of couple dyads. Second, I draw on parenting stress research (Abidin 1992; Crnic et al. 2002) which suggests that stress in the parenting role is associated with parent, child, and situational factors. Similar to prior research, the present study focuses on supportive relationship quality between intimate partners as a situational factor (e.g., Bronte-Tinkew, Horowitz, and Carrano 2010). Being in a supportive relationship refers to the extent to which individuals in intimate

partnerships receive emotional sustenance (i.e., love, care, support, etc. Cutrona 1996), which works as an important resource that buffers stress (e.g., Lavee 2013). Taken together, the theoretical perspectives highlight the interdependent nature of families (Cox and Paley 1997; O'Brien 2005), the importance of stress in intimate relationships (Umberson and Reczek 2007), and the determinants and consequences of parenting stress (Abidin 1992). Figure 4.1 presents the conceptual model and is used for the review of research (outlined below) and orientates the hypothesized association between constructs guiding the empirical analyses.

<INSERT FIGURE 4.1 ABOUT HERE>

Parenting Stress and the Quality of Mother-Father Relationships

Stress, parenthood, and the quality of intimate relationships has been a concern for many scholars (Lavee 2013; Randall and Bodenmann 2009; Umberson and Reczek 2007; Umberson et al. 2005). Although stress is conceptualized and operationalized in different ways across studies, the general conclusion is that stress negatively impacts individuals and families. Stress in the parenting role, which is distinct from work stress and marital stress, has important implications for parent, child, and family functioning (e.g., Belsky 1984). Parenting stress is the result of the combination of individual, couple, child, and contextual factors that directly affects the parenting role (Abidin 1992). In some instances, parental stress may interfere with the quality of the intimate partnership between parents. As parents adjust to and deal with the demands and challenges of parenthood, a significant amount of emotional and physical energy is dispensed (Deater-Deckard 2008). Subsequently, intimate partners begin to show less intimacy, become less supportive, and spend less time together (Cowan and Cowan 2000; Kirby, Baucom, and Peterman 2005), which, in turn, affects the quality and stability of the intimate

partnership (e.g., Bradbury, Fincham, and Beach 2000) which is represented in Figure 4.1, Path A.

Prior research shows that stress in the parenting role is associated with a decrease in relationship quality between partners. For example, in a sample of 287 married Israeli couples, Lavee, Sharlin, and Katz (1996) found that mother's and father's parenting stress was associated with lower levels of marital quality. In a study of resident fathers, Bronte-Tinkew, Horowitz, and Carrano (2010) revealed that paternal parenting stress was associated with less supportive coparenting between mothers and fathers. Quittner, Glueckauf, and Jackson (1990) found that parenting stress lowered mothers' perceptions of emotional support. Additional research on the transition to parenthood and parents with children with disabilities and health problems (mental and physical) gives credence to the negative association between parenting stress and couple's relationship quality (e.g., Gupta 2007; Schulz, Cowan, and Cowan 2006).

The Quality of Mother-Father Relationships and Parenting Stress

The quality of intimate relationships have important implications for family functioning that affect both individual and family well-being (Bradbury, Fincham, and Beach 2000; Fincham and Beach 2010a). Relationship quality between partners can be positive (e.g., supportiveness) or negative (e.g., conflict; Fincham and Beach 2010b). The current study examines a positive dimension of relationship quality. In many ways, positive relationship quality is similar to the notion of partner support. Couples in a supportive relationship offer each other important emotional sustenance and coping resources (Lavee 2013) that can be extremely helpful to reduce the effects of stressful situations as indicated in Figure 4.1, Path B. For many intimate partners, the feeling of support,

respect, and love (Cutrona 1996; Don and Mickelson 2012) leads to better psychological well-being and parental adjustment (Thoits 2011; Umberson and Montez 2010). Supportive relationships may be especially true for couples after the birth of a child given the stress associated with having young children (e.g., Umberson et al. 2010).

Although prior studies have not explicitly examined the association between couple's relationship quality and parenting stress per se, research shows that partner's social support lowers parenting stress. For example, in a study of mothers and infants, Mulsow et al. (2002) found that intimacy between partners (e.g., my spouse/partner can really understand my hurts and joys, etc.) and general social support reduced parenting stress. Also, in a sample of Swedish mothers, Ostberg and Hagekull (2000) discovered that parents with higher levels of support were associated with less stress in parenting. In contrast, Raikes and Thompson (2005), in a sample of 65 low-income mothers, revealed no significant association between social support and parenting stress. Nevertheless, many prior studies support the notion that a positive, supportive partner serves as a buffer against stress in parenting (e.g., Wadsworth 2012). Studies also show that parents in supportive relationships tend to display positive parenting behaviors (Green, Furrer, and McAllister 2007) and feel positive about parenting (Suarez and Baker 1997). Similarly, low levels of support are associated with higher levels of parenting stress (Gelfand, Teti, and Radin Fox 1992; Ostberg and Hagekull 2000). Based on the reviewed research above, I expect a mutual influence between parenting stress and couple's relationship quality over time. Specifically, I present the following hypotheses:

H1: Early parenting stress levels will be associated with lower levels of couple's relationship quality for both mothers and fathers over time.

H2: Early couple's relationship quality levels will be associated with higher levels of parenting stress both mothers and fathers over time.

Changes in the Associations as Children Develop Over Time

Although prior studies have examined changes in both parenting stress and relationship quality over time (Putnick et al. 2010; Umberson et al. 2010; Umberson, Williams, Powers, Liu, et al. 2005), these studies have been investigated in isolation of each other. Thus, we know less about the potential longitudinal and reciprocal effects between parenting stress and relationship quality as children develop. One may expect that during the early stages of development—from infant to toddler—may be the most stressful and straining for parents (in comparison with toddler to preschooler). Specifically, during the early years, children need more attention (feeding, changing diapers, etc.), and thus require more parental time and energy (e.g., Nomaguchi and Milkie 2003) which leads to elevated stress and strain. Thus, parental stress may be more influential in lowering couple's relationship quality during the infant to toddler years compared to the toddler to preschool year. Likewise, being in a positive supportive relationship may matter more in reducing stress during the early developmental stages compared to the later stages. More explicitly, I present the following hypothesis:

H3: The longitudinal and reciprocal effects between parenting stress and couple's relationship quality will be stronger for both parents from Year-1 to Year-3 than from Year-3 to Year-5

Living in (and out) of Poverty

Poverty has several deleterious consequences for families and children (Conger, Conger, and Martin 2010; Edin and Kissane 2010). The extent to which families experience poverty, however, is complex. Specifically, with the availability of longitudinal data, the empirical evidence suggests that poverty is a dynamic process whereby families undergo stability and change in poverty experiences (Cellini, McKernan, and Ratcliffe 2008; Edin and Kissane 2010). For example, results from the Survey of Income and Program Participation (SIPP) revealed that from 2009 to 2011, only 3.5 percent of people lived in poverty; of people who were poor in 2009, 5.4 percent remained poor in 2011; and, of the people who were in poverty in 2009, 35.4 percent were not in poverty in 2011 (Edwards 2014). Indeed, these statistics confirm the continuity and change in the experiences of poverty histories for many families.

Although prior studies have examined how poverty and other indices of economic disadvantage affect stress in the parenting role and relationship quality (e.g., Hardie and Lucas 2010; Raikes and Thompson 2005), many studies have been limited by cross-sectional data. This is surprising given the deleterious effects that both persistent poverty and changes in poverty have on families (Corcoran and Chaudry 1997; Edin and Kissane 2010). Thus, the present study examines the longitudinal and reciprocal effects of parenting stress and relationship quality across three poverty groups: (a) no poverty, (b) transient poverty, and (c) persistent poverty. The extent to which poverty affect these familial processes are described below.

The potential moderating effect of families' poverty histories on the association between stress in the parenting role and relationship quality may work differently (or similarly) across families as shown in Figure 4.1, Paths C and D. On the one hand, parenting stress may be salient for all parents, regardless of socioeconomic status. Given that parenthood is stressful even under the most optimal conditions (Rodriguez-JenKins and Marcenko 2014), the quality of intimate partnerships may be vulnerable to parental stress for all couples, thus leading to no differences between families across poverty groups. Prior research using samples of middle-class parents (e.g., Crnic et al. 2005; Deater-Deckard and Scarr 1996) and low-income parents (e.g., Middlemiss 2003; Raikes and Thompson 2005) shows that parenting stress leads to unfavorable outcomes for both family types. Similarly, being in a supportive relationship may help to reduce stress in the parenting role for all families. Indeed, supportive partners serve as a valuable resource to aid individuals to cope with different forms of stress and distress (Lavee 2013; Turner, Turner, and Hale 2014). As a result, regardless of families' poverty histories, all couples in a positive, healthy relationship may reap the benefits of supportive resources that intimate partners provide.

On the other hand, poverty may further exacerbate stress in the parenting role as families lack the resources to deal with stressful events associated with parenting (Wadsworth 2012). As a result, families living in poverty may experience an increases in stress and strain that often manifest itself through less than optimal parenting practices and generate conflict between partners (Hardie and Lucas 2010; Kotchick, Dorsey, and Heller 2005). Simply put, poverty tends to amplify other stressors, making bad situations worse. In regards to couples' relationship quality, having a positive, supportive partners may be more helpful for families in poverty than families experiencing no poverty. More specifically, strong partner support in the context of economic disadvantage may help to generate higher levels of self-efficacy (Green et al. 2007) leading to more positive parenting (Coleman and Karraker 1998). Some studies show, however, that supportive relationships may not be as effective for families in poverty (Raikes and Thompson 2005) as many members of the same support network may also be experiencing similar stressful events (e.g., Ceballo and McLoyd 2002) rendering social support less effective. That is,

although supportive relationships may serve as an important resource for some families, it is no panacea (e.g., Wadsworth 2012).

Taken together, the research presented above paints a complicated picture on the ways poverty may moderate the association between parenting stress and couples' relationship quality. Given that poverty generates its own stress within families (Wadsworth 2012), the longitudinal and reciprocal association between parenting stress and relationship quality may be more detrimental for families who live in transient or persistent poverty compared to families who experience no poverty. Alternately, stress in the parenting role may be challenging for all parents, regardless of socioeconomic status, leading to lower levels of couple's relationship quality across families' poverty histories. Moreover, being in a positive, support relationship may be a valuable resource for all families. Thus, the longitudinal and reciprocal association between parenting stress and relationship quality may yield no differences between families across poverty histories. Given the equally important ways poverty may (or may not) affect parental stress and couples' relationship quality, no specific hypothesis is offered on the differences (or similarities) across poverty groups.

Additional Factors

The current study employs several control variables that are associated with both parenting stress and couple's relationship quality. Indeed, scholars have noted that parents, child, and contextual factors contribute to parenting stress (Deater-Deckard 2008) and relationship quality (Bradbury and Karney 2004). Thus, the current study adjusts for such factors. For instance, parenting stress may vary between union types (e.g., married versus cohabiting parents; Cooper et al. 2009) and studies show that married couples display higher levels of relationship quality (Brown and Booth 1996) compared to cohabiting parents. Race and ethnic minorities report higher levels of parenting stress (Nomaguchi and House 2013) and lower levels of relationship quality (Bulanda and Brown 2007) compared to white couples. Age is associated with stress in the parenting role (Ostberg and Hagekull 2000) and relationship quality over time (Umberson, Williams, Powers, Chen, et al. 2005). Education (Cooper et al. 2009), employment status (Gyamfi, Brooks-Gunn, and Jackson 2001), parents' mental and physical health (Crnic and Acevedo 1995) are associated with parenting stress and relationship quality. Social support (from family and friends) has been linked to lower levels of parenting stress (Lavee 2013) and associated with an increase in relationship quality (Cutrona 1996). The number of children affects both parenting stress and relationship quality (Lavee et al. 1996), birth order is associated with parenting stress (Krieg 2007), multi-partnered fertility, incarceration is associated with stress in the parenting role (Loper et al. 2009) and strains couple's relationship quality (Lopoo and Western 2005). In addition, child characteristics such as child's sex, temperament, and physical health is associated with parenting stress (Crnic et al. 2002; McBride, Schoppe, and Rane 2002). These factors are adjusted for in the analyses (see Tables 4.9 through 4.16).

METHODS

Data

Data for this study are from the Fragile Families and Child Well-being Study (FFCW). The FFCW is a nationally representative, longitudinal study that follows an urban birth cohort of 4,898 children and their parents (3,712 unmarried and 1,186 married births) in 20 U.S. cities with populations of 200,000 or more. The study is based on a stratified, multistage probability sample with an oversample of unmarried births in urban cities. The study began 1998-2000 and contains 4,898 mothers and 3,830 fathers. At baseline, mothers were interviewed in person while in the hospital within 48 hours of the birth, and fathers were interviewed in person or by phone once he was located (for more detailed information, see Reichman et al. 2001). Parents were re-interviewed when the child was one, three, and five years of age. The response rate for eligible mothers and fathers at baseline was 86% and 78%, respectively. Subsequent 1-, 3-, and 5-year follow-ups yielded 90%, 88%, and 87% response rates for eligible mothers, and 74%, 72%, and 70% for eligible fathers (Bendheim-Thomas Center on Child Wellbeing 2008).

The sample includes couples (biological mothers and fathers of the focal child) who were living together (either married or cohabiting) at the baseline survey and remained in the subsequent surveys and has no missing values on the focal variables. All survey waves are used; however, the main analyses are based on the 1-, 3-, and 5-year follow-up surveys because the focal variables (i.e., parenting stress and relationship quality) were measured at the 1-, 3-, and 5-year surveys only. The selection criteria of the sample for the data analyses are parents who were either married or cohabiting from the 1-year to the 3-year surveys, and from the 3-year to the 5-year surveys. This is done in order to maintain couples who were romantically involved consistently over all survey years.

As a result of the selection criteria, of the couples living together at the 1-year follow-up (N = 2, 341), 347 cases were dropped (15%) from the sample because either the mother or father was not interviewed at the 3-year survey, 317 cases (14%) were dropped because the parents ended their relationship between the 1- and 3-year surveys,

and 397 cases (17%) were dropped because information on parenting stress and relationship quality were missing. This resulted in a sample of 1,280 (55%). For couples who were living together at the 3-year follow-up (N=2,032), 324 cases (16%) were dropped because either mother or father were not interviewed at the 5-year follow-up, 288 cases (14%) because the relationship ended between the 3- and 5-year waves, and 141 cases (7%) because of missing data on parenting stress and relationship quality. This resulted in a sample of 1,279 (63%) couples who were romantically involved and consistently living together over 3- and 5-year follow-up surveys. Taken together, the final sample resulted in 942 couples (mothers and fathers) across the 1-, 3-, and 5-year follow-up surveys. Given that the random and fixed effects models (see Analytic Strategy below) pooled cases across survey years, parents were observed at multiple time points which results in 2,826 person-year observations.

In analyses of attrition, the excluded cases were more likely to be racial/ethnic minorities, somewhat younger, and had lower levels of education compared to parents who remain in the sample (see Tables 4.5 through 4.8). Comments about potential implications of attrition are in the Discussion section. To maximize sample size, multiple imputation technique was employed to impute missing data on the covariates, but not the focal variables for which Full Information Maximum Likelihood (FIML) was used (Acock 2005; Allison 2002; Enders and Bandalos 2001). Ten multiple imputation data sets were constructed using imputation by chained equations in Stata and then the analyses were conducted and combined using Rubin's rules (Rubin and Little 2002) in Mplus. City sampling weights are used for the descriptive statistics to adjust for oversampling of nonmarital births but not for the analyses because the study controls for key characteristics associated with the weights (e.g., marital status at the birth of the child, age, race, and education; see Winship and Radbill 1994).

Measures

Parenting Stress

Parenting stress was measured at the 1-, 3-, and 5-year follow-up surveys using four items adapted from the Parenting Stress Index developed by Abidin (1983). The FFCW study used a short form to identify stress in parenting. Mothers (α = .56, .59, and .64) and fathers (α = .58, .61, and .61) were asked about their agreement (1 = *strongly disagree* to 4 = *strongly agree*) to four questions about parenting at Year-1, 3, and 5, respectively: (1) "Being a parent is harder than I thought it would be," (2) "I feel trapped by my responsibilities as a parent," (3) "I find that taking care of my child(ren) is much work than pleasure," and (4) I often feel tired, worn out, or exhausted from raising a family." Higher scores reflect higher levels of parenting stress. Despite the modest alpha scores, these items have been employed elsewhere (e.g., Bronte-Tinkew et al. 2010; Cooper et al. 2009).

Relationship Quality

Relationship quality was measured at the 1-, 3-, and 5-year follow-up surveys using five items. Mothers and fathers were asked identical questions about the frequency of their partner in the following supportive ways: (1) "is fair and willing to compromise when you have a disagreement," (2) "expresses affection or love for you," (3) "encourages or helps you to do things that are important to you," (4) "listens to you when you need someone to talk to," and (5) "really understands your hurts and joys." Response choices range from (1) never to (3) often. The identical items for mothers and fathers were combined to create a dyadic measure ($\alpha = .76, .77, and .80$) for Year-1, 3, and 5,

respectively. Further, the combined items were averaged with higher scores reflect higher levels of supportive relationship quality. This measure is similar to a prior study using the FFCW study (e.g., Carlson et al. 2011).

Poverty Histories

Poverty histories were gauged using mothers' reports at the baseline, 1-, 3-, and 5-year follow-up surveys using the household income-to-needs ratio based on official U.S. poverty thresholds from the Census Bureau which is adjusted by family size, family composition, and year. A ratio of 1 or less indicated that the family lived in poverty. Thus, the poverty measure was dichotomized to indicate (0) not in poverty and (1) in poverty. The items were summed across the four survey years (i.e., baseline, 1-, 3-, and 5-year follow-up years) to create an index ranging from 0 (not in poverty) to 4 (live in poverty across the four survey years). Further, the item was collapsed into three categories to gauge poverty histories: no poverty, transient poverty (experience poverty once or twice), and persistent poverty (experienced poverty three or four times).

Control Variables

Control variables were included in all statistical models (described above) that are expected to be associated with parenting stress and couple's relationship quality. Identical measures were used for both mothers and fathers at the baseline survey (unless otherwise specified). *Marital status* was measured by a dichotomous variable with (0) indicating married (reference group) and (1) indicating cohabitation. Race/ethnicity was gauged (using mother's self-report) by a series of dummy variables reflecting non-Hispanic white (reference), non-Hispanic Black, Hispanic, and non-Hispanic other. A dichotomized measure of mixed race couples was also included. Mothers and fathers' *age* was measured (in years) as continuous variables. Mothers and fathers' *education* level was

measured using four categories: (1) less than high school (reference), (2) high school or equivalent, (3) some college or tech training, and (4) college graduate or more. *Physical health* (measured at the 1-year follow-up for parents and child) was measured by asking parents the following question: "In general, how is your health?" Mothers reported child's health with responses ranging from (1) poor to (5) excellent. Both parents' *mental health* was gauged using the Composite International Diagnostic Interview Short Form (CIDI-SF) to indicate feelings of dysphoria or anhedonia—a dummy variable was employed to represent having met the criteria. *Employment status* was measured with a dichotomous item indicating whether each parent "Did any regular work for pay last week?" Response were (0) no and (1) yes.

Social support was measured with a dichotomous question (0=no, 1=yes) asking both parents "since child was born, have you received any financial help or money from anyone other than [partner]?" Mothers reported on fathers' *incarceration history* (at the 1-year follow-up) indicating whether fathers have ever been in jail or prison (0=no, 1=yes). Mothers reported the *number of children in the household* at the 1-year followup. Parents' fertility history was gauged with two separated measures: First, a measured was created to indicate whether the focal child is a *higher order birth or first birth* (0=first birth, 1=higher order birth). Second, a measured was created to indicate a series of dummy variables indicating *multipartnered fertility* (at 1-year follow-up) which reflects whether mothers and fathers reported having a child with another partner: neither parent has a child by another partner (reference), father has child by another partner only, mother has child by another partner only, and both parents has a child by another partner. The study also takes into account additional child characteristics such as child's *sex* (boy=1) and *temperament*—which is measured by 6-items that gauges the difficulty of a child's temperament with higher scores indicating higher levels of difficult temperament.

Analytic Strategy

The current study uses several analytic techniques to address the longitudinal and dyadic association between parenting stress and relationship quality. Each technique takes advantage of the longitudinal design of the data with repeated measures spanning over three years. First, random effects models are employed to examine the association between parenting stress and relationship quality, and vice versa. Random effects model assumes that the variation across individuals is random and uncorrelated with the independent variables included the model, capturing variation between and within respondents. Second, fixed effects models are also used to test the research hypothesis in the current study. Fixed effects models, unlike random effects, takes into account timeinvariant factors (e.g., race/ethnicity), which only account for the variation within respondents. Although supplementary analysis (i.e., Hausman test) suggests that the model favors fixed effects over random effects, the parameter estimates are shown to allow the reader to see the extent to which the magnitude of the association changes between models (for a detailed explanation of fixed and random effects models, see Greene 2010). Third, cross-lagged path models are used to examine the directionality of parenting stress and relationship quality. More specifically, is early parenting stress associated with later relationship quality, or vice versa, or both? Last, the aforementioned models were re-estimated across the different family experiences of poverty (no poverty, transient poverty, persistent poverty). For each analytic technique, the parameter estimates, for the focal variables (i.e., parenting stress and couple's relationship quality)

are standardized which indicates that a change in the independent variable is associated with a standardized change in the dependent variable.

RESULTS

Descriptive Statistics

Table 4.1 shows the descriptive statistics for the variables in the study for all couples and by poverty histories (the statistics are weighted by city sampling weights). A substantial amount of cohabiting couples experienced both transient and persistent poverty. Moreover, couples living in transient or persistent poverty were more likely to be from a race/ethnic minority group, have lower levels of education, tend to be younger, and reported lower levels of physical health compared to couples who did not experience poverty over the four survey years. Mothers living in persistent poverty were more likely to be unemployed (83%) compared to mothers who experienced no poverty and transient poverty. Fathers who experienced transient and persistent poverty were more likely to ever been incarcerated (15% and 11%, respectively) compared to fathers who experienced no poverty.

Table 4.2 provides the means for parenting stress and couple's relationship quality with indications of significant differences by poverty histories (weighted ANOVA)². The results show that mothers reported relatively low levels of parenting stress and no significant differences emerged between mothers across poverty histories. Fathers also reported lower levels of parenting stress but some differences emerged between fathers across poverty histories. For example, fathers who experienced no poverty across survey years reported significantly higher levels of parenting stress (mean score of 2.08) than

² The F-Test results vary across variables. For mothers, the F-Test results for parenting stress was nonsignificant across the survey years. The F-Test results for paternal parenting stress and couple's relationship quality were significant (p < .05).

fathers who experienced transient poverty (2.23) but higher levels than fathers who experienced persistent poverty (1.90) at the Year-1 survey. No significant differences emerged during the Year-3 survey. For Year-5, fathers who lived in no poverty (2.02) had a significantly higher level of parenting stress that fathers who lived in persistent poverty (1.86). In regards to couple's relationship quality, couples who experienced no poverty across survey years, on average, reported higher levels of relationship quality compared to couples who lived in both transient and persistent poverty across survey years.

Multivariate Models

To address the first research question—is there an association between parenting stress and relationship quality, the analyses were executed using both random and fixed effects models (see Table 4.3; Models 1 and 2, respectively). First, the association between parenting stress and couple's relationship quality was examined. Model 1 presents the results for the random effects models for mothers. The findings show that a standard deviation change in parenting stress is associated with a .12 (p < .001) standard deviation decrease in couple's relationship quality. Similar to the random effects results, the fixed effects model (Model 2) shows that mother's parenting stress is significantly association with couple's relationship quality: a standard deviation change in mother's parenting stress is associated with a .06 standard deviation decrease in couple's relationship quality. For fathers (Model 1), parenting stress was also associated with a .11 standard deviation decrease in couple's relationship quality. The fixed effects also yielded significant result. That is, a standard deviation increase in parenting stress is associated with a .09 standard deviation decrease in couple's relationship quality. The effects, however, were modest. The results demonstrate that, across two analytic techniques, both mother's and father's parenting stress is associated with lower levels of couple's relationship quality.

Also, the converse association was examined—is couple's relationship quality associated with lower levels of parenting stress for both mothers and fathers? To address this question, random and fixed effect models were also employed. Model 1 shows the random effects results for the association between couple's relationship and parenting stress for mothers. The findings reveal that a standard deviation change in couple's relationship quality is associated with a standard deviation decrease in parenting stress (b = .11; p < .001). Again, the fixed effects model show similar results. Specifically, couple's relationship quality is associated with a .05 standard deviation decrease in maternal parenting stress (p < .05). For fathers (Model 1), a standard deviation change in couple's relationship quality is associated with a .10 standard deviation decrease in parenting stress. Moreover, the fixed effects results (Model 2) were similar to the fixed effects model which shows that a standard deviation change in couple's relationship quality is associated with a .08 standard deviation decrease in parenting stress. Similar to the results presented above, the effects are also modest. Taken together, these results support the notion that there is an association between parenting stress and couple's relationship quality.

Although the random and fixed effects models (discussed above) demonstrate an association between both parents' parental stress and couple's relationship quality, and vice versa, Model 3 displays the results for the cross-lagged path models in order to test for reciprocity—the second research question. For mothers (Panel A), there was no significant association between early parenting stress to later couple's relationship quality

from Years 1 to 3 or Years 3 to 5. Reciprocally, the association between couple's relationship and maternal stress yield significant results. Specifically, a standard deviation increase in couple's relationship quality at Years 1 and 3 is associated with a standard deviation decrease in maternal stress at Years 3 and 5, respectively (b = -.07 and -.06; p < .05). This suggests that when couples display supportive relationship quality, maternal parenting stress is significantly lowered in the subsequent survey years. These results give additional support that couple's relationship quality is associated with lower levels of parenting stress. For fathers, the longitudinal and reciprocal association between parenting stress and relationship quality was not statistically significant. Thus, hypothesis *H1* was not support. Moreover, given that early couple's relationship quality was associated with lower levels of parenting stress at the later years for mothers only, the hypothesis *H2* was partially supported.

In regards to change over time—the third research question, the analyses were executed to examine whether the association between parenting stress and couple's relationship was significantly different between Year-1 to Year-3 and Year-3 to Year-5. In order to test for differences, two models were compared: one model with the parameters estimated across survey years are allowed to estimated freely, and another model with the parameters constrained to be equal—that is, Years 1-3 equal to Years3-5. The models were compared using the χ^2 difference test. The results indicated that the differences between Years 1-3 and Years3-5 were not statistically significant for both mothers (p = .80) and fathers (p = .88). The findings suggest that there is no added or diminishing effect between parenting stress and relationship quality for either parent over time. Thus, the research hypothesis *H3* was not supported.

To address the last question—do the associations (discussed above) vary across families with different poverty histories—the models were re-estimated across poverty groups (i.e., no poverty, transient poverty, and persistent poverty). In regards to the association between parenting stress and couple's relationship quality, the results are presented in Table 4.4. First, for couples who did not experience poverty over the four survey years, Panel A (Model 1) shows the results for the random effects models for mothers. The results indicate that a standard deviation change in parenting stress is associated with a .15 standard deviation decrease in couple's relationship quality (p < 1.001). In Model 2, the fixed results demonstrate similar finding with a .08 standard deviation decrease in couple's relationship quality. For Model 3, the cross-lagged model, the results reveal a modest significant association for early maternal parenting stress Year-1 and later couple's relationship quality Year-3 (b = -.06; p < .10) and a significant association from Year-3 to Year-5 (b = -.07; p < .05). For fathers, the results are also consistent across the random and fixed effects results (b = -.11 and -.07) indicating that parenting stress is associated with lower levels of couple's relationship quality. The crosslagged model (see Model 3), however, yielded no significant results for fathers. All in all, the results show that, for couples who experience no poverty over time, parental stress consistently lowers couple's relationship quality for mothers but not fathers across models.

Second, for couples who experienced transient poverty across the four survey years (Panel B), the random and fixed effects models revealed no significant association between maternal parenting stress and couple's relationship quality. Moreover, the crosslagged association yielded no significant association from early parenting stress to later couple's relationship quality. For fathers (Panel B, Model 1), the random effects show that paternal parenting stress is modestly associated with a .07 standard deviation decrease in couple's relationship quality (p < .10). The fixed effects results also show a modest association (b = -.09; p < .10). In Model 3, the cross-lagged model reveals that the effects of paternal parenting was associated with couple's relationship quality only from Year-3 to Year 5 (b = -.17; p < .01).

Last, turning to couples who experienced persistent poverty over the four survey years (Panel C, Model 1), the random effects results indicate that a standard deviation change in maternal parenting stress is associated with a .12 standard deviation decrease in couple's relationship quality. The fixed effects (Panel C, Model 2) results showed a modest association (b = -.11; p < .10). The cross-lagged model revealed no significant association. For fathers, the results show that paternal parenting stress is associated with lower levels of couple's relationship quality in the random effects (b = -.12; p < .05) and fixed effects (b = -.14; p < .05) models. The cross-lagged model (Model 3) revealed no significant results for fathers, however.

The reverse association—the association between couple's relationship quality and parenting stress—across poverty histories was also examined. For couples who experienced no poverty across survey years (Panel A), the results revealed that for mothers, couple's relationship quality is associated with lower levels of maternal parenting stress for the random effects (b = -.13; p < .001, see Model 1) and the fixed effects (b = -.07; p < .05, see Model 2). The cross-lagged model shows that couple's relationship quality at Year 1 was associated with a .08 standard deviation decrease in maternal parenting stress (p < .05) at Year-3. The findings were not significant from
Years 3 to 5, however. For fathers, the results show that couple's relationship quality is associated with a decrease in paternal parenting stress for the random effects (b = -.09; p < .001) and the fixed effects (b = -.06; p < .05) models. No significant results emerged for fathers in the cross-lagged model.

For couples who experience transient poverty across time (Panel B), couple's relationship quality was not significantly associated with maternal parenting stress for the random and fixed effects models, as well as the cross-lagged model. For fathers, the random effects model displayed a significant but modest association (b = -.09; p < .10) and the fixed effects model also yielded modest results (b = -.08; p < .10). The cross-lagged results revealed no significant results for fathers. These findings suggest couples who experience transient poverty, relationship quality does not affect parental stress for both parents.

For couples who experienced persistent poverty across the four survey years, the random effects show that couple's relationship quality is associated with lower levels of maternal parenting stress (b = -.13; p < .05) and the fixed effects model also revealed a modest association (b = -.12; p < .10). In the cross-lagged model, the results were not significant from Year-1 to Year-3; however, couple's relationship quality at Year-3 was associated with a .15 standard deviation decrease in maternal parenting stress at Year-5. In regards to fathers, the random effects and fixed models yielded significant results (b = -.14 and -.15, respectively; p < .05). The cross-lagged results, for fathers, revealed no significant associations. Overall, couple's relationship quality consistently lowers parental stress for mothers across models but not for fathers.

DISCUSSION

Extant research shows that the demands of having a young child is associated with an increase in the day-to-day hassles for many parents, which generates elevated levels of stress and distress for many parents (e.g., Umberson et al. 2010) and leads to decreases in relationship quality among intimate partners (e.g., Umberson and Reczek 2007). Although prior studies revealed that stress in the parenting role and relationship quality are empirically related, these studies have been limited to small, non-representative samples. Moreover, the direction of the association between parental stress and relationship quality is less clear. The current study extends prior research by using a representative and longitudinal sample of couples living in urban areas 1-, 3-, and 5-years after the birth of a child to examine the following questions: (a) is there an association between parenting stress and relationship quality; (b) does the association change over time as children age from infant to toddler, and from toddler to preschooler; and (c) does the association vary across families' poverty histories?

First, the results revealed that parental stress was associated with couple's relationship quality, and vice versa. Particularly, the findings corroborate previous research that indicates that stress in the parenting role lowers the quality of the relationship between partners (Lavee et al. 1996; Quittner et al. 1990), which gives credence to stress spillover research (Bolger et al. 1989) suggesting that stress in one family domain (e.g., parenting) affects the family functioning in another domain (e.g., relationship quality). Conversely, the current study also shows that couple's with higher levels of relationship quality between parents reduces parenting stress, supporting the idea that positive partner support works as a valuable resource in the time of stress (Lavee 2013; Randall and Bodenmann 2009; Umberson and Reczek 2007). Although

these findings are consistent with prior studies, the models do not account for the lagged scores or bi-directionality between parenting stress and couple's relationship quality. In many ways, these results are confirmatory, with the possibility that parameter estimates are biased—this is especially true for the random effects models due to omitted variables.

Second, with respect to mutual influence in the association between parenting stress and couple's relationship quality appeared to follow mostly in one direction—from early relationship quality to later parenting stress (for mothers only). Compared to the random- and fixed-effects results above, the cross-lagged approach appears to be more robust as it adequately controls for prior scores and reverse causation. The results suggest that a positive, supportive relationship is a helpful resource for mothers, which has long-term benefits in parenting a young child. The finding is extremely important because mothers are more likely to engage in child rearing responsibilities than fathers (Bianchi and Milkie 2010) and less stress in the parenting role can lead to positive parenting (Green et al. 2007) and subsequent favorable outcomes for children (e.g., Conger, Conger, and Martin 2010).

In contrast, there was no indication that early parenting stress affected later relationship quality between mother and fathers when controlling for lagged scores and reciprocal associations. These results are contrary to cross-sectional studies (e.g., Quittner et al. 1990; Raikes and Thompson 2005). The unidirectional finding suggests that supportive partners are much more of a valuable resource in reducing stress than the role of stress in undermining the quality of the romantic relationship. The current study moves this line of research forward by addressing and clarifying the directionality in the parenting stress and relationship quality nexus, with the results pointing to the significance of supportive intimate partnerships.

In regards to change over time, the cross-lagged results reveal that the association between parenting stress and couple's relationship quality remain relatively stable across survey years. Although prior studies suggest that young children increases parental stress and put strain on intimate partnerships (e.g., Nomaguchi and Milkie 2003), the association between these factors is similar across time as children develop. The general conclusion from these findings suggest that the association between couple's relationship quality and parenting stress is positively related for couples who are romantically involved five years after the birth of a child—with the results being more beneficial for mothers than fathers. Simply put, consistent positive partnerships equal less stress in the parenting role for mothers.

Last, in regards to variations in the association between families experiencing different poverty histories, the results yielded important and interesting results. The random and fixed effects results suggest that, regardless of families' poverty history, the association between parenting stress and couple's relationship quality was similar— although the results are less compelling for mothers and fathers who experience transient poverty. The directionality results were consistent with the full sample. That is, supportive relationship quality reduced maternal parenting stress. More specifically, whether under optimal circumstances (i.e., no poverty) or under adverse conditions (i.e., persistent poverty), having a supportive partner is *modestly* beneficial in reducing stress in the parenting role.

Although the results were largely similar across families experiencing different poverty histories, some additional findings are worth noting: First, no significant association emerged between parenting stress and couple's relationship quality for mothers who experience transient poverty. These results are somewhat surprising given the stress associated with poverty (e.g., Wadsworth 2012). Some studies, however, show that families can be resilient in the midst of poverty (e.g., Mullin and Arce 2008). To some degree, these findings are consistent with a prior study that found no significant association between couple's quality and parenting stress for parents living in poverty (Raikes and Thompson 2005). Second, paternal stress was associated with lower levels of couple's relationship quality for fathers who experience transient poverty. To speculate, fathers who experience transient poverty may find such variations particularly frustrating. As a consequence, parenting stress increases, leading to unfavorable interactions between parents. Last, the cross-lagged results show, among mothers experiencing no poverty and transient poverty, that supportive relationship quality between parents mattered more for decreasing maternal parenting stress when their child was younger (infant to toddler) whereas for mothers who experienced persistent poverty, supportive relationship mattered more when their child was older (toddler to preschool). The divergence in these findings may be a result of mothers' interpretation of having a child. For instance, although parenting can be stressful, for many low-income mothers having a child can give meaning to life (e.g., Edin and Kefalas 2005). Once a child ages to the preschool years and become more independent, especially living in persistent poverty, being in a positive, supportive relationship may be paramount to decreasing maternal stress.

Implications

The current study has implications for research. Specifically, the results highlights family systems theory which suggests that individuals within families are interdependent (Cox and Paley 1997) and the dyadic nature of stress and family functioning among parents (e.g., spillover; Bolger et al. 1989; Larson and Almeida 1999). The directionality results are more persuasive in regards to couple's relationship quality reduces parenting stress than parenting stress affecting relationship quality. Examining the association as children get older would be a valuable contribution for future research. Moreover, additional factors such as race/ethnicity, marital status, and child's health may be moderate the association between parenting stress and relationship and thus warrant consideration.

In regards to implications for policy, the current study can be particularly useful. For example, recent public policy has emphasized building and improving relationship quality among parents (Hawkins, Amato, and Kinghorn 2013). The findings in the current study demonstrates that improving couple's relationship skills to promote positive relationship quality may be valuable in reducing stress in the parenting role, especially for mothers. This is particularly important given that maternal stress has far-reaching implications for children's well-being (Bornstein and Bradley 2003). Interestingly, although current policy aims at low-income families (Dion 2005), the findings suggest that promoting supportive relationship quality can be beneficial for all parents in terms of parenting stress, regardless of socioeconomic status (e.g., poverty histories). Moreover, given that many family policies target mothers, targeting both parents could yield beneficial results (e.g., Carlson et al. 2011).

Limitations

Although the current study makes an important contribution to the study of parenting stress, relationship quality, and poverty histories, there are notable limitations that must be discussed. First, although longitudinal designs are beneficial in many regards, some respondents drop out of the original sample—through attrition or relationship dissolution. Given that many parents leave the study are disadvantaged, display higher levels of parenting stress, and lower levels of relationship quality, the parameter estimates may be biased downward for parenting stress and biased upward for relationship quality. Second, one obstacle for survey research is missing variable bias. Although the current study employs several analytic techniques to the test robustness of the association, each technique is vulnerable to the threat to missing variable bias. Third, the parenting stress measure reflects parents' responses to broad notions of parenting stress and may miss particular hassles associated with the parenting role. Last, given that the sample reflects couples living in urban areas who had a child in the late 1990s, the generalizability of the results are limited to this population and not the general population.

Strengths

Despite the limitations, there are several strengths. First, the current study uses a large, diverse, urban sample of mothers and fathers who had a child in the 1990s, and the families were followed over time and re-interviewed when their child was an infant, toddler, and in preschool. Early studies on parenting stress have been limited by small, homogenous samples (i.e., middle-class white families). Second, taking advantage of the panel design, the current study examines the longitudinal and reciprocal associations between parenting stress and couple's relationship. Last, although several studies have examined parenting stress and relationship quality in the context of economic inequality

(e.g., low-income, economic hardship, poverty), many of these studies have been crosssectional. Given the dynamic nature of poverty, understanding the association between parenting stress and relationship quality over time simultaneously as families move in and out of poverty may be fruitful to understanding how economic inequality affects individual and family processes.

CONCLUSION

The current study takes an important step in addressing the association between parenting stress and couple's relationship quality in the context of poverty using a large sample of urban parents who had a child in the late 1990s. The longitudinal design permits the use of multiple empirical techniques to examine the aforementioned association during their child's early developmental period: infant, toddler, and preschooler. The findings reveal consistent evidence that couple's supportive relationship quality is a valuable resource that protects mothers from elevated levels of stress in the parenting role, even across poverty histories. Simply put, being in a supportive partnership means less parenting stress for mothers. Future research can move this line of inquiry forward by examining parenting stress and relationship quality as children get older and gain more independence.

REFERENCES

- Abidin, Richard R. 1992. "The Determinants of Parenting Behavior." *Journal of Clinical Child Psychology* 21(4):407–12.
- Acock, Alan C. 2005. "Working with Missing Values." *Journal of Marriage and Family* 67(4):1012–28.
- Allison, P. D. 2002. "Missing Data: Quantitative Applications in the Social Sciences." British Journal of Mathematical and Statistical Psychology 55(1):193–96.
- Belsky, Jay. 1984. "The Determinants of Parenting: A Process Model." *Child Development* 55(1):83–96.
- Bendheim-Thomas Center on Child Wellbeing. 2008. Introduction to the Fragile Families Public Use Data. Princeton University.
- Benzies, Karen M., Margaret J. Harrison, and Joyce Magill-Evans. 2004. "Parenting Stress, Marital Quality, and Child Behavior Problems at Age 7 Years." *Public Health Nursing* 21(2):111–21.
- Bianchi, Suzanne M., and Melissa A. Milkie. 2010. "Work and Family Research in the First Decade of the 21st Century." *Journal of Marriage and Family* 72(3):705–25.
- Bolger, Niall, Anita DeLongis, Ronald C. Kessler, and Elaine Wethington. 1989. "The Contagion of Stress across Multiple Roles." *Journal of Marriage and the Family* 175–83.
- Bornstein, Marc H., and Robert H. Bradley. 2003. Socioeconomic Status, Parenting, and Child Development. Routledge.
- Bradbury, Thomas N., Frank D. Fincham, and Steven R. H. Beach. 2000. "Research on the Nature and Determinants of Marital Satisfaction: A Decade in Review." *Journal of Marriage and Family* 62(4):964–80.
- Bradbury, Thomas N., Frank D. Fincham, and Steven RH Beach. 2000. "Research on the Nature and Determinants of Marital Satisfaction: A Decade in Review." *Journal* of Marriage and Family 62(4):964–80.
- Bradbury, Thomas N., and Benjamin R. Karney. 2004. "Understanding and Altering the Longitudinal Course of Marriage." *Journal of Marriage and Family* 66(4):862–79.
- Bronte-Tinkew, Jacinta, Allison Horowitz, and Jennifer Carrano. 2010. "Aggravation and Stress in Parenting: Associations with Coparenting and Father Engagement among Resident Fathers." *Journal of Family Issues* 31(4):525–55.
- Brown, Susan L., and Alan Booth. 1996. "Cohabitation versus Marriage: A Comparison of Relationship Quality." *Journal of Marriage and the Family* 58(3):668–78.

- Bulanda, Jennifer Roebuck, and Susan L. Brown. 2007. "Race-Ethnic Differences in Marital Quality and Divorce." *Social Science Research* 36(3):945–67.
- Carlson, Marcia J., Natasha V. Pilkauskas, Sara S. McLanahan, and Jeanne Brooks-Gunn. 2011. "Couples as Partners and Parents over Children's Early Years." *Journal of Marriage and Family* 73(2):317–34.
- Ceballo, Rosario, and Vonnie C. McLoyd. 2002. "Social Support and Parenting in Poor, Dangerous Neighborhoods." *Child Development* 73(4):1310–21.
- Cellini, S. R., S. M. McKernan, and C. Ratcliffe. 2008. "The Dynamics of Poverty in the United States: A Review of Data, Methods, and Findings." *Journal of Policy Analysis and Management* 27(3):577–605.
- Coleman, Priscilla K., and Katherine H. Karraker. 1998. "Self-Efficacy and Parenting Quality: Findings and Future Applications." *Developmental Review* 18(1):47–85.
- Conger, Rand D., Katherine J. Conger, and Monica J. Martin. 2010. "Socioeconomic Status, Family Processes, and Individual Development." *Journal of Marriage and Family* 72(3):685–704.
- Cooper, Carey E., Sara S. McLanahan, Sarah O. Meadows, and Jeanne Brooks-Gunn. 2009. "Family Structure Transitions and Maternal Parenting Stress." *Journal of Marriage and Family* 71(3):558–74.
- Corcoran, Mary E., and Ajay Chaudry. 1997. "The Dynamics of Childhood Poverty." *The Future of Children* 40–54.
- Cowan, Carolyn Pape, and Philip A. Cowan. 2000. *When Partners Become Parents: The Big Life Change for Couples.* Lawrence Erlbaum Associates Publishers.
- Cox, Martha J., and Blair Paley. 1997. "Families as Systems." Annual Review of Psychology 48(1):243–67.
- Crnic, Keith A., Catherine Gaze, and Casey Hoffman. 2005. "Cummulative Parenting Stress Across the Preschool Period: Relations to Maternal Parenting and Child Behavior at Age 5." *Infant and Child Development* 14:117–32.
- Crnic, Keith A., Christine Low, and M. Bornstein. 2002. "Everyday Stresses and Parenting." *Handbook of Parenting: Vol. 5. Practical Issues in Parenting* 243–67.
- Crnic, Keith, and M. Acevedo. 1995. "Everyday Stresses and Parenting." Pp. 277–97 in *Handbook of Parenting*. Mahwah, NJ: Lawrence Erlbaum.
- Cui, Ming, M. Brent Donnellan, and Rand D. Conger. 2007. "Reciprocal Influences between Parents' Marital Problems and Adolescent Internalizing and Externalizing Behavior." *Developmental Psychology* 43(6):1544.

- Cunradi, Carol B., Raul Caetano, Catherine Clark, and John Schafer. 2000.
 "Neighborhood Poverty as a Predictor of Intimate Partner Violence Among White, Black, and Hispanic Couples in the United States: A Multilevel Analysis." Annals of Epidemiology 10(5):297–308.
- Cutrona, Carolyn E. 1996. Social Support in Couples: Marriage as a Resource in Times of Stress. Sage Publications, Inc.
- Deater-Deckard, Kirby. 2008. Parenting Stress. Yale University Press.
- Deater-Deckard, Kirby, and Sandra Scarr. 1996. "Parenting Stress among Dual-Earner Mothers and Fathers: Are There Gender Differences?" *Journal of Family Psychology* 10(1):45–59.
- Dion, M. Robin. 2005. "Healthy Marriage Programs: Learning What Works." *The Future* of Children 15(2):139–56.
- Don, Brian Paul, and Kristin D. Mickelson. 2012. "Paternal Postpartum Depression: The Role of Maternal Postpartum Depression, Spousal Support, and Relationship Satisfaction." *Couple and Family Psychology: Research and Practice* 1(4):323.
- Edin, Kathryn, and Maria Kefalas. 2005. *Promises I Can Keep: Why Poor Women Put Motherhood Before Marriage*. Berkeley, CA: University of California Press.
- Edin, Kathryn, and Rebecca Joyce Kissane. 2010. "Poverty and the American Family: A Decade in Review." *Journal of Marriage and Family* 72(3):460–79.
- Edwards, Ashley N. 2014. *Dynamics of Economic Well-Being: Poverty, 2009-2011*. U.S. Census Bureau.
- Enders, C. K., and D. L. Bandalos. 2001. "The Relative Performance of Full Information Maximum Likelihood Estimation for Missing Data in Structural Equation Models." *Structural Equation Modeling* 8(3):430–57.
- Evenson, Ranae J., and Robin W. Simon. 2005. "Clarifying the Relationship between Parenthood and Depression." *Journal of Health and Social Behavior* 46(4):341– 58.
- Fincham, Frank D., and Steven R. H. Beach. 2010. "Marriage in the New Millennium: A Decade in Review." *Journal of Marriage and Family* 72(3):630–49.
- Fincham, Frank D., and Steven R. H. Beach. 2010. "Of Memes and Marriage: Toward a Positive Relationship Science." *Journal of Family Theory & Review* 2(1):4–24.
- Gelfand, Donna M., Douglas M. Teti, and Cathie E. Radin Fox. 1992. "Sources of Parenting Stress for Depressed and Nondepressed Mothers of Infants." *Journal of Clinical Child and Adolescent Psychology* 21(3):262–72.

- Green, Beth L., Carrie Furrer, and Carol McAllister. 2007. "How Do Relationships Support Parenting? Effects of Attachment Style and Social Support on Parenting Behavior in an at-Risk Population." *American Journal of Community Psychology* 40(1-2):96–108.
- Greene, William H. 2010. Econometric Analysis. Seventh. Boston, MA: Prentice Hall.
- Gupta, Vidya Bhushan. 2007. "Comparison of Parenting Stress in Different Developmental Disabilities." *Journal of Developmental and Physical Disabilities* 19(4):417–25.
- Gyamfi, Phyllis, Jeanne Brooks-Gunn, and Aurora P. Jackson. 2001. "Associations between Employment and Financial and Parental Stress in Low-Income Single Black Mothers." *Women & Health* 32(1-2):119–35.
- Halliday Hardie, Jessica, and Amy Lucas. 2010. "Economic Factors and Relationship Quality Among Young Couples: Comparing Cohabitation and Marriage." *Journal* of Marriage and Family 72(5):1141–54.
- Hardie, Jessica Halliday, and Amy Lucas. 2010. "Economic Factors and Relationship Quality among Young Couples: Comparing Cohabitation and Marriage." *Journal* of Marriage & Family 72(5):1141–54.
- Hawkins, Alan J., Paul R. Amato, and Andrea Kinghorn. 2013. "Are Government-Supported Healthy Marriage Initiatives Affecting Family Demographics? A State-Level Analysis." *Family Relations* 62(3):501–13.
- Kirby, Jennifer S., Donald H. Baucom, and Michael A. Peterman. 2005. "An Investigation of Unmet Intimacy Needs in Marital Relationships." *Journal of Marital and Family Therapy* 31(4):313–25.
- Kotchick, Beth A., Shannon Dorsey, and Laurie Heller. 2005. "Predictors of Parenting among African American Single Mothers: Personal and Contextual Factors." *Journal of Marriage and Family* 67(2):448–60.
- Krieg, Dana Balsink. 2007. "Does Motherhood Get Easier the Second-Time around? Examining Parenting Stress and Marital Quality among Mothers Having Their First or Second Child." *Parenting: Science and Practice* 7(2):149–75.
- Larson, Reed W., and David M. Almeida. 1999. "Emotional Transmission in the Daily Lives of Families: A New Paradigm for Studying Family Process." *Journal of Marriage and the Family* 61(1).
- Lavee, Yoav. 2013. "Stress Processes in Families and Couples." Pp. 159–76 in Handbook of Marriage and the Family. Springer.

- Lavee, Yoav, Shlomo Sharlin, and Ruth Katz. 1996. "The Effect of Parenting Stress on Marital Quality: An Integrated Mother-Father Model." *Journal of Family Issues* 17(1):114–35.
- Loper, Ann Booker, L. Wrenn Carlson, Lacey Levitt, and Kathryn Scheffel. 2009. "Parenting Stress, Alliance, Child Contact, and Adjustment of Imprisoned Mothers and Fathers." *Journal of Offender Rehabilitation* 48(6):483–503.
- Lopoo, Leonard M., and Bruce Western. 2005. "Incarceration and the Formation and Stability of Marital Unions." *Journal of Marriage and Family* 67(3):721–34.
- McBride, Brent A., Sarah J. Schoppe, and Thomas R. Rane. 2002. "Child Characteristics, Parenting Stress, and Parental Involvement: Fathers versus Mothers." *Journal of Marriage and Family* 64(4):998–1011.
- McLeod, Jane D., and Michael J. Shanahan. 1993. "Poverty, Parenting, and Children's Mental Health." *American Sociological Review* 351–66.
- McLoyd, Vonnie C. 1995. "Poverty, Parenting, and Policy: Meeting the Support Needs of Poor Parents." Pp. 269–303 in *Children in Poverty: Research, Health, and Policy Issues*. Vol. 23. NY: Springer.
- Middlemiss, Wendy. 2003. "Poverty, Stress, and Support: Patterns of Parenting Behaviour among Lower Income Black and Lower Income White Mothers." *Infant and Child Development* 12(3):293–300.
- Mullin, Walter J., and Miguel Arce. 2008. "Resilience of Families Living in Poverty." *Journal of Family Social Work* 11(4):424–40.
- Mulsow, Miriam, Yvonne M. Caldera, Marta Pursley, Alan Reifman, and Aletha C. Huston. 2002. "Multilevel Factors Influencing Maternal Stress during the First Three Years." *Journal of Marriage and Family* 64(4):944–56.
- Neff, Lisa A., and Benjamin R. Karney. 2007. "Stress Crossover in Newlywed Marriage: A Longitudinal and Dyadic Perspective." *Journal of Marriage and Family* 69(3):594–607.
- Nelson, S. Katherine, Kostadin Kushlev, Tammy English, Elizabeth W. Dunn, and Sonja Lyubomirsky. 2013. "In Defense of Parenthood Children Are Associated With More Joy Than Misery." *Psychological Science* 24(1):3–10.
- Nomaguchi, Kei, and Amanda N. House. 2013. "Racial-Ethnic Disparities in Maternal Parenting Stress The Role of Structural Disadvantages and Parenting Values." *Journal of Health and Social Behavior* 54(3):386–404.
- Nomaguchi, Kei M., and Melissa A. Milkie. 2003. "Costs and Rewards of Children: The Effects of Becoming a Parent on Adults' Lives." *Journal of Marriage and Family* 65(2):356–74.

- O'Brien, Marion. 2005. "Studying Individual and Family Development: Linking Theory and Research." *Journal of Marriage and Family* 67(4):880–90.
- Ostberg, Monica, and Berit Hagekull. 2000. "A Structural Modeling Approach to the Understanding of Parenting Stress." *Journal of Clinical Child Psychology* 29(4):615–25.
- Putnick, Diane L. et al. 2010. "Stability, Continuity, and Similarity of Parenting Stress in European American Mothers and Fathers across Their Child's Transition to Adolescence." *Parenting: Science and Practice* 10(1):60–77.
- Quittner, Alexandra L., Robert L. Glueckauf, and Douglas N. Jackson. 1990. "Chronic Parenting Stress: Moderating versus Mediating Effects of Social Support." *Journal of Personality and Social Psychology* 59(6):1266.
- Raikes, H. Abigail, and Ross A. Thompson. 2005. "Efficacy and Social Support as Predictors of Parenting Stress among Families in Poverty." *Infant Mental Health Journal* 26(3):177–90.
- Randall, Ashley K., and Guy Bodenmann. 2009. "The Role of Stress on Close Relationships and Marital Satisfaction." *Clinical Psychology Review* 29(2):105– 15.
- Reichman, Nancy E., Julien O. Teitler, Irwin Garfinkel, and Sara S. McLanahan. 2001. "Fragile Families: Sample and Design." *Children and Youth Services Review* 23(4/5):303–26.
- Rodriguez-JenKins, Jessica, and Maureen O. Marcenko. 2014. "Parenting Stress among Child Welfare Involved Families: Differences by Child Placement." *Children and Youth Services Review*.
- Rubin, D. B., and R. J. A. Little. 2002. "Statistical Analysis with Missing Data." *Hoboken, NJ: J Wiley & Sons.*
- Schulz, Marc S., Carolyn Pape Cowan, and Philip A. Cowan. 2006. "Promoting Healthy Beginnings: A Randomized Controlled Trial of a Preventive Intervention to Preserve Marital Quality during the Transition to Parenthood." *Journal of Consulting and Clinical Psychology* 74(1):20.
- Suarez, Liza M., and Bruce L. Baker. 1997. "Child Externalizing Behavior and Parents' Stress: The Role of Social Support." *Family Relations* 373–81.
- Thoits, Peggy A. 2011. "Mechanisms Linking Social Ties and Support to Physical and Mental Health." *Journal of Health and Social Behavior* 52(2):145–61.
- Turner, R. Jay, J. Blake Turner, and William Beardall Hale. 2014. "Social Relationships and Social Support." Pp. 1–20 in *Sociology of Mental Health*. Springer.

- Umberson, Debra, and Jennifer Karas Montez. 2010. "Social Relationships and Health: A Flashpoint for Health Policy." *Journal of Health and Social Behavior* 51(1 suppl):S54–S66.
- Umberson, Debra, Tetyana Pudrovska, and Corinne Reczek. 2010. "Parenthood, Childlessness, and Well-Being: A Life Course Perspective." *Journal of Marriage and Family* 72(3):612–29.
- Umberson, Debra, and Corinne Reczek. 2007. "Interactive Stress and Coping Around Parenting: Explaining Trajectories of Change in Intimate Relationships Over the Life Course." Advances in Life Course Research 12:87–121.
- Umberson, Debra, Kristi Williams, Daniel A. Powers, Meichu D. Chen, and Anna M. Campbell. 2005. "As Good as It Gets? A Life Course Perspective on Marital Quality." *Social Forces* 84(1):493–511.
- Umberson, Debra, Kristi Williams, Daniel A. Powers, Hui Liu, and Belinda Needham. 2005. "Stress in Childhood and Adulthood: Effects on Marital Quality over Time." *Journal of Marriage and Family* 67(5):1332–47.
- Wadsworth, Martha E. 2012. "Working with Low-Income Families: Lessons Learned from Basic and Applied Research on Coping with Poverty-Related Stress." *Journal of Contemporary Psychotherapy* 42(1):17–25.
- Winship, Christopher, and Larry Radbill. 1994. "Sampling Weights and Regression Analysis." *Sociological Methods and Research* 23(2):230–57.



Figure 4.1: Conceptual Model for Parenting Stress, Couple's Relationship Quality, Poverty Histories Over Time

	All Cou	ıples	No pov	erty	Transient p	overty	Persistent	poverty
	M or %	SD	M or %	SD	M or %	SD	M or %	SD
Marital Status (Baseline)								
Married								
Cohabiting	25.14		14.20		56.30		46.60	
Parent's Age								
Mother's Age	28.96	5.63	30.54	5.16	24.85	5.36	25.51	4.32
Father's Age	31.05	6.15	32.57	5.68	26.85	6.35	27.98	4.90
Mother's Race/Ethnicity								
Non-Hispanic White	44.04		58.60		12.30		5.47	
Non-Hispanic Black	18.61		15.50		34.90		16.80	
Hispanic	29.11		15.30		48.20		77.30	
Non-Hispanic Other	8.24		10.60		4.56		.42	
Parents are a different race/ethnicity	10.97		13.10		7.43		4.14	
Mother's Education								
Less than High School	17.40		4.10		34.10		65.60	
High school diploma or equivalent	24.46		18.70		45.00		31.30	
Some college	21.14		26.10		15.10		2.98	
Bachelor's degree or higher	36.99		51.10		5.83		.07	
Father's Education								
Less than High School	19.01		5.42		33.60		70.90	
High school diploma or equivalent	16.18		13.30		29.70		16.10	
Some college	28.23		30.80		31.40		12.10	
Bachelor's degree or higher	36.59		50.40		5.36		0.91	
Parent's Physical Health								
Mother	4.08	0.83	4.24	0.77	3.77	0.79	3.60	0.91
Father	4.05	0.90	4.19	0.79	3.72	0.96	3.73	1.15
Parent's Mental Health								
Mother	6.42		5.63		2.88		14.00	
Father	5.20		4.36		11.10		3.20	
Employment Status								
Mother has job 1yr	59.54		71.60		43.20		17.20	
Father has job 1yr	92.16		95.50		79.00		89.10	
Social Support								
Mother received social support	23.05		21.40		26.20		27.90	
Father received social support	29.35		31.50		27.90		20.20	
Incarceration History								
Father Incarcerated	7.29		4.84		15.00		11.40	
# of children in HH	0.94	1.20	0.73	0.89	1.474	1.50	1.416	1.76
Fertility History								

Table 4.16: Descriptive Statistics, by Poverty Histories

Table 4.1, continued

	All Cou	ples	No pov	erty	Transient p	overty	Persistent poverty	
	M or %	SD	M or %	SD	M or %	SD	M or %	SD
Couples Higher Order Birth (Yes=1)	49.32		45.6		63.60		52.70	
Multi-partnered Fertility								
Neither parent has a child by another partner	73.79		81.5		51.20		59.10	
Father has a child by another partner	9.70		7.54		14.20		15.70	
Mother has a child by another partner	11.26		7.16		25.80		16.30	
Both parents have a child by another partner	5.26		3.79		8.77		8.87	
Below poverty line = 1 (Baseline)	13.49							
Child Characteristics								
Child Health(Yr1)	4.63	0.65	4.69	0.58	4.73	0.57	4.23	0.85
Boy=1	60.02		57.3		55.70		78.20	
Child Temperament	2.43	0.67	2.35	0.59	2.52	0.75	2.74	0.83
N	942		596		207		139	

Table 4.17: Means on Parenting Stress and Couple's Relationship Quality

	All Co	uples	No Po	verty	Transient	Poverty	Persistent	Poverty
	Mean	SD	Mean	SD	Mean	SD	Mean	SD
Parenting Stress (Mothers)								
Year 1	2.16	0.60	2.15	0.60	2.10	0.60	2.26	0.60
Year 3	2.29	0.59	2.29	0.59	2.37	0.62	2.22	0.57
Year 5	2.23	0.58	2.22	0.55	2.21	0.71	2.33	0.61
Parenting Stress (Fathers)								
Year 1	2.08	0.61	2.08 ^{ab}	0.55	2.23 ^{ac}	0.58	1.90 ^{bc}	0.83
Year 3	2.10	0.64	2.14	0.60	2.02	0.63	2.03	0.79
Year 5	1.98	0.64	2.02 ^d	0.61	1.90	0.62	1.86 ^d	0.80
Relationship Quality (Couple)								
Year 1	2.76	0.21	2.78 ^{ef}	0.19	2.72 ^e	0.23	2.72 ^f	0.23
Year 3	2.77	0.19	2.77 ^g	0.19	2.73 ^g	0.20	2.77	0.20
Year 5	2.75	0.23	2.77 ^h	0.22	2.69 ^h	0.23	2.73	0.27

Note: means with identical superscripts denotes statistical difference.

	Model 1: Rand	om Effects	Model 2: Fixed	l Effects	Model 3: Ci	ross-Lagged
	В	SE	В	SE	В	SE
Mothers						
Parenting Stress> Rel. Quality	12***	0.02	-0.06*	0.02	-0.04	0.03 ^a
					-0.04	0.03 ^b
Rel. Quality> Parenting Stress	11***	0.02	-0.05*	0.02	-0.07*	0.03ª
					-0.06*	0.03 ^b
athers						
Parenting Stress> Rel. Quality	11***	0.02	09***	0.02	-0.04	0.03 ^a
					-0.04	0.03 ^b
Rel. Quality> Parenting Stress	10***	0.02	-0.08***	0.02	-0.03	0.03 ^a
					-0.02	0.03 ^b

Table 4.18: Standardized Parameter Estimates

a = Year 1 to Year 3; b = Year 3 to Year 5; ***p < .001; **p < .01; *p < .05; †p < .10

Table 4.19: Standardized Parameter Estimates, by Poverty Histories

		Pa	nel A: No	Poverty				P	Panel B: Tra	nsient P	overty			Panel C: Persistent Poverty				
	Model Rando Effec	1: om ts	Mode Fixe Effe	el 2: ed cts	Mode Cro Lag	el 3: ss- ged	Mode Rand Effe	el 1: lom cts	Mod Fix Effe	el 2: ted ects	Mod Cro Lag	el 3: ss- ged	Moo Ran Eff	lel 1: dom fects	Mod Fiz Effe	lel 2: ked ects	Mod Cro Lag	lel 3: oss- ged
	В	SE	В	SE	В	SE	В	SE	В	SE	В	SE	В	SE	В	SE	В	SE
Mothers																		
Parenting Stress> Rel. Quality	15***	.03	08*	.03	06†	.04ª	04	.04	.02	.05	006	.06 ^a	12*	.05	11†	.06	.07	.08ª
					07*	.03 ^b					001	.06 ^b					02	.08 ^b
Rel. Quality> Parenting Stress	13***	.02	07*	.03	08*	.03ª	.03	.04	.02	.05	103	.06ª	13*	.05	12†	.06	.06	.07ª
					03	.03 ^b					09	.06 ^b					15*	.07 ^b
Fathers																		
Parenting Stress> Rel. Quality	11***	.03	07*	.03	05	.03ª	07†	.04	09†	.05	.02	.06ª	12*	.05	14*	.06	.04	.08ª
					.01	.03 ^b					17**	.06 ^b					.004	.07 ^b
Rel. Quality> Parenting Stress	09***	.02	06*	.03	02	.03ª	09†	.04	08†	.05	.03	.06ª	14**	.05	15*	.06	001	.07ª
					02	.03 ^b					.02	.06 ^b					02	.07 ^b

***p < .001; **p < .01; *p < .05; †p < .10; Note: a = Year 1 to Year 3; b = Year 3 to Year 5

	Moth	ner	Fathe	r
	exp(b)	t	exp(b)	t
Parenting Stress(Yr1)	0.89	-0.77	1.10	-0.75
Couple's Relationship Quality (Yr1)	1.19	-0.52	0.66	-1.38
Cohabiting(Baseline)	1.07	-0.29	2.121**	-3.22
Mother's Age	1.01	-0.63	0.99	-0.41
Non-Hispanic Black	1.19	-0.64	0.76	-1.10
Hispanic	1.931*	-2.49	1.46	-1.54
Non-Hispanic Other	2.03	-1.59	0.45	-1.29
High School Diploma or Equivalent	0.76	-1.17	0.91	-0.45
Some College	0.78	-1.00	0.80	-0.88
Bachelor's Degree or higher	0.526+	-1.67	0.93	-0.22

Table 4.20: Attrition for Mothers and Fathers between Years 1 and Year 3

***p < .001; **p < .01; *p < .05; †p < .10

Table 4.21: Attrition for Mothers and Fathers between Year 1 and Year 3

	Moth	ers	Father	S
	exp(b)	t	exp(b)	t
Parenting Stress (Yr3)	1.13	-0.86	0.91	-0.70
Couple's Relationship Quality (Yr3)	1.03	-0.09	1.39	-1.01
Cohabiting(Baseline)	1.15	-0.63	1.14	-0.62
Mother's Age	1.02	-0.90	0.98	-1.28
Non-Hispanic Black	0.92	-0.35	1.09	-0.34
Hispanic	1.46	-1.53	1.00	-0.01
Non-Hispanic Other	1.921+	-1.72	0.91	-0.17
High School Diploma or Equivalent	0.70	-1.40	1.603*	-2.16
Some College	1.04	-0.15	0.576+	-1.88
Bachelor's Degree or higher	0.73	-0.93	0.64	-1.21

	Mothe	r	Father	
	exp(b)	t	exp(b)	t
Parenting Stress (Yr1)	1.01	-0.07	0.830+	-1.74
Couple's Relationship Quality (Yr1)	0.331***	-4.68	0.318***	-4.56
Cohabiting(Baseline)	2.763***	-4.98	2.584***	-4.73
Mother's Age	0.940***	-4.16	0.957***	-3.57
Non-Hispanic Black	1.589**	-2.61	1.519*	-2.11
Hispanic	0.548**	-2.70	0.578*	-2.29
Non-Hispanic Other	0.95	-0.13	1.76	-1.46
High School Diploma or Equivalent	1.15	-0.83	1.23	-1.21
Some College	1.10	-0.47	0.71	-1.64
Bachelor's Degree or higher	0.485+	-1.93	0.175***	-3.50

Table 4.22: Relationship Dissolution for Mothers and Fathers between Year 1 and Year 3

***p < .001; **p < .01; *p < .05; †p < .10

Table 4.23: Relationship Dissolution for Mothers and Fathers between Year 3 and Year 5

	Mother	-s	Fathers	3
	exp(b)	t	exp(b)	t
Parenting Stress (Yr3)	0.96	-0.42	0.99	-0.12
Couple's Relationship Quality (Yr3)	0.263***	-5.99	0.263***	-5.46
Cohabiting(Baseline)	3.209***	-5.89	2.852***	-5.17
Mother's Age	0.972*	-2.06	0.976*	-2.00
Non-Hispanic Black	1.530*	-2.30	1.926**	-2.97
Hispanic	0.699+	-1.70	1.18	-0.69
Non-Hispanic Other	1.10	-0.23	0.54	-0.97
High School Diploma or Equivalent	0.87	-0.81	1.03	-0.15
Some College	0.698+	-1.86	0.85	-0.75
Bachelor's Degree or higher	0.519*	-2.09	0.81	-0.65

			М	others					F	athers		
	Random	Effects		Cross-	Lagged		Random F	ffects		Cross-	Lagged	
	Relationshi	p Quality	Relationship Qua	ulity (Year 3)	Relationship Qua	lity (Year 5)	Relationship	Quality	Relationship Qua	lity (Year 3)	Relationship Qua	lity (Year 5)
	В	SE	В	SE	В	SE	В	SE	В	SE	В	SE
Relationship Quality (Year-1 lagged)	_	-	1.90***	0.09	1.01***	0.12	-		1.87***	0.09	1.03***	0.11
Relationship Quality (Year-3 lagged)	_	-	—		0.38***	0.03	-		—	_	0.37***	0.03
Cohabitation = 1	-0.12†	0.07	-0.06	0.08	-0.18*	0.07	-0.09	0.07	-0.06	0.07	-0.18**	0.07
Non-Hispanic Black	-0.18*	0.07	-0.05	0.08	-0.07	0.07	-0.14*	0.07	-0.05	0.07	-0.05	0.07
Hispanic	0.02	0.07	-0.03	0.08	0.00	0.08	0.04	0.07	-0.02	0.08	0.02	0.07
Non-Hispanic Other	-0.12	0.12	0.07	0.13	0.12	0.13	-0.15	0.12	0.06	0.13	0.13	0.13
Couple is mixed race	-0.10	0.08	-0.08	0.09	0.01	0.09	-0.06	0.08	-0.04	0.09	0.01	0.09
High school diploma or equivalent	0.12	0.08	0.07	0.09	0.00	0.08	-0.09	0.08	-0.10	0.09	-0.09	0.08
Some college	0.15†	0.09	0.03	0.09	-0.03	0.09	-0.01	0.08	-0.12	0.09	-0.05	0.08
Bachelor's degree or higher	0.13	0.10	0.03	0.11	-0.07	0.11	0.19†	0.10	0.02	0.11	0.03	0.10
Parent's Age (in years)	-0.01	0.01	0.00	0.01	0.00	0.01	-0.01	0.00	-0.01	0.01	0.00	0.01
Employed = 1	0.06	0.05	0.05	0.06	0.02	0.06	0.01	0.08	0.03	0.09	0.10	0.09
Social Support = 1	-0.12	0.06	-0.20**	0.06	-0.06	0.06	-0.05	0.06	-0.14*	0.06	0.07	0.06
Parent's Health	0.09**	0.03	0.04	0.03	0.01	0.03	0.06*	0.03	0.08*	0.03	0.04	0.03
Parent's Depression	-0.19*	0.09	-0.02	0.10	0.09	0.09	-0.55***	0.12	-0.37**	0.13	-0.19	0.12
Lived in poverty = 1 (baseline)	0.08	0.07	0.05	0.08	0.01	0.08	0.07	0.08	0.07	0.08	0.02	0.08
Higher-order birth = 1	0.05	0.06	0.09	0.07	0.02	0.07	0.06	0.06	0.10	0.07	0.03	0.06
# of children in $HH < 18$	-0.03	0.03	-0.03	0.03	-0.02	0.03	-0.03	0.03	-0.03	0.03	-0.01	0.03
Father's multipartnered fertility	-0.14†	0.08	-0.06	0.09	-0.02	0.09	-0.12	0.09	-0.01	0.07	-0.01	0.06
Mother's multipartnered fertility	-0.03	0.08	-0.05	0.09	-0.02	0.09	-0.04	0.08	-0.01	0.07	-0.01	0.06
Both parent's multipartnered fertility	0.11	0.10	-0.01	0.11	0.01	0.10	0.16	0.10	0.08	0.08	0.07	0.07
Father been incarcerated (Yr1)	-0.22**	0.08	0.09	0.09	-0.17*	0.08	-0.21*	0.08	-0.05	0.07	-0.04	0.06
Child's health	0.06	0.04	0.04	0.04	0.00	0.04	0.18*	0.04	0.06	0.04	0.00	0.04
Child is boy = 1	0.07	0.05	0.03	0.06	0.09†	0.05	0.07	0.05	0.03	0.06	0.09†	0.05
Child's Temperament	-0.07+	0.04	-0.04	0.04	0.07†	0.04	-0.09*	0.04	-0.05	0.04	0.07†	0.04

Table 4.24: Parameter Estimates for Control Variables on Relationship Quality (Y-Standardized)

			Mo	others					F	athers		
	Random	Effects		Cross	-Lagged		Random	Effects		Cross-	·Lagged	
	Parenting	g Stress	Parenting Stree	ss (Year 3)	Parenting Stree	ss (Year 5)	Parenting	Stress	Parenting Stre	ss (Year 3)	Parenting Stres	s (Year 5)
	В	SE	В	SE	В	SE	В	SE	В	SE	В	SE
Variables												
Parenting Stress (Year-1 lagged)		—	0.88***	0.04	0.40***	0.05			0.87***	0.03	0.51***	0.05
Parenting Stress (Year-3 lagged)	_	—	—	_	0.45***	0.03	-	_	_	_	0.39***	0.03
Cohabitation = 1	-0.03	0.07	-0.02	0.07	0.09	0.07	-0.06	0.07	-0.08	0.07	0.02	0.06
Non-Hispanic Black	-0.11	0.07	0.02	0.07	-0.22**	0.07	-0.11	0.07	-0.12†	0.07	0.00	0.07
Hispanic	-0.13†	0.07	-0.04	0.08	-0.17*	0.07	-0.12	0.08	-0.06	0.08	-0.04	0.07
Non-Hispanic Other	0.20	0.13	-0.08	0.13	-0.07	0.12	0.07	0.13	0.03	0.13	0.13	0.12
Couple is mixed race	-0.09	0.09	0.03	0.09	0.10	0.08	-0.01	0.09	-0.18*	0.09	-0.16†	0.08
High school diploma or equivalent	-0.02	0.08	0.00	0.08	0.01	0.08	0.07	0.08	0.16†	0.08	0.14†	0.08
Some college	0.05	0.09	0.07	0.09	0.07	0.08	-0.09	0.09	0.10	0.09	0.06	0.08
Bachelor's degree or higher	0.04	0.11	0.16	0.11	0.05	0.10	0.12	0.11	0.13	0.10	0.21*	0.10
Parent's Age (in years)	0.013*	0.006	0.01	0.01	0.01	0.01	-0.01	0.01	-0.01*	0.01	-0.01	0.01
Employed = 1	-0.13*	0.06	-0.08	0.06	0.00	0.05	-0.16†	0.09	-0.09	0.09	0.04	0.08
Social Support = 1	0.17**	0.06	0.14*	0.06	-0.03	0.06	0.05	0.06	0.02	0.06	-0.05	0.06
Parent's Health	-0.08*	0.03	-0.02	0.03	-0.01	0.03	-0.07*	0.03	-0.05†	0.03	-0.06†	0.03
Parent's Depression	0.33***	0.09	0.07	0.09	0.15†	0.09	0.12	0.12	-0.03	0.12	0.12	0.12
Lived in poverty = 1 (baseline)	0.12	0.08	0.01	0.08	0.14†	0.07	0.08	0.08	0.03	0.08	-0.13†	0.07
Higher-order birth = 1	0.06	0.06	0.06	0.07	-0.08	0.06	0.14*	0.07	0.05	0.07	-0.02	0.06
# of children in HH < 18	0.04	0.03	0.04	0.03	-0.03	0.03	0.01	0.03	0.01	0.03	0.03	0.03
Father's multipartnered fertility	-0.07	0.09	-0.12	0.09	0.13	0.08	-0.13	0.09	-0.08	0.06	-0.08	0.06
Mother's multipartnered fertility	-0.14	0.09	-0.05	0.09	0.00	0.08	-0.18*	0.09	-0.15*	0.06	-0.15*	0.06
Both parent's multipartnered fertility	-0.12	0.10	0.03	0.10	0.18†	0.10	-0.11	0.10	0.08	0.07	0.08	0.07
Father been incarcerated (Yr1)	-0.05	0.08	-0.02	0.08	0.03	0.08	0.17*	0.09	0.13*	0.06	0.13*	0.06
Child's health	-0.08†	0.04	-0.05	0.04	-0.09*	0.04	-0.03	0.04	-0.06	0.04	0.00	0.04
Child is boy = 1	0.06	0.05	0.03	0.05	0.04	0.05	0.09	0.05	0.01	0.05	0.01	0.05
Child's Temperament	0.17***	0.04	0.07†	0.04	0.01	0.04	0.01	0.04	0.05	0.04	-0.03	0.04

Table 4.25: Parameter Estimates for Control Variables on Parenting Stress (Y-Standardized)

			*						-			
	Random Relation	Effects Iship	Relationship Qu	Cross- ality (Year 3)	Lagged Relationship Qua	lity (Year 5)	Random	Effects nship	Relationship Qua	Cross- lity (Year 3)	Lagged Relationship Qual	lity (Year 5
	B	SE	В	SE	В	SE	B	SE	В	SE	В	SE
ariables												
Relationship Quality (Year-1 lagged)	—	—	2.11	0.12	1.16***	0.15	-	-	2.07***	0.12	1.17***	0.15
Relationship Quality (Year-3 lagged)	_	-	_	-	0.40***	0.04	-	-	—	—	0.41***	0.04
Cohabitation = 1	-0.14	0.09	-0.09	0.09	-0.12	0.09	-0.08	0.09	-0.05	0.09	-0.13	0.09
Non-Hispanic Black	-0.16†	0.09	-0.04	0.09	-0.09	0.09	-0.10	0.09	-0.03	0.09	-0.04	0.08
Hispanic	0.11	0.10	0.07	0.10	-0.08	0.10	0.18†	0.10	0.08	0.10	-0.03	0.10
Non-Hispanic Other	0.05	0.13	0.13	0.14	0.18	0.13	-0.01	0.14	0.14	0.14	0.15	0.14
Couple is mixed race	-0.05	0.11	0.00	0.11	-0.06	0.10	-0.04	0.11	0.01	0.11	-0.01	0.10
High school diploma or equivalent	0.22	0.16	0.14	0.16	0.20	0.15	-0.01	0.14	-0.04	0.14	-0.15	0.13
Some college	0.21	0.15	0.08	0.16	0.17	0.15	0.07	0.13	-0.07	0.14	-0.05	0.13
Bachelor's degree or higher	0.17	0.16	0.18	0.17	0.11	0.16	0.27†	0.14	0.07	0.14	-0.01	0.14
Parent's Age (in years)	-0.01	0.01	-0.01	0.01	0.00	0.01	0.00	0.01	-0.01	0.01	0.00	0.01
Employed = 1	0.09	0.07	0.04	0.08	-0.01	0.07	-0.01	0.15	0.28†	0.15	-0.01	0.14
Social Support = 1	-0.15*	0.07	-0.16*	0.08	-0.13†	0.07	-0.07	0.07	-0.16*	0.08	0.04	0.07
Parent's Health	0.13**	0.04	-0.01	0.05	-0.05	0.04	0.11†	0.04	0.12**	0.04	0.03	0.04
Parent's Depression	-0.07	0.11	-0.13	0.12	0.11	0.11	-0.57**	0.17	-0.45**	0.17	-0.18	0.16
Higher-order birth = 1	0.18*	1.09	0.17†	0.09	0.08	0.09	0.21*	0.09	0.18*	0.09	0.06	0.08
# of children in HH < 18	-0.08†	0.04	-0.09*	0.05	-0.02	0.04	-0.09*	0.04	-0.09*	0.04	-0.01	0.04
Father's multipartnered fertility	-0.31**	0.11	-0.26*	0.12	0.11	0.11	-0.24*	0.11	0.01	0.07	0.01	0.06
Mother's multipartnered fertility	0.02	0.12	0.11	0.12	0.02	0.12	0.04	0.12	0.01	0.06	0.01	0.06
Both parent's multipartnered fertility	-0.10	0.15	-0.05	0.16	0.09	0.15	-0.01	0.15	0.10	0.08	0.09	0.07
Father been incarcerated (Yr1)	-0.20	0.13	0.25†	0.14	-0.33*	0.13	-0.12	0.14	-0.08	0.07	-0.07	0.06
Child's health	0.07	0.05	0.05	0.06	-0.04	0.05	0.10†	0.05	0.06	0.06	-0.05	0.05
Child is boy = 1	0.11†	0.06	-0.01	0.07	0.12†	0.06	0.10	0.07	-0.02	0.07	0.11†	0.06
Child's Temperament	-0.13*	0.05	-0.07	0.06	0.04	0.05	-0.14**	0.05	-0.09	0.05	0.03	0.05

Table 4.26: Parameter Estimates for Control Variables on Relationship Quality (Y-Standardized), by No Poverty

Table 4.27: Parameter Estimates for Control	Variables on Parenting Stress (Y-Standardized), by No Poverty
Tuble 1.27. I diameter Estimates for Condo	i fulluoles on fullenting buess (i Standardized), by ito i overty

	Mot	her							Fa	thers			
	Random 1	Effects		Cross-	Lagged		Random	Effects	Cross-Lagged				
	Parenting	Stress	Parenting Stres	Parenting Stress (Year 3)		Parenting Stress (Year 5)		Parenting Stress		Parenting Stress (Year 3)		Parenting Stress (Year 5)	
Variables													
Parenting Stress (Year-1 lagged)		_	0.94***	0.05	0.49***	0.06	-	_	0.99***	0.05	0.65***	0.06	
Parenting Stress (Year-3 lagged)	—	—	_	_	0.45***	0.04	-		_	_	0.38***	0.04	
Cohabitation = 1	-0.05	0.09	-0.02	0.09	0.07	0.08	-0.03	0.09	-0.09	0.09	0.05	0.08	
Non-Hispanic Black	-0.11	0.09	0.06	0.09	-0.26**	0.08	-0.16†	0.09	-0.26**	0.08	0.02	0.08	
Hispanic	-0.31**	0.10	-0.16†	0.10	-0.19*	0.09	-0.11	0.10	0.02	0.10	0.13	0.09	
Non-Hispanic Other	0.14	0.13	-0.13	0.13	-0.21†	0.13	-0.04	0.14	-0.02	0.14	0.22†	0.13	
Couple is mixed race	-0.08	0.10	-0.01	0.11	0.20*	0.10	-0.03	0.11	-0.32**	0.11	-0.23*	0.10	
High school diploma or equivalent	-0.05	0.15	-0.06	0.15	0.07	0.14	-0.07	0.14	0.07	0.14	0.09	0.12	
Some college	0.08	0.15	0.00	0.15	0.14	0.14	-0.12	0.14	0.04	0.13	0.03	0.12	
Bachelor's degree or higher	0.05	0.16	0.11	0.16	0.14	0.15	0.08	0.15	0.03	0.14	0.22†	0.13	
Parent's Age (in years)	0.01	0.01	0.00	0.01	0.01	0.01	0.00	0.01	-0.01	0.01	-0.01	0.01	
Employed = 1	-0.18*	0.07	-0.11	0.07	-0.11†	0.07	0.06	0.15	0.12	0.15	0.13	0.13	
Social Support = 1	0.13†	0.07	0.16*	0.07	-0.02	0.07	0.16*	0.08	0.13†	0.07	0.07	0.07	
Parent's Health	-0.11**	0.04	0.02	0.04	-0.05	0.04	-0.11*	0.04	-0.09*	0.04	-0.06	0.04	
Parent's Depression	0.41***	0.11	0.09	0.11	0.12	0.11	0.33†	0.17	0.26	0.17	0.15	0.15	
Higher-order birth = 1	0.02	0.09	0.03	0.09	-0.01	0.08	0.12	0.09	-0.04	0.09	-0.01	0.08	
# of children in HH < 18	0.04	0.04	0.04	0.04	-0.07†	0.04	0.01	0.04	-0.02	0.04	0.08*	0.04	
Father's multipartnered fertility	-0.19†	0.11	-0.24*	0.11	0.30**	0.10	-0.05	0.12	-0.10	0.07	-0.09	0.06	
Mother's multipartnered fertility	-0.26*	0.12	-0.17	0.12	0.17	0.11	-0.19	0.12	-0.16	0.06	-0.15	0.06	
Both parent's multipartnered fertility	-0.20	0.15	-0.10	0.15	0.22	0.14	-0.17	0.15	0.07	0.08	0.07	0.07	
Father been incarcerated (Yr1)	-0.07	0.13	-0.09	0.13	0.04	0.12	0.25†	0.14	0.16	0.07	0.15	0.06	
Child's health	-0.09	0.05	-0.13*	0.05	-0.08	0.05	-0.06	0.06	-0.05	0.05	0.05	0.05	
Child is boy = 1	0.12†	0.06	0.08	0.06	-0.01	0.06	0.10	0.07	-0.07	0.07	0.03	0.06	
Child's Temperament	0.20***	0.05	0.09†	0.05	-0.05	0.05	0.06	0.05	0.14**	0.05	-0.02	0.05	

			Μ	lothers					I	Fathers		
	Random H	Effects		Cross-	Lagged	_	Rand	om Effects		Cross-	Lagged	
	Relation Qualit	ship y	Relationship Qual	ity (Year 3)	Relationship Qua	lity (Year 5)	Rel (tionship uality	Relationship Qua	lity (Year 3)	Relationship Qual	ity (Year 5)
Variables												
Relationship Quality (Year-1 lagged)		—	1.52***	.21	0.75**	.26	-	—	1.45***	0.20	0.79**	0.23
Relationship Quality (Year-3 lagged)		—	—	—	0.28***	.07	-	—	_	—	0.25***	0.07
Cohabitation = 1	16	.13	-0.13	0.14	-0.20	0.15	-0.22	† 0.13	-0.17	0.14	-0.23†	0.14
Non-Hispanic Black	-0.38*	.15	-0.45**	0.17	-0.14	0.18	-0.37	* 0.16	-0.38*	0.18	-0.16	0.18
Hispanic	16	.15	-0.48**	0.17	-0.05	0.18	-0.1	0.15	-0.42*	0.17	-0.05	0.17
Non-Hispanic Other	-1.12***	.33	-0.86*	0.37	-0.50	0.39	-1.07	** 0.35	-0.67†	0.41	-0.18	0.40
Couple is mixed race	-0.37*	.18	-0.52*	0.20	-0.05	0.22	-0.3	0.19	-0.50*	0.21	-0.14	0.21
High school diploma or equivalent	.17	.13	0.17	0.14	-0.03	0.15	-0.1	o 0.14	-0.20	0.15	-0.25†	0.15
Some college	0.48**	.14	0.35*	0.16	-0.06	0.17	0.09	0.15	-0.06	0.17	-0.07	0.16
Bachelor's degree or higher	0.71*	.32	-0.01	0.36	0.11	0.38	0.32	0.34	-0.17	0.38	-0.19	0.37
Parent's Age (in years)	02	.01	0.00	0.01	-0.01	0.01	-0.0	0.01	0.01	0.01	-0.01	0.01
Employed = 1	.06	.11	0.06	0.12	0.06	0.12	-0.0	0.14	-0.03	0.17	0.06	0.16
Social Support = 1	12	.11	-0.43**	0.13	-0.09	0.14	-0.0	0.12	-0.12	0.14	0.02	0.13
Parent's Health	03	.06	0.13*	0.07	0.00	0.07	-0.0	0.06	0.04	0.07	0.04	0.06
Parent's Depression	31	.21	0.13	0.23	0.18	0.24	-0.46	* 0.21	-0.25	0.24	-0.11	0.23
Higher-order birth = 1	09	.12	-0.12	0.13	-0.05	0.14	-0.1	0.12	-0.11	0.13	-0.02	0.13
# of children in $HH < 18$	05	.05	-0.01	0.05	-0.04	0.06	-0.0	0.05	-0.02	0.05	-0.03	0.05
Father's multipartnered fertility	04	.16	0.22	0.18	-0.30	0.19	-0.1	0.17	0.01	0.07	0.01	0.06
Mother's multipartnered fertility	04	.15	-0.10	0.17	-0.16	0.18	-0.0	0.15	0.01	0.07	0.01	0.06
Both parent's multipartnered fertility	.26	.17	0.12	0.19	-0.17	0.20	0.30	0.18	0.10	0.08	0.09	0.07
Father been incarcerated (Yr1)	-0.24†	.13	0.00	0.15	0.00	0.15	-0.28	* 0.13	-0.08	0.07	-0.07	0.06
Child's health	0.16†	.08	0.00	0.09	0.20*	0.09	0.18	• 0.08	0.07	0.09	0.17†	0.09
Child is boy = 1	.04	0.10	0.21†	0.11	-0.01	0.12	0.07	0.10	0.28*	0.12	0.06	0.12
Child's Temperament	.13	.07	0.03	0.07	0.27***	0.07	0.10	0.07	0.02	0.07	0.28***	0.07

Table 4.28: Parameter Estimates for Control Variables on Relationship Quality (Y-Standardized), by Transient Poverty

	Random	Effects		Cross	-Lagged		Random 1	Effects		Cross	Lagged	
-	Parenting	g Stress	Parenting Stres	s (Year 3)	Parenting Stres	s (Year 5)	Parenting	Stress	Parenting Stres	s (Year 3)	Parenting Stress	s (Year 5)
-	В	SE	В	SE	В	SE	в	SE	В	SE	В	SE
Variables												
Parenting Stress (Year-1 lagged)			0.82***	0.08	0.18†	0.10	-		0.66***	0.08	0.41***	0.09
Parenting Stress (Year-3 lagged)			—	—	0.52***	0.06	-		—	—	0.39***	0.06
Cohabitation = 1	-0.03	0.15	-0.14	0.14	0.13	0.13	-0.16	0.14	-0.09	0.14	-0.07	0.13
Non-Hispanic Black	0.04	0.18	0.04	0.17	0.04	0.17	0.10	0.18	0.11	0.18	-0.17	0.17
Hispanic	0.28	0.18	0.18	0.17	-0.02	0.16	0.12	0.17	0.07	0.17	-0.32*	0.16
Non-Hispanic Other	0.66†	0.38	0.05	0.36	0.46	0.35	1.10**	0.38	0.84*	0.40	0.04	0.38
Couple is mixed race	0.21	0.22	0.30	0.20	0.06	0.20	-0.03	0.21	-0.04	0.21	0.00	0.20
High school diploma or equivalent	0.01	0.15	0.12	0.14	-0.04	0.13	0.15	0.15	0.11	0.15	0.40**	0.14
Some college	-0.10	0.17	0.10	0.16	0.06	0.15	-0.35*	0.16	-0.06	0.16	0.10	0.15
Bachelor's degree or higher	-0.27	0.37	-0.29	0.35	0.06	0.34	-0.31	0.37	-0.32	0.38	0.12	0.35
Parent's Age (in years)	0.02†	0.01	0.02†	0.01	0.00	0.01	-0.02†	0.01	-0.01	0.01	-0.01	0.01
Employed = 1	0.11	0.12	0.01	0.12	0.22†	0.11	-0.18	0.16	0.01	0.16	0.08	0.15
Social Support = 1	0.09	0.13	0.02	0.13	-0.09	0.12	-0.12	0.13	-0.11	0.13	-0.13	0.12
Parent's Health	0.00	0.07	-0.01	0.07	-0.06	0.06	0.09	0.06	-0.02	0.06	0.05	0.06
Parent's Depression	0.17	0.24	0.10	0.23	0.05	0.22	-0.33	0.23	-0.26	0.23	-0.15	0.22
Higher-order birth = 1	0.18	0.13	0.11	0.13	0.00	0.12	0.15	0.13	0.22†	0.13	0.07	0.12
# of children in HH < 18	0.06	0.05	0.04	0.05	-0.08†	0.05	0.04	0.05	0.08†	0.05	-0.05	0.05
Father's multipartnered fertility	-0.01	0.19	-0.08	0.18	-0.32†	0.17	-0.23	0.18	-0.09	0.06	-0.10	0.06
Mother's multipartnered fertility	-0.06	0.18	-0.16	0.17	-0.06	0.16	-0.19	0.16	-0.14*	0.06	-0.16*	0.06
Both parent's multipartnered fertility	-0.22	0.20	-0.08	0.19	0.08	0.18	-0.22	0.20	0.07	0.07	0.07	0.07
Father been incarcerated (Yr1)	0.21	0.15	0.19	0.14	0.14	0.14	0.11	0.15	0.14*	0.06	0.16*	0.06
Child's health	0.07	0.09	0.06	0.09	0.04	0.08	-0.08	0.09	-0.14	0.09	-0.02	0.08
Child is boy = 1	0.12	0.12	-0.07	0.11	0.33**	0.11	0.12	0.12	0.16	0.12	-0.14	0.11
Child's Temperament	0.16*	0.08	0.07	0.07	0.04	0.07	0.00	0.07	-0.02	0.07	-0.07	0.07

Table 4.29: Parameter Estimates for Control Variables on Parenting Stress (Y-Standardized), by Transient Poverty

		Ν	Aother							I	athers			
	Random	Effects		Cross	·Lagged	agged				Cross-Lagged				
	Relatio Qual	nship ity	Relationship Qu	Relationship Quality (Year 3)		ality (Year 5)		Relationship Quality		Relationship Quality (Year 3)		Relationship Quality (Year 5)		
	В	SE	В	SE	В	SE		В	SE	В	SE	В	SE	
Variables														
Relationship Quality (Year-1 lagged)		_	1.60***	0.26	0.77*	.3				1.64***	0.26	0.78**	0.28	
Relationship Quality (Year-3 lagged)	_	_		—	0.39***	.08				—	—	0.37***	0.07	
Cohabitation = 1	.01	.21	-0.058	0.219	-0.155	0.21		0.02	0.20	-0.26	0.21	-0.14	0.19	
Non-Hispanic Black	15	.26	-0.216	0.273	0.404	0.262		-0.06	0.26	-0.25	0.28	0.52*	0.24	
Hispanic	32	.25	-0.298	0.261	0.49†	0.25		-0.23	0.24	-0.19	0.26	0.52*	0.23	
Non-Hispanic Other	53	.65	0.033	0.682	0.118	0.656		-0.33	0.63	0.33	0.68	0.40	0.61	
Couple is mixed race	09	.26	0.024	0.273	0.344	0.259		0.05	0.24	-0.06	0.26	0.45†	0.23	
High school diploma or equivalent	03	.17	-0.218	0.178	-0.068	0.173		-0.08	0.17	-0.03	0.18	0.09	0.16	
Some college	.12	.25	-0.119	0.263	0.157	0.252		-0.17	0.24	0.09	0.26	-0.47*	0.23	
Bachelor's degree or higher	.25	.84	-0.77	0.883	0.345	0.864		0.09	0.49	0.91†	0.53	0.51	0.48	
Parent's Age (in years)	.02	.02	0.03†	0.019	-0.004	0.019		0.01	0.02	0.01	0.02	0.00	0.01	
Employed = 1	02	.16	0.116	0.174	0.044	0.167		0.24	0.17	-0.15	0.19	0.29†	0.17	
Social Support = 1	04	.17	-0.166	0.174	0.118	0.168		0.15	0.16	-0.01	0.17	0.23	0.15	
Parent's Health	.09	.08	0.08	0.083	0.048	0.081		0.05	0.08	0.05	0.08	0.00	0.07	
Parent's Depression	26	.21	0.23	0.227	-0.01	0.222		-0.39	0.30	-0.21	0.32	-0.51†	0.28	
Higher-order birth = 1	19	.17	0.154	0.182	-0.081	0.176		-0.20	0.17	0.19	0.18	-0.13	0.16	
# of children in HH < 18	.01	.06	0.009	0.059	-0.012	0.057		0.03	0.06	0.00	0.06	-0.02	0.05	
Father's multipartnered fertility	.34	.24	0.41†	0.247	-0.154	0.242		0.33	0.25	0.01	0.07	0.01	0.06	
Mother's multipartnered fertility	12	.2	-0.309	0.214	0.018	0.206		-0.13	0.20	0.01	0.07	0.01	0.05	
Both parent's multipartnered fertility	.28	.21	0.018	0.225	0.124	0.215		0.30	0.22	0.11	0.08	0.08	0.07	
Father been incarcerated (Yr1)	-0.33†	.18	-0.34†	0.182	-0.19	0.181		-0.36*	0.17	-0.09	0.07	-0.07	0.05	
Child's health	04	.09	0.077	0.092	-0.065	0.088		0.00	0.08	0.05	0.09	-0.07	0.08	
Child is boy = 1	.17	0.15	0.038	0.16	0.082	0.154		0.17	0.15	-0.08	0.16	0.10	0.14	
Child's Temperament	-0.23*	0.10	-0.073	0.103	-0 17÷	0.098		-0 28**	0.10	-0.10	0.11	-0.14	0.09	

Table 4.30: Parameter Estimates for Control Variables on Relationship Quality (Y-Standardized), by Persistent Poverty

	Random Effects Cross-Lagged						Danden Effects							
	Random	Effects	Deropting Stree	Cross-	Departure Stree	(Voor 5)	Randor	n Effects	Paranting Stree	Cross-	Lagged	e (Voor 4		
	Tarenting	50003	T architing Sires	s (1 cai 5)	T architing Sucs	s (i cai 5)		ing Sucess		s (rear 5)	T architing Sucs	s (i cai .		
	В	SE	В	SE	В	SE	В	SE	В	SE	В	SE		
ibles														
Parenting. Stress (Year-1 lagged)			0.66***	0.10	0.40***	0.11	_		0.75***	0.08	0.25*	0.1		
arenting Stress (Year-3 lagged)	_	_	_	_	0.39***	0.08	_	_	_	_	0.43***	0.08		
Cohabitation = 1	-0.11	0.22	0.07	0.20	-0.07	0.19	0.03	0.24	-0.05	0.19	-0.06	0.19		
Jon-Hispanic Black	0.07	0.28	0.21	0.25	-0.26	0.24	-0.33	0.30	0.05	0.24	-0.19	0.24		
Iispanic	-0.17	0.26	0.02	0.24	-0.12	0.23	-0.36	0.29	0.02	0.23	-0.40†	0.23		
Non-Hispanic Other	0.10	0.69	-0.14	0.63	0.59	0.60	-0.33	0.74	-0.29	0.60	-0.24	0.60		
Couple is mixed race	-0.37	0.27	0.13	0.25	-0.39	0.24	0.20	0.28	0.16	0.23	-0.23	0.2		
ligh school diploma or equivalent	0.00	0.18	-0.19	0.17	0.13	0.16	0.25	0.20	0.40*	0.16	0.03	0.1		
ome college	-0.09	0.26	0.17	0.24	0.07	0.23	0.15	0.28	0.26	0.23	0.23	0.23		
achelor's degree or higher	0.70	0.89	2.41**	0.80	0.16	0.80	0.11	0.58	0.24	0.48	0.14	0.4		
'arent's Age (in years)	0.04†	0.02	0.04*	0.02	0.01	0.02	-0.01	0.02	-0.03*	0.01	0.00	0.02		
mployed = 1	-0.30†	0.17	-0.19	0.16	0.13	0.15	-0.36†	0.20	-0.35*	0.16	0.01	0.1		
Social Support = 1	0.39*	0.17	0.20	0.16	0.08	0.16	-0.10	0.18	-0.19	0.15	-0.42**	0.1:		
Parent's Health	-0.15†	0.08	-0.13	0.08	0.08	0.07	-0.19*	0.09	0.02	0.08	-0.15†	0.0		
Parent's Depression	0.35	0.22	0.30	0.21	0.10	0.20	-0.05	0.35	-0.41	0.28	0.40	0.2		
Higher-order birth = 1	-0.19	0.18	-0.12	0.17	-0.25	0.16	0.28	0.20	0.13	0.16	-0.26†	0.1		
of children in HH < 18	0.02	0.06	0.09	0.06	0.02	0.05	-0.04	0.07	-0.02	0.05	0.06	0.0		
ather's multipartnered fertility	-0.01	0.25	-0.15	0.23	0.39†	0.22	-0.33	0.28	-0.08	0.06	-0.09	0.0		
Mother's multipartnered fertility	-0.09	0.21	0.13	0.20	-0.06	0.19	-0.09	0.23	-0.14*	0.05	-0.14*	0.0		
Both parent's multipartnered fertility	0.12	0.22	0.27	0.21	0.30	0.20	0.03	0.25	0.06	0.06	0.06	0.0		
Father been incarcerated (Yr1)	-0.19	0.19	-0.22	0.17	0.05	0.17	0.12	0.20	0.14*	0.06	0.14*	0.0		
Child's health	-0.17†	0.09	-0.02	0.09	-0.23**	0.08	0.06	0.10	0.01	0.08	-0.09	0.0		
hild is boy $= 1$	-0.41**	0.16	-0.18	0.15	-0.17	0.14	-0.06	0.17	0.06	0.14	0.03	0.1		
'bild's Temperament	-0.04	0.10	0.01	0.10	0.08	0.00	0.21*	0.11	-0.05	0.09	-0.10	0.0		

Table 4.31: Parameter Estimates for Control Variables on Parenting Stress (Y-Standardized), by Persistent Poverty

CHAPTER V

Discussion & Conclusion

RE-INTRODUCTION

Empirical studies indicate that, on average, children living in stable, two-parent, wellfunctioning, and higher income families tend to have better overall well-being (Brown 2004, 2010; Teachman 2008). In contrast, children living with unmarried parents (or lowincome married couples) are at an increased risk of experiencing family instability and economic disadvantage (Avellar and Smock 2005; Jalovaara 2003; Lichter, Qian, and Mellott 2006), which is associated with adverse outcomes such as academic and behavioral problems (McLanahan 2004, 2009; Thomson, Hanson, and McLanahan 1994). These disadvantages can set the stage for long-term intergenerational inequality for many children and families (Amato and Cheadle 2005; Sobolewski and Amato 2005).

Based on these prior findings, and with intentions to improve the stability and quality of intimate partners with children, policymakers have put forth efforts to build strong families by strengthening couple relationships as a way to encourage family stability and favorable outcomes for children. Initial arguments about strengthening couple's relationships were based on assumptions that many low-income families did not value marriage (e.g., Karney and Bradbury 2005). To the contrary, qualitative and quantitative studies show that low-income couples desire to marry but other factors such as the lack of financial resources and other interpersonal stressors hinder many parents from getting or staying married (Edin and Kefalas 2005; Edin 2000; Gibson-Davis, Edin, and McLanahan 2005; Randles 2012). Thus, if state and federal policy-makers are serious about building strong families, programmatic efforts must also help to directly tackle the interpersonal and contextual stressors many couples experience (Johnson 2012; Karney and Bradbury 2005).

Since the emergence of policies that are aimed at strengthening couple's relationships, empirical studies have consistently found that intimate partners are faced with challenges that affect both the parent-child and mother-father relationship after the birth of a child (Nomaguchi and Milkie 2003). Yet, as current family policies stand, many scholars question the effectiveness of state and federal strategies for strengthening couple relationships because they do not address the intricacies of dyadic relationships, especially interpersonal and contextual stressors (for a review of the debate, see Hawkins, Amato, and Kinghorn 2013; Hawkins and Ooms 2012; Hawkins, Stanley, et al. 2013; Johnson 2012, 2013, 2014). Moreover, given the differences between social groups in the levels of both parental stress/distress and dyadic/couple functioning, can policy efforts be culturally sensitive and context-specific in the implementation of programs to strengthen couple's relationships? There is no simple answer to this question. However, it may be beneficial to address couple's parental and contextual stressors as these factors contribute to the stability and quality of intimate partners.

Thus, the research presented in this dissertation is concerned with parental stress/distress that affect couple's relationships, and the extent to which positive, supportive relationships reduce parental stress and distress. Further, the three empirical essays that are included in this dissertation addressed how these processes unfold following the birth of a child as this is a critical time when couples are more vulnerable to parental stress and relationship strain (Nomaguchi and Milkie 2003). Last, this research examined contextual factors moderated he associations. Undoubtedly, there is much

complexity in family and dyadic relationships. Examining parental stress and dyadic functioning together can help unpack and highlight this complexity by revealing the relative importance of these factors and how they vary across social contexts. A discussion of the research findings from the prior chapters is provided below.

DISCUSSION

Conceptual & Theoretical Model

Drawing on several theoretical perspectives, I have attempted to create an integrated conceptual and theoretical model that addresses the longitudinal and reciprocal association between parental stress/distress and dyadic functioning, whether these processes vary across social context, and whether the association changes over time. This framework builds on prior conceptual models that emphasize the importance of stress and contextual factors for studying intimate partners (Huston 2000; Johnson 2012; Karney and Bradbury 1995). By emphasizing the direct and reciprocal pathways between parental stress/distress and dyadic functioning, and examining the continuity and change in these processes as a function of children's early developmental stages, this dissertation pays attention to the multilayered and interdependent nature of many factors that affect each individual parent and the intimate partnership. Scholars continue to suggest that taking a more holistic approach (as implemented in this research) advances our understanding of couple dyads in theoretically and empirically informed ways (Helms 2013). The conceptual and theoretical model was thus used as a blueprint to guide the empirical analyses across the analytic studies in Chapters II, III, and IV.

In Chapter II, I investigated the association between economic hardship, parents' depressive symptoms, and relationship distress. This study was executed in two parts: the first part examined the longitudinal, dyadic, and reciprocal association between economic

hardships and parents' depressive symptoms, and the second part focused on the effects of both economic hardship and parents' depressive symptoms on mothers' and fathers' report of relationship distress. The findings revealed that hardships lead to higher levels of depressive symptoms as children age from infant to toddler for both mothers and fathers, and from toddler to preschool for mothers only. Reciprocally, maternal depressive symptoms was associated with an increase in family hardships as children aged from infant to toddler. Also, during the preschool years (age 5), economic hardship and depressive symptoms increased relationship distress for both mothers and fathers. These findings were similar across married and cohabiting couples.

Overall, consistent with family stress (Conger, Conger, and Martin 2010; Umberson and Reczek 2007) and stress crossover (Bolger et al. 1989; Neff and Karney 2007; Thompson and Bolger 1999) perspectives, Chapter II showed that economic hardship have consistent effects on parents' mental health over time, and mother's early mental health contributes to later family hardships. Moreover, each parent's mental health creates distress within the relationship revealing both an actor and cross-partner effect which extends prior studies on parental mental health and relationship outcomes.

Chapter III explored parents' depressive symptoms and coparenting as a longitudinal and reciprocal process. Using a series of actor-partner interdependent models, the results showed that depressive symptoms lowered coparenting for mothers and fathers as their child aged from one to three-years of age, and for mothers only as their child aged from three- to five-years of age. Conversely, coparenting reduced the rate of depressive symptomology for parents at the Year-1 to Year-3 (infant to toddler) but not Year-3 to Year-5 (toddler to preschool). These findings corroborate the few studies that previously investigated this association (e.g., Cabrera, Shannon, and La Taillade 2009; Elliston et al. 2008). Further, the results also revealed that depressive symptoms was associated with a decline in coparenting for Hispanic mothers and fathers, White mothers, and Black mothers and fathers. Conversely, cooperative coparenting was associated with lower depressive symptoms levels for Black fathers, White fathers and mothers.

Taken together, the Chapter III results indicated that mental health is detrimental to coparenting relationship, especially among mothers. Conversely, cooperative coparenting reduced mental health risk for fathers. Not only do these results corroborate findings from prior studies, they also advance our knowledge of the importance of coparenting within couple dyads by showing that parents who work together in the joint enterprise of child rearing has mental health advantages. These findings also suggest that more culturally sensitive approaches to mental health and coparenting are needed because the processes that drive them can differ across race and ethnicity and children's developmental stages.

Chapter IV examined the longitudinal and reciprocal association between parenting stress and couple's relationship quality. Using several analytic techniques, the findings revealed that parenting stress for both mothers and fathers was associated with lower levels of couple's relationship quality. Couple relationship quality was also linked to lower levels of parenting stress in both the fixed- and random-effects models. The cross-lagged model revealed that, for mothers, early positive reports of couple's relationship quality lowered maternal parenting stress at the later years. The findings also showed that, among families who experienced no poverty and families who experienced persistent poverty, being in a positive relationship reduced maternal parenting stress. By

In many ways, Chapter IV demonstrated that parenting stress and couple's relationship quality are empirically related, as shown in with prior studies. The longitudinal results extend prior finding, however, by showing that couple's relationship quality in the early years lowers maternal parenting stress at the later years. Early parenting stress, however, does not affect the quality of the relationship at the later years. Simply put, supportive partnerships leads to less stress in the parenting role, at least for mothers.

The theoretical and conceptual model guiding this dissertation provides a framework for understanding intimate partnerships after the birth of a child. The framework underscores the complexity of couple dyads by highlighting the parental stressors, dyadic functioning, the social context, and the stability and change in these factors as their child develops over time. Given that the findings in the analytic chapters corroborates findings from previous research, this dissertation moves this line of research on dyadic relationships forward by simultaneously examining both mediating and moderating factors that contribute to the quality of intimate partnerships. Echoing Huston's (2000: 299) social ecological model of marriage and other intimate unions, the present framework builds on the idea that:

(a) marriages are interpersonal systems (and hence must be studied as small groups), (b) that spouses' psychological and physical qualities shape their individual and collective efforts maintain a successful union, (c) that both marriage relationships and the partner themselves are dynamic (i.e., they change by context and they evolve over time), and (d) the martial union are embedded in a social context.

200
The research presented here is rooted in similar concepts and attempts to shine new light on the complex portrait of families by revealing that family life involves interrelated systems that contribute to parental stress/distress and the quality of intimate partnerships. The intentions of this framework was to paint a more holistic picture of couple dyads, and provide a step forward in theorizing and empirically testing the association between important familial processes.

Indeed, the analytic chapters within this dissertation provide novel evidence that research among relational dyads can be best understood when both members of the family (i.e., mothers and fathers) are examined in an ecologically sensitive way. These findings can be used to assist in efforts to help strengthen couple's relationships and thus children's well-being.

Implications

This dissertation has important implications for both research and policy. In terms of research, the results underscore and highlight the notion that individuals within families are interdependent (Cox and Paley 1997; O'Brien 2005) and parents are involved in "interlocking trajectories" as their child ages and develops over time (Elder, George, and Shanahan 1996; Elder 1998). Clearly, focusing on stress or support for only one parent is unlikely to be sufficient. The variations across chapters points to the overall complexity of family life. Thus, rather than driving home a consistent message, the results illustrate that different domains, whether dyadic or individual, personal or interpersonal, move according to their own rules. The implication is that targeting a single specific pathway may not be sufficient to influence a multiplicity of outcomes. To positively influence family life, multiple pathways must be targeted if we, as a society, are willing to help families achieve adequate financial support and family stability. These findings enhance

our understating of interpersonal and contextual stressors, dyadic functioning, reciprocity among couples, and the cross-partner associations within couples. Future research can continue to expand our understanding on complex, multi-layered familial processes by using integrated theoretical frameworks with advanced analytic techniques (e.g., Helms 2013).

In regards to policy implications, policymakers have put forth efforts (e.g., Deficit Reduction Act of 2005) to strengthen couples' relationships by allocating \$100 million a year for 5 years to help couples form and maintain healthy marriages and relationships through free marriage and relationship education (MRE) programs, especially among low-income families (Brown 2010; Dion 2005). Recently, in 2011, funding was continued for three additional years—although the amount was reduced to \$75 million (Hawkins, Amato, et al. 2013). The results from this dissertation reveal that there are some important stressors (e.g., economic hardships, depressive symptoms, and parenting stress) that affect the quality of intimate partnerships. Moreover, being in a positive, wellfunctioning partnership that promotes healthy interactions between partners and allows partners to work together for the benefit of the child (i.e. cooperative coparenting) plays an important role in the reduction in parental stress/distress.

Thus, these findings suggest that social policies, in tandem with building relationship skills, should also target families' economic, mental health, and stress in the parenting role which could yield important benefits for couple's relationships (e.g., Letiecq, Anderson, and Joseph 2013). In addition, it appears that strengthening couple's relationships has promising benefits for parents, and policy efforts should also incorporate ways to focus on increasing coparenting between parents as these family processes have been shown to be valuable for children's well-being.

Limitations

Although the research presented in this dissertation provide important insight to the study of stress/distress and dyadic functioning among couple dyads up to five years after the birth of a child, there are some limitations that must be noted. First, given that each chapter uses data from parents who live together over time, some of the respondents drop out of the study due to attrition or relationship dissolution. These respondents are more economically disadvantage, race and ethnic minorities, and cohabitors, who are all likely to experience more stress/distress and less positive interactions. As such, the findings may be underestimating the association between parental stress/distress on dyadic functioning, and overestimating the association between dyadic functioning on parental stress/distress. Second, because the FFCW study focuses exclusively on families living in urban areas, the inferences do not extend to the general population. Notably, the findings are consistent with predominately white, middle-class samples, with some nuance in the results across social groups. Lastly, given the complexity of the theoretical model, I did not explicitly test for parents' personal qualities such as self-esteem, mastery, problem solving abilities, and a host of other factors that may serve as important and mediating factors that affect the longitudinal and reciprocal association between parental stress/distress and dyadic functioning (e.g., Helms 2013). Indeed, future research extending the dyadic models used here is warranted.

Strengths

Despite the limitation, this dissertation makes an important contribution to the study of couple dyads after a recent birth. First, by combining complex family processes into an

integrated model, each empirical study attempts to unveil the mechanisms that affect couple dyads. This approach yielded important advances to family research because it allows multiple factors to interact, and thus shines new light on family processes that are often hidden in cross-sectional models, or models that aggregate over broad age ranges of the child and so cannot capture how processes vary after a child's birth. Second, taking advantage of the dyadic data, the analytic techniques employed in the empirical chapters simultaneously take into account both the intra-individual and inter-dyadic processes. Last, using a representative sample of births to urban parents, the findings from this dissertation corroborate and extend prior research on couples that have been disproportionately small, White middle-class samples. Specifically, in an era of increased need for more diverse empirical studies on couple dyads, this research is well-positioned to add to growing body of knowledge that seeks to improve couple's relationships in order to reduce inequality, and ensure children's well-being.

CONCLUSION

This dissertation used a large sample of urban parents who had a child in the late 1990s to examine the following: (a) stress/distress affects dyadic functioning, (b) positive dyadic functioning can reduce stress/distress, (c) how these processes vary across social context, and (d) over time. Taking advantage of the longitudinal design of these data and using multiple analytic strategies, the findings reveal novel associations between parental stress/distress and dyadic functioning, which appears to be critically important for family life. In summary, the aggregate findings in this dissertation revealed that stressed parents make less supportive partners, and having a supportive partner is a valuable resource for stressed parents. Taking together, these findings suggest that the stress-dyadic functioning conundrum may be best understood using multiple theoretical perspectives and advanced analytic techniques to address the complex and dynamic ways in which couples experience their lives, especially after the birth of a child.

REFERENCES

- Amato, Paul R., and Jacob Cheadle. 2005. "The Long Reach of Divorce: Divorce and Child Well-Being across Three Generations." *Journal of Marriage and Family* 67(1):191–206.
- Avellar, Sarah, and Pamela J. Smock. 2005. "The Economic Consequences of the Dissolution of Cohabiting Unions." *Journal of Marriage and Family* 67(2):315– 27.
- Bolger, Niall, Anita DeLongis, Ronald C. Kessler, and Elaine Wethington. 1989. "The Contagion of Stress across Multiple Roles." *Journal of Marriage and the Family* 175–83.
- Brown, Susan L. 2004. "Family Structure and Child Well-Being: The Significance of Parental Cohabitation." *Journal of Marriage & Family* 66(2):351–67.
- Brown, Susan L. 2010. "Marriage and Child Well-Being: Research and Policy Perspectives." *Journal of Marriage and Family* 72(5):1059–77.
- Cabrera, Natasha J., Jacqueline D. Shannon, and Jaslean J. La Taillade. 2009. "Predictors of Coparenting in Mexican American Families and Links to Parenting and Child Social Emotional Development." *Infant mental health journal* 30(5):523–48.
- Conger, Rand D., Katherine J. Conger, and Monica J. Martin. 2010. "Socioeconomic Status, Family Processes, and Individual Development." *Journal of Marriage and Family* 72(3):685–704.
- Cox, Martha J., and Blair Paley. 1997. "Families as Systems." *Annual Review of Psychology* 48(1):243–67.
- Dion, M. Robin. 2005. "Healthy Marriage Programs: Learning What Works." *The Future* of Children 15(2):139–56.
- Edin, Kathryn. 2000. "What Do Low-Income Single Mothers Say about Marriage?" Social Problems 112–33.
- Edin, Kathryn, and Maria Kefalas. 2005. Promises I Can Keep: Why Poor Women Put Motherhood Before Marriage. Berkeley, CA: University of California Press.
- Elder, Glen H. Jr. 1998. "The Life Course as Developmental Theory." *Child Development* 69(1):1–12.
- Elder Jr, Glen H., Linda K. George, and Michael J. Shanahan. 1996. "Psychosocial Stress over the Life Course." Pp. 247-292 in *Psychosocial Stress: Perspectives on Structure, Theory, Life-Course, and Methods*. San Diego, CA: Academic Press

- Elliston, Donna, James McHale, Jean Talbot, Meagan Parmley, and Regina Kuersten-Hogan. 2008. "Withdrawal from Coparenting Interactions during Early Infancy." *Family process* 47(4):481–99.
- Gibson-Davis, Christina M., Kathryn Edin, and Sara McLanahan. 2005. "High Hopes but Even Higher Expectations: The Retreat from Marriage among Low-Income Couples." *Journal of Marriage and Family* 67(5):1301–12.
- Hawkins, Alan J., Scott M. Stanley, et al. 2013. "A More Optimistic Perspective on Government-Supported Marriage and Relationship Education Programs for Lower Income Couples." *American Psychologist* 68(2):110-111
- Hawkins, Alan J., Paul R. Amato, and Andrea Kinghorn. 2013. "Are Government-Supported Healthy Marriage Initiatives Affecting Family Demographics? A State-Level Analysis." *Family Relations* 62(3):501–13.
- Hawkins, Alan J., and Theodora Ooms. 2012. "Can Marriage and Relationship Education Be an Effective Policy Tool to Help Low-Income Couples Form and Sustain Healthy Marriages and Relationships? A Review of Lessons Learned." *Marriage* & Family Review 48(6):524–54.
- Helms, Heather M. 2013. "Marital Relationships in the Twenty-First Century." Pp. 233– 54 in *Handbook of Marriage and the Family*. Springer.
- Huston, Ted L. 2000. "The Social Ecology of Marriage and Other Intimate Unions." *Journal of Marriage and Family* 62(2):298–320.
- Jalovaara, Marika. 2003. "The Joint Effects of Marriage Partners' Socioeconomic Positions on the Risk of Divorce." *Demography* 40(1):67–81.
- Johnson, Matthew D. 2012. "Healthy Marriage Initiatives: On the Need for Empiricism in Policy Implementation." *American Psychologist* 67(4):296–308.
- Johnson, Matthew D. 2013. "Optimistic or Quixotic? More Data on Marriage and Relationship Education Programs for Lower Income Couples." *American Psychologist* 68(2):111-112
- Johnson, Matthew D. 2014. "Government-Supported Healthy Marriage Initiatives Are Not Associated with Changes in Family Demographics: A Comment on Hawkins, Amato, and Kinghorn (2013)." *Family Relations* 63(2):300–304.
- Karney, Benjamin R., and Thomas N. Bradbury. 1995. "The Longitudinal Course of Marital Quality and Stability: A Review of Theory, Method, and Research." *Psychological Bulletin* 118(1):3–34.
- Karney, Benjamin R., and Thomas N. Bradbury. 2005. "Contextual Influences on Marriage Implications for Policy and Intervention." *Current Directions in Psychological Science* 14(4):171–74.

- Letiecq, Bethany L., Elaine A. Anderson, and Alfred L. Joseph. 2013. "Social Policies and Families through an Ecological Lens." Pp. 751–79 in *Handbook of Marriage and the Family*.
- Lichter, Daniel T., Zhenchao Qian, and Leanna M. Mellott. 2006. "Marriage or Dissolution? Union Transitions among Poor Cohabiting Women." *Demography* 43(2):223–40.
- McLanahan, Sara. 2004. "Diverging Destinies: How Children Are Faring under the Second Demographic Transition." *Demography* 41(4):607–27.
- McLanahan, Sara. 2009. "Fragile Families and the Reproduction of Poverty." *The Annals* of the American Academy of Political and Social Science 621(1):111–31.
- Neff, Lisa A., and Benjamin R. Karney. 2007. "Stress Crossover in Newlywed Marriage: A Longitudinal and Dyadic Perspective." *Journal of Marriage and Family* 69(3):594–607.
- Nomaguchi, Kei M., and Melissa A. Milkie. 2003. "Costs and Rewards of Children: The Effects of Becoming a Parent on Adults' Lives." *Journal of Marriage and Family* 65(2):356–74.
- O'Brien, Marion. 2005. "Studying Individual and Family Development: Linking Theory and Research." *Journal of Marriage and Family* 67(4):880–90.
- Randles, Jennifer M. 2012. "Marriage Promotion Policy and Family Inequality." Sociology Compass 6(8):671–83.
- Sobolewski, Juliana M., and Paul R. Amato. 2005. "Economic Hardship in the Family of Origin and Children's Psychological Well-Being in Adulthood." *Journal of Marriage and Family* 67(1):141–56.
- Teachman, Jay D. 2008. "The Living Arrangements of Children and Their Educational Well-Being." *Journal of Family Issues* 29(6):734–61.
- Thompson, A., and N. Bolger. 1999. "Emotional Transmission in Couples under Stress." *Journal of Marriage and the Family* 61(1):38–48.
- Thomson, Elizabeth, Thomas L. Hanson, and Sara S. McLanahan. 1994. "Family Structure and Child Well-Being: Economic Resources vs. Parental Behaviors." *Social Forces* 73(1):221–42.
- Umberson, Debra, and Corinne Reczek. 2007. "Interactive Stress and Coping around Parenting: Explaining Trajectories of Change in Intimate Relationships over the Life Course." Advances in Life Course Research 12:87–121.